HUMAN PLATELET ACTIVATION BY ADRENALINE

by

HELEN AUDREY CAMERON

A thesis submitted for the degree of Doctor of Philosophy of the Australian National University.

May,1985.

STATEMENT

Except where acknowledged, the investigations described in this thesis are my own original work.

H.A.CAMERON

St. A. Bameron

May, 1985.

I wish to thank Dr. Neville Ardlie for his assistance and supervision of my thesis.

I also wish to thank Mrs. Carmel Boatwright for excellent technical advice and assistance throughout the course of this study; Mrs. Carol Cvetanoski for assistance with many aggregation studies and radioimmunoassays; Dr. Mark White for valuable discussion in biochemistry; Drs. N. Hunt,J.Jakubowski,B. Creese,J. Sullivan and P.Han for helpful discussion; Mrs. Gillian Scott and the Word Processing Bureau, John Curtin School of Medical Research for typing this thesis; Mrs. Jill Garrett for preparing the figures and assistance with chromatography; the staffs of Department of Medicine and Clinical Science, Royal Canberra Hospital and A.C.T.Red Cross Blood Bank for blood donations, and the library and photography department,John Curtin School of Medical Research for their respective assistance.

Financial support from the Australian National University is gratefully acknowledged. To Carl

Midleto:

PREFACE

Adrenaline may facilitate platelet reactions in the normal haemostatic response and, if increased platelet reactivity leads to atherosclerosis, adrenaline could thereby contribute to atherogenesis. Adrenaline levels rise during stress, and in other risk factor conditions for coronary artery desease. The action of adrenaline by itself on platelets can be inhibited by cyclooxygenase inhibitors, but the ability of adrenaline to potentiate the response to other agonists is not completely inhibitable by these drugs. The mechanisms of potentiation of platelet responses by adrenaline is the subject of this thesis.

Chapter 1 emphasises the platelet as a subject of greater interest in the pathogenesis of atherosclerosis and its complications and outlines its structure and functions in relation to this. The known effects of adrenaline on the platelet, together with current understanding of pathways of platelet activation are discussed. Chapter 2 describes the materials and methods employed in the course of the investigations carried out.

Chapters 3-7 describe the experimental work undertaken to investigate some aspects of adrenaline-platelet interaction. The exploration of the effects of multiple agonists, including adrenaline and noradrenaline, is described in Chapter 3. Investigation of the mechanism(s) of potentiation by adrenaline is commenced by the verification in Chapter 4 of previously published evidence that adrenaline stimulates both a prostaglandin-independent and a prostaglandin-dependent pathway.

Chapter 5 reports the unexpected finding of stimulated thromboxane formation, by adrenaline, in the presence of the phospholipase A_2 inhibitors, BPB and TPCK, suggesting involvement of adrenaline in arachidonic acid release by a pathway not involving phospholipase A_2 enzymes. The hypothesis is put forward that adrenaline in the presence of other agonists stimulates the release of arachidonic acid from diacylglycerol by diacylglycerol lipase.

Chapter 6 examines the characteristics of platelet aggregation, release of nucleotide, thromboxane formation and phospholipid changes due to exogenous phospholipase C. In the presence of BPB, exogenous phospholipase C is apparently more active as indicated by the increased breakdown of PC, but does not result in further PA production, indicating that perhaps diacylglcerol lipase is more active in these circumstances. Thrombin-stimulated endogenous phospholipase C did not, however, behave in this manner in the presence of BPB.

The possibility was considered that adrenaline influences PI turnover and thereby achieves potentiation. Studies of phospholipid metabolism were therefore undertaken. The effects of adrenaline on ADP- and thrombin-induced changes in phospholipids is the subject of Chapter 7. The development of a paper chromatographic technique for the estimation of PI and PA turnover and amounts is reported.

vi

Chapter 8 reviews and discusses the findings of Chapters 3-7 and attempts to establish hypotheses for the mechanism(s) by which adrenaline influences platelets to ensure that they respond more effectively to other agonists. Phospholipid changes due to adrenaline proved difficult to demonstrate. Calcium redistribution was not explored in this thesis and there obviously remains, many aspects of the adrenaline-platelet interaction which need to be investigated.

CONTENTS

	page
STATEMENT ACKNOWLEDGMENTS DEDICATION DEFEACE	ii iii iv
LIST OF FIGURES LIST OF TABLES ABBREVIATIONS	xi xiii xiv
CHAPTER 1 GENERAL INTRODUCTION	
1.1 Historical Background	1
1.2 Platelet Structure 1.2.1 The Platelet Surface 1.2.2 Microfilaments and Microtubules 1.2.3 Platelet Organelles	3 3 7 8
1.3 Platelet Function 1.3.1 Maintenance of Integrity of the Vasculature 1.3.2 Inflammation 1.3.3 Haemostasis 1.3.4 Aberrant Haemostasis: Thrombosis and Atheroslerosis	9 10 11 12 14
<pre>1.4 Platelet Interactions 1.4.1 Catecholamines 1.4.2 Serotonin(5-hydroxytryptamine) 1.4.3 Thrombin 1.4.4 Conective Tissue 1.4.5 Adenine Nucleotides</pre>	19 19 24 25 27 27
1.5 Mechanisms of Platelet Responses 1.5.1 The Necessity for Calcium 1.5.2 Control by c'AMP 1.5.3 Prostaglandins and the Release Reaction 1.5.4 Phospholipid Turnover 1.5.5 Platelet Activating Factor	29 29 34 36 40 47
1.6 Aims of this Project in Relation to Present Models of Platelet Aggregation	49
CHAPTER 2 METHODS AND MATERIALS	
2.1 Methods 2.1.1 Aggregation Studies 2.1.2 Radioimunoassay of ThromboxaneB ₂ 2.1.3 Phospholipid Analysis	54 54 61 63
2.2 Materials 2.2.1 Buffers and Solutions 2.2.2 Agggregating Agents 2.2.3 Inhibitors 2.2.4 Anticoagulants	67 67 69 70 72

			page
		<pre>2.2.5 RIA Reagents 2.2.6 Solvents for Paper Chromatography 2.2.7 Reagents for Phosphorus Assay</pre>	72 73 73
CHAPTER	3 N 3.0	AULTIPLE AGONISTS <u>IN VITRO</u> Summary	75
	3.1	Introduction	76
	3.2	Methods and Materials	79
	3.3	Results 3.3.1 The Use of Low Concentrations of	81
		Heparin to Prepare PRP 3.3.2 Agonist Interactions Involving Adrenaline, ADP, Vasopressin and	81
		Noradrenaline	88
	3.4	Discussion	97
CHAPTER	4 N	MECHANISM OF THE FACILITATING EFFECT OF ADRENALINE ON PLATELET AGGREGATION	
	4.0	Summary	100
	4.1	Introduction	101
	4.2	Methods and Materials	104
	4.3	Results 4.3.1 Potentiation of ADP-induced	104
		Aggregation by adrenaline 4.3.2 Potentiation of Potassium Arach-	104
		idonate(KA)-induced Aggregation 4.3.3 Potentiation of ADP-induced and KA- induced Aggregation by Clonidine and	109
		Phenylephrine 4.3.4 Post-receptor Inhibitors	114
	4.4	Discussion	117
CHADTED	5 1	THE INFILIENCE OF PHOSPHOLIPASE A	
OIM TER	5	INHIBITORS ON ADRENALINE POTENTIATION	
	5.0	Sumary	124
	5.1	Introduction	125
	5.2	Methods and Materials	129
	5.3	Results	130
		5.3.1 Potentiation of the ADP-response by Adrenaline in the Presence of BPB	130

	5.3.2	Potentiation of the ADP-response by	page
		Presence of BPB	133
	5.3.3	A comparison of the Effects of TPCK and BPB on Adrenaline Potentiation	133
	5.3.4	Influence of Phenothiazines on the Potentiation of ADP by Adrenaline	134
	5.4 Discus	sion	137
CHAPTER	6 THE EFF ON PLAT	TECTS OF EXOGENOUS PHOSPHOLIPASE C	
	6.0 Summar	У	144
	6.1 Introd	luction	145
	6.2 Method	s and Materials	148
	6.3 Result	S Charactoristics of the Peeperso to	149
	6 2 2	Phospholipase C	149
	6.3.3	Exogenous Phospholipase C The Effects of Inhibitors on Phospholipid Changes Due to	156
	624	exogenous Phospholipase C	158
	0.3.4	Phospholipid Changes Due to Thrombin	162
	6.4 Discus	sion	164
CHAPTER	7 THE EFF MEMBRAN	ECT OF ADRENALINE ON PLATELET NE PHOSPHOLIPIDS	169
	7.0 Summar	У	169
	7.1 Introd	luction	169
	7.2 Method	s and Materials	171
	7.3 Result 7.3.1 7.3.2 7.3.3	s Chromatography with Solvent System A Chromatography with Solvent System B The Effects of Adrenaline on ADP-	172 172 176
		induced Metabolism of PI	179
	7.4 Discus	sion	182
CHAPTER	8 GENERAL	DISCUSSION	
	8.1 Discus	sion	186
	8.2 Conclu	sions	194
REFERENC	CES		197
PERSONAL	L PUBLICATI	ONS	222

LIST OF FIGURES

		200
	The latitude of a het light of the old out the posterior	age
1.1	Arachidonic acid metabolism.	37
1.2	The phosphatidylinositol cycle.	42
3.1	The effect of heparin concentration on ADP threshold.	83
3.2	High concentrations of heparin prevent biphasic aggregation.	84
3.3	The effect of added heparin on the response to the combine addition of adrenaline, vasopressin and ADP.	86
3.4	The potentiation of adrenaline and ADP by thrombin in PRP anticoagulated with 10mM citrate.	87
3.5	The effect of preincubation with hirudin(1U/ml), a specific thrombin inhibitor, on the response to adrenaline, vasopressin and ADP.	88
3.6	The positive interaction of adrenaline and ADP, vasopressin and ADP.	92
3.7	The positive interaction of adrenaline, vasopressin and ADP.	93
3.8	The positive interaction of adrenaline, vasopressin and ADP.	94
3.9	The positive interaction of adrenaline, noradrenaline, vasopressin and ADP.	95
3.10	The positive interaction of adrenaline, vasopressin, 5-hydroxytryptamine and ADP.	96
4.1	Potentiation of ADP-induced aggregation by adrenaline.	106
4.2	The influence of aspirin on the potentiation of ADP-induced aggregation by adrenaline	107
4.3	The influence of indomethacin on the potentiation of ADP-induced aggregation by adrenaline.	108
4.4	The potentiation of KA-induced aggregation by adrenaline.	110
4.5	The influence of aspirin on the potentiation of KA-induced aggregation by adrenaline.	111

xi

xii

		page
4.6	The influence of n-butylimidazole on the potent- iation of KA-induced aggregation by adrenaline.	113
4.7	Potentiation of ADP-induced aggregation by adrenaline, clonidine and phenylephrine.	116
5.1	The release of arachidonic acid from platelet phospholipids.	126
5.2	Potentiation of the response to ADP by adrenaline in the presence of BPB.	132
5.3	Potentiation of the response to ADP by adrenaline in the presence of BPB or TPCK.	135
5.4	Influence of chlorpromazine(CPZ) on adrenaline potentiation of the ADP-response.	136
6.1	Platelet aggregation and release caused by exogenous phospholipase C.	150
6.2	The effects of verapamil on aggregation and release due to exogenous phospholipase C.	152
6.3	The additive inhibitory effect of BPB and indomethacin on aggregation due to exogenous phospholipase C.	153
6.4	The effects of BPB on aggregation and release due to phospholipase C.	155
6.5	Chromatograms of unstimulated(A) and phospholipase C-activated(B) platelets.	157
6.6	Production of phosphatidic acid during platelet aggregation due to phospholipase C.	159
7.1	Chromatogram of platelet membrane lipids using solvent system A.	174
7.2	The mobilities of standard phosphatidic acids in each phase of solvent system A.	175
7.3	Chromatograms of platelet membrane lipids from unstimulated and thrombin-stimulated platelets.	177
7.4	Chromatograms pf platelet membrane lipids from unstimulated and phospholipase C-stimulated platelets.	178
7.5	Chromatograms of standard phospholipids using solvent system B.	180

LIST OF TABLES

		page
3.1	Comparison between threshold concentrations of ADF for induction of biphasic aggregation in citrated and heparinized(1-2u/ml) PRP.	82
3.2	Comparison of lowest concentration of agonist causing in vitro aggregation with reported physiological and pathological levels in circulating blood.	91
4.1 a	Aggregation and TXB ₂ levels asociated with adrenaline potentiation of arachidonate-induced aggregation in the presence or absence of aspirin.	112
5.1	The ability of bromophenacyl bromide to inhibit aggregation due to adrenaline, ADP and potassium arachidonate.	131
6.1	Phospholipid analysis by phosphorus content for unstimulated and phospholipase C-treated platelets	.158
6.2	Changes in PC and PA due to exogenous phospholip- ase C(PLC) in the presence of BPB and/or verapamil	.160
6.3	Changes in PC and PA due to exogenous phospholip- ase in the presence of trimetoquinol of spermine.	162
6.4	Changes in PC, PI and PA due to thrombin in the presence of BPB and/or verapamil.	163
6.5	Changes in PI and PA due to thrombin in the presence of trimetoquinol or spermine.	164
7.1	Phospholipid content of unstimulated and thrombin- stimulated platelets.	176
7.2	Phospholipid changes during adrenaline potent- iation of ADP-induced aggregation in the presence or absence of BPB.	182

xiii

ABBREVIATIONS

ACD	acid	citrate	dextrose
-----	------	---------	----------

- ADP adenosine diphosphate
- ASA acetylsalicylic acid
- ATP adenosine triphosphate
- BHT butylated hydroxytoluene
- BPB bromophenacyl bromide
- Bt₂c'AMP dibutryl cyclic adenosine monophosphate
- betaTG beta-thromboglobulin
- BW755C 3-amino-1-[M-(trifluoromethyl)-phenyl]-2-pyrazoline
- Ca²⁺ calcium ions
- c'AMP cyclic adenosine monophosphate
- c'GMP cyclic guanine monophosphate
- DMSO dimethylsulphoxide
- HETE 12-hydroxyeicosatrienoic acid
- HPETE 12-hydroperoxyeicosatrienoic acid
- 5HT 5-hydroxytryptamine (serotonin)
- KA potassium arachidonate
- NSB non-specific binding
- PA phosphatidic acid
- PAF platelet activating factor
- PC phosphatidylcoline
- PDGF platelet-derived growth factor
- PE phosphatidylethanolamine
- PF4 platelet factor 4

PGE_1, D_2	prostaglandins E1,D2
PGI2	prostacyclin
PI	phosphatidylinositol
PPP	platelet-poor plasma
PRP	platelet-rich plasma
PS	phosphatidylserine
RD	reagent diluent
SD	standard deviation
SG	silica gel
SPH	sphingomyelin
TPCK	tosylphenylalaninechloromethylketone
TXA2.B2	thromboxane A2,B2

TMB-8 8-(N,N-diethylamino)-octyl-3,4,5-trimethoxy-benzoate

CHAPTER 1

GENERAL INTRODUCTION

With the use of entitiongulants, platelets were more easily obterved. Lintertan (1940) repeated the work of Sizon (1842) to establish the presence of colourless bodies, 2-3s in size but throught they tune from the lymphetic system and ware the origin of the red call.

In 1850, whatton Jones described colouriess corpuscies incorporated in a mass which adhered to blood wessel walls and blocked flow of blood in the web of a frog's foot. Through the remainder of the airersenth ceatury there were neveral more developments concerning the role of platelets in elec formation. Only (1874) fillustrated single

1.1 HISTORICAL BACKGROUND

The study of blood platelets commenced over one hundred years ago. In the early nineteenth century, as microscopes improved, so did the observations of blood components. In 1842, Donné reported smaller 'globules' than the red and white elements, although he erroneously placed their origins in the lymph. At about the same time, two independent English medical practitioners, Gulliver and Addison noted minute spherules in the blood. Both these men also reported the association of these particles with fibrin, and made the first drawings of a platelet-fibrin clot. There was still confusion over the existence of platelets for several years however, with the result that only red and white cells were investigated for involvement in coagulation and inflammation.

With the use of anticoagulants, platelets were more easily observed. Zimmerman (1846) repeated the work of Simon (1842) to establish the presence of colourless bodies, 2-5µ in size but thought they came from the lymphatic system and were the origin of the red cell.

In 1850, Wharton Jones described colourless corpuscles incorporated in a mass which adhered to blood vessel walls and blocked flow of blood in the web of a frog's foot. Through the remainder of the nineteenth century there were several more developments concerning the role of platelets in clot formation. Osler (1874) illustrated single elements in the circulating blood which had the capacity to

aggregate when the blood escaped from the blood vessel. Hayem (1878) added another function for platelets, namely 'regenerating' the blood after haemorrhage. Bizzozero (1882) seems to be the first to have observed circulating platelets in living animals, and demonstrated conclusively that the white portion of thrombi consisted almost exclusively of platelets, then called "petites plaques". These observations were later confirmed and extended by Eberth and Schimmelbusch (1885) and Laker (1889).

In 1906, Wright determined the genetic relationship of the platelet to the megakaryocyte, denying all theories of platelets originating from the white cell or as an early stage of the red cell. There was still doubt about the validity of these early observations, however, until the mid-twentieth century.

Even before the participation of platelets in thrombus or haemostatic plug formation was established, mural thrombi were thought to be involved in the development of atherosclerosis. In 1852, von Rokitansky proposed that atheroma resulted from the degeneration of a deposit of blood products, largely fibrin. Virchow (1856) and later, Hogerstedt and Nemser (1897) proposed that thrombi might contribute to the development of atherosclerosis, a theory revived in 1936 by Clark, Graef and Chasis and by Duguid in 1948. A large part of current work on platelets is directed towards determining the mechanism whereby platelets and thrombi contribute to the inception and progression of atherosclerosis.

1.2 PLATELET STRUCTURE

The platelet is normally discoid, 2-4µ in diameter, and is formed by the fragmentation of megakaryocyte cytoplasm. The circulating platelet count in man is about 2x10⁸/ml with a turnover rate of about 3.5x10⁷/ml/day (Harker and Finch, 1969). Its structure is far from simple.

1.2.1 The Platelet Surface

The platelet surface consists of the plasmatic atmosphere, an ill-defined cloud of plasma proteins, and the plasma membrane proper which resembles that of other cells. This membrane possesses several exceptional features enabling the platelet to carry out its unique functions. The plasmatic atmosphere has not been well characterized but it has been proposed as the cause of the observed differences in behaviour between platelets in plasma and washed platelets in <u>in vitro</u> studies(Ardlie and Han, 1974).

The glycoproteins of the outermost membrane layer, the glycocalyx, contribute largely to the responsiveness of the platelet. While more than 30 electrophoretically different glycoprotein bands have been identified (Berndt and Phillips, 1981a) only five have become the subject of detailed investigation. The glycoproteins Ib, IIb, III and IV are present in particularly high concentrations. Several lines of evidence are consistent with glycoprotein Ib/glycocalicin being the thrombin receptor on human platelets (Okumura and Jamieson, 1976; Okumura <u>et al</u>., 1978; Ganguly, 1977; Ganguly and Gould, 1979). There is considerable evidence that glycoproteins IIb and III are subunits of a single glycoprotein, and together may participate in the aggregation process (Phillips <u>et al</u>., 1980). No biological function has yet been ascribed to glycoprotein IV. The distinguishing feature of the single chain polypeptide, glycoprotein V is that it is the sole thrombin substrate on the surface of intact, washed human platelets (Berndt and Phillips, 1981a,b). Purified glycoprotein V behaves as a thrombin substrate equally with that on intact platelets (Berndt and Phillips, 1981b,c).

The plasma membrane is similar to other cell plasma membranes in having an apparent trilaminar structure of protein and lipid. It differs from other cells in possessing many invaginations and 'knobs' (Feagler <u>et al.</u>, 1974) as seen in freeze-etched fixed platelets. The membrane extends into the open canalicular system or surface connecting system and the inner surface of the membrane contains numerous fibrillar structures.

Compositional analysis of the isolated plasma membrane indicates it is composed of 52% lipid, 36% protein and 7% carbohydrate which can be divided into two lipoglycoprotein subfractions of two different densities (Barber and Jamieson, 1970). The 52% lipid is further divided into 42% phospholipid and 10% cholesterol. The carbohydrate consists of glucose, galactose, mannose, hexosamine, sialic acid and fucose. The sialic acid contributes to the negative charge on the platelet surface (Seaman and Vassar,

1966) and the removal of terminal sialic acid residues from membrane glycoproteins and glycolipids by neuraminidase slightly enhances platelet responses to aggregating agents (Greenberg <u>et al.</u>, 1975).

Phospholipids of platelet membranes include: sphingomyelin(SM), phosphatidylcholine(PC), phosphatidylethanolamine(PE), phosphatidylserine(PS) and phosphatidylinositol(PI). Platelet membranes(including outer plasma membrane and membrane from intracellular structures such as dense tubular system) contain, relative to other phospholipids, more PI and PC than does the granule fraction (Marcus et al., 1969). Only minor amounts of PS and PE are present at the exterior of the plasma membrane, but platelet activation may increase accessibility of PE(Schick et al., 1976; Rittenhouse-Simmons et al., 1976). Exposure of phophatidylserine by collagen or thrombin increases the procoagulant activity of the platelet (Bevers et al., 1983) and enhances the conversion prothrombin to thrombin. Perret et al. (1979) published the proportions of the phospholipids in the exterior of the outer membrane to be: PC 23%, PE 11%, PI 10%, PS 4.5% and SM 70%, using enzymes which hydrolyzed phospholipids without causing lysis of the platelets. Phospholipid metabolism and its role in platelet function will be discussed later (1.5.4).

Procoagulant activities of platelets, reside in the membrane. One of these activities has been termed platelet factor 3, and this activity resides in phospholipids(mainly

phosphatidylserine) on the inside leaf of the bilayer membrane(Hemker <u>et al</u>.,1983). Because platelet lipids have this role in blood coagulation, and because phospholipids have been shown to participate in the aggregation and secretion process, platelet lipids are an important subject of research in the platelet field. Platelet lipids are indeed closely linked to platelet reactivity. The role of arachidonic acid metabolism in platelet activation, in particular, will be discussed (1.5.3).

Platelet cholesterol content reflects the initial composition of the megakaryocyte from which platelets are derived, but there can also be some exchange with lipoprotein-cholesterol in the platelet environment (Derksen and Cohen, 1973). Platelets cannot synthesize cholesterol. Carvalho et al. (1973) found that patients with Type II hyperbetalipoproteinemia have enhanced platelet responses to adrenaline, collagen and ADP but the mechanism for this cholesterol influence has not been established. Shattil et al. (1975) found that incubation of platelets with cholesterol-rich liposomes raised the membrane cholesterol level and increased 35-fold the sensitivity of platelets to adrenaline and ADP. Correspondingly, a reduction in membrane cholesterol was associated with an 18-fold reduction in platelet sensitivity to adrenaline.

Of further interest are the findings of Hassall <u>et</u> <u>al</u>.(1983) who established a correlation between low density lipoproteins and total cholesterol concentrations with

sensitivity to aggregation by adrenaline but no other agonists. A possible mechanism for the influence of cholesterol has been suggested by Kramer <u>et al</u>.(1982) who found that elevated platelet cholesterol content affected thrombin-induced changes in platelet phospholipids with an increased metabolism of [³H]arachidonic acid from phosphatidylcholine. It is proposed that cholesterol affects lipid organization and thus affects phospholipase activity for the liberation of arachidonic acid.

The role of platelet membrane glycolipids in platelet function is unknown. Gangliosides and ceramides have only been the subject of limited experimentation to date. 1.2.2. Microfilaments and Microtubules

The fibre system of the platelets in the cytoplasm supports the discoid shape of unstimulated platelets. Electron micrographs show the circumferential band of microtubules as marginal bundles with long straight sections, possibly under tension (Behnke, 1965; Haydon and Taylor, 1965). The bundle of microtubules never directly contacts the cell wall of the platelet but forces necessary to maintain the discoid shape can be transmitted through submembrane filaments (White, 1972a). The discoid shape is also regulated by intrinsic properties of the surface membrane, by cyclic nucleotides, and by the maintenance of high levels of ATP as will be discussed later (1.5.1).

Platelet stimulation inducing a shape change, is accompanied by a contraction of the ring of microtubules and microfilaments around tightly packed central organelles

(White, 1968). In addition, the dense tubular system is a calcium sequestering site in platelets, resembling the sarcotubules of muscle cells. The channels of the open canalicular system, with their specialized microfilaments, are similar to the transverse tubules of muscle (White, 1972b). The platelet therefore has all the contractile elements necessary to undergo shape change when required. The microtubules and microfilaments are also involved in the release reaction of platelets (MacIntyre, 1976).

1.2.3 Platelet Organelles

The cytoplasm of platelets also contains mitochondria, a-granules, dense bodies, lysosomes, occasional Golgi apparatus and glycogen granules (White and Gerrard, 1978). The number of mitochondria in platelets, however, are comparatively low.

Dense bodies contain serotonin, ADP, calcium and phosphorus; the last two constituents contribute to the electron dense properties of dense bodies (Skaer, 1975). The phosphorus is only partially accounted for by ATP and ADP. Dense bodies also contain large amounts of phospholipid, probably rich in arachidonic acid (Skaer <u>et</u> <u>al</u>., 1976). This was suggested by the finding of arachidonic acid by autoradiography in a structure regarded as a stage in the discharge of dense bodies (Skaer and Skaer, 1979). Some of the dense body contents promote growth of platelet aggregates, and are released from the dense bodies into the platelet environment. Skaer (1981) proposed a mechanism involving movement of dense bodies to

the surface of the platelet where they distort the membrane, fusing with it and then enlarging. The resultant spheroid then releases its contents, leaving behind an empty shell.

The α -granules also participate in the release reaction. Holme <u>et al</u>. (1973) observed the presence of fibrin, and White (1972b) observed nucleoids which originated from the α -granules, in the canalicular system and concluded that fusion of the α -granule membranes with the inner surface of the open canalicular system occurs, allowing the contents to diffuse out. The contents include fibrinogen, platelet factor 4 (anti-heparin factor), betathromboglobulin and platelet derived growth factor(PDGF).

Lysosomes contain enzymes including cathepsins, glycosidases, a chemotactic factor for leucocytes and a cationic protein which increases vascular permeability and induces histamine release from mast cells. Lysosomal degranulation is a fairly slow process(Holmsen <u>et al</u>.,1969) and the mechanism is unclear. These diverse inclusions of the platelet and its complex structure enable this cell to have many functions.

1.3 PLATELET FUNCTION

The smallness of the platelet with respect to the other blood cells should not lead to an underestimation of its importance.

1.3.1 Maintenance of Integrity of the Vasculature

Gimbrone <u>et al</u>. (1969) found that capillary integrity, in isolated perfused organs, was better preserved when the perfusate contained intact viable platelets in addition to plasma proteins and the necessary ions. Thrombocytopenia is associated with an increase in capillary fragility. Evidence for the close association of platelets with endothelial cells lining capillaries was obtained by Johnson (1971) in transfusion studies in thrombocytopenic animals. She found that the transfusion of radioactively labelled platelets labelled the capillaries of many organs whereas labelled platelet-poor plasma did not. Ultrastructural studies confirmed the association but could not explain the interaction of platelets with endothelial cells.

The endothelial lining depends on circulating platelets for sustained normal functioning, while the endothelium protects platelets from activation and subsequent aggregation. Tissue culture studies of endothelial cells indicate that a platelet factor promotes replacement of a partial removal of endothelial cells from monolayers. Migration of endothelial cells into the damaged area is enhanced by platelet-rich serum (Wall et al., 1978) though smooth muscle cell migration is more enhanced by platelet factors (Thorgeirsson <u>et al</u>., 1979). Mitogenic responses, as determined by tritiated thymidine incorporation were increased by platelet-rich serum added to pre-confluent monolayers of endothelial cells (Fratkin

et al., 1980). On the other hand, confluent monolayers did not respond in agreement with the work of Thorgiersson and Robertson (1978). Smooth muscle cells again differed from endothelial cells in that they exhibited a mitogenic response whilst confluent. The mitogenic factor found in serum is the platelet derived growth factor from the α granules (Antoniades <u>et al</u>., 1975; Busch <u>et al</u>., 1976; Antoniades and Scher, 1977). Platelets probably assist in the repair of injured blood vessels by regulating endothelial cell and smooth muscle cell growth. This facility forms the basis of the response to injury hypothesis for atherogenesis(see Section 1.3.4).

1.3.2 Inflammation

Platelets accumulate at sites of tissue damage and aid in the inflammatory response by increasing vascular permeability and by being chemotactic for leucocytes. Nachman <u>et al.</u> (1972) found a permeability factor in the lysosomes associated with histamine release. Chemotaxis of leukocytes could be brought about by several possible candidates. Weksler and Coupal (1973) found that a cationic protein from the lysosomes liberated a chemotactic factor from the fifth component of complement. The lipoxygenase product, 12L-hydroxy-5,8,10,14eicosatetraenoic acid (HETE) was shown by Turner <u>et al</u>. (1975) to induce leukocyte chemotaxis and Kitchen <u>et al</u>. (1978) showed some chemotactic activity of thromboxane B_2 (TXB₂).

Through the action of proteolytic enzymes, platelets also have the potential to participate in the destruction of connective tissue in the chronic inflammatory disease state (Zucker, 1974). Thrombin releases a considerable proportion of platelet β -gluceronidase, β -Nacetylglucosaminidase and cathepsin, hydrolytic enzymes of lysosomes. Further, antigen-antibody complexes interact with the platelet Fc-receptor and thereby induce the release reaction (Humphrey and Jacques, 1955).

1.3.3 Haemostasis

The most important role of platelets is in the haemostatic process. Platelets initiate the formation of a haemostatic plug by accumulating at the site of vascular injury. These adherent and stimulated platelets then release factors (eg. ADP, TXA₂) promoting aggregation and coagulation. The membrane of the platelet also acts as a cofactor in the clotting cascade.

In order to plug a wound in a blood vessel wall, the platelets have to adhere to the tissue of the wound. Artificial surfaces, (Friedman <u>et al</u>., 1970) and natural vascular surfaces (Baumgartner, 1973) both have the capacity to cause adherence of platelets. Removal of the endothelium of the blood vessel wall exposes the subendothelium collagenous layer. In animal studies, where the endothelial layer is removed artificially, platelets adhere within minutes (Baumgartner and Spaet, 1970; Stemerman and Ross, 1972; Nam <u>et al</u>., 1973). Electron micrographs (Baumgartner et al., 1967; Baumgartner, 1977) illustrate the adhesion of platelets to the subendothelium.

Degranulation of adherent platelets brings about the formation of a platelet plug by the process of platelet aggregation. An electron micrograph taken 10 minutes after removal of the rabbit endothelium shows platelets which have undergone a shape change. Pseudopodia interlink platelets and those platelets closest to the subendothelial surface show a loss of alpha granules and serotonin storage granules. After forty minutes, all that remains is a thin covering of flattened, degranulated platelets.

Exposed subendothelium also activates coagulation factor XII. The intrinsic pathway of coagulation is further promoted by platelets at several steps when they accumulate at sites of vascular injury (Walsh, 1974). The ADP-stimulated platelet membrane surface provides a suitable surface for the activation of factor XII. The collagen-stimulated platelet membrane activates factor XI. Phospholipids normally found on the interior of the platelet membrane may be exposed by disruption of the platelet or by a specific mechanism called flip-flop where different phospholipids are swapped from inside to outside and vice versa. This mechanism is stimulated by the simultaneous presence of small amounts of collagen and thrombin (Hemker et al., 1983). Exposed platelet membrane procoagulant phospholipids bind factors IXa and VIIIa to form the factor X converting enzyme. They also provide for the interaction of factor Xa and factor Va to form

prothrombinase which forms thrombin. Fibrin formation is thus promoted and the haemostatic plug is consolidated.

White cell accumulation occurs secondarily to platelet adhesion, probably as a result of release of chemotactic substances as mentioned before. These cells could serve to phagocytize cellular debris.

When a blood vessel is intact, the endothelium may counteract platelet activity in several ways. Moncada et al.(1977) found that prostacyclin(PGI₂) production by the arterial wall was highest in the intimal region. Martin <u>et al</u>. (1980) found that PGI₂ prevents platelet aggregation <u>in vivo</u> and possibly promotes disaggregation. PGI₂ has been shown to increase c'AMP levels in platelets which inhibits aggregation (Gorman <u>et al</u>., 1977a). Use of PGI₂neutralizing antibodies by Steer <u>et al</u>. (1980) indicated that circulating levels of the hormone are insufficient to prevent <u>in vitro</u> aggregation. This may mean that PGI₂ acts in cooperation with other anti-aggregatory factors <u>in vivo</u>.

Other means by which the endothelium may be antiaggregatory are the production of plasminogen activator activity, heparin-like activity (heparan sulphate), ADP-ase activity and thrombin-binding activity. These factors in addition to the endothelium being a physical barrier between the blood and the underlying connective tissue and smooth muscle cells, make the endothelial lining an important protective organ (Nalbandian <u>et al.</u>, 1979).

1.3.4 Aberrant Haemostasis: Thrombosis and

Atherosclerosis.

Haemostasis, as summarized above, involves the interaction of platelets, the vessel wall and plasma coagulation factors. Thrombosis may be considered as an aberrant form of the haemostatic process. Atherosclerosis can be considered also as an undesirable reactive response of the blood vessel wall to injury, although there is still controversy over the mechanism of atherogenesis.

The characteristics of atherosclerosis are: localized lesions of the arterial intima which contain smooth muscle cells, fibrous tissue and acculumated lipid, associated with a narrowing of the arterial lumen (Wissler,1976). The lesion commences in the intima mainly around vessel orifices and branches. The intimal thickenings are rich in smooth muscle cells, connective tissue and proteoglycans (Geer and Haust, 1972; Strong and McGill, 1962). Within, and between cells, lipids collect, particularly low density lipoproteins (LDL) and cholesterol crystals. A fibrous cap overlies the resultant plaque.

No single theory of atherogenesis has been proven. It is, however, undisputed that advanced atherosclerotic lesions often have thrombi associated with their surfaces. Histological and morphological evidence indicates that intimal thickenings have incorporated thrombus material (Mustard and Packham, 1975). The platelet-endothelial interaction is therefore of considerable importance in the development of atherosclerosis, be it biochemical or biophysical.

The response to injury hypothesis of Ross and Glomset (1976) postulates that injury to the endothelium gives rise to events which, if they occur repeatedly, cause an atherosclerotic lesion. The injury to the endothelium may take the form of desquamation, lysis or changes in the permeability characteristics of the cells. There could then be a focal adherence of platelets at the injured site, followed by release of platelet granule contents including PDGF which enhances proliferation of arterial smooth muscle cells(Section 1.3.1). If there is also platelet aggregation, there would be even further release. This hypothesis is not entirely new, but rather a modification of that documented by Duguid (1949) who followed the line of thinking from several workers as previously mentioned (Section 1.1).

According to the response to injury hypothesis, PDGF provides a stimulus for focal migration and proliferation of smooth muscle cells. These cells may be derived either from the media of the artery wall or from pre-existing cells within the intima. These proliferative cells are stimulated to also synthesize connective tissue, including collagen, elastic fibre proteins and glycosaminoglycans. Lipids accumulate both inside and outside cells, particularly if plasma lipoprotein levels are raised, or the endothelium is damaged.

An important feature of the response to injury hypothesis is that the 'injury' must occur on a chronic or recurrent basis. The response to single injury is a normal

haemostatic response, such lesions regressing and disappearing once repair is complete (Ross and Harker, 1976). Repetition of such injuries could conceivably occur in the presence of several known risk factors for coronary heart disease. Major risk factors include smoking, hypertension and hypercholesterolaemia.

Hypertension could lead to chronic damage of endothelial cells at bifurcations of arteries where also, platelet reactivity may be increased. Lipoproteins have been shown to have some ability to stimulate proliferation of smooth muscle cells (Fisher-Dzoga et al., 1974) and to inhibit PGI₂ production (Nordoy et al., 1978). Lipoproteins are also incorporated into the atherosclerotic plaque, the consequences of which are uncertain. Hypercholesterolaemia was shown by Ross and Harker(1976) to decrease platelet survival, probably by loss of endothelium sustained by chronic hypercholesterolaemia. Also, platelet reactivity is increased by increased cholesterol incorporation (Shattil et al., 1975) and by increased saturated fats in the diet (Jakubowski and Ardlie, 1978). Smoking, which is another risk factor, has been found to affect the haemostatic system in several and complex ways. For example, it increases the sensitivity of platelets to ADP (Hawkins, 1972). Aryl hydrocarbons (premutagens) from cigarettes are taken up and carried by the same fractions in the plasma as cholesterol, namely the lipoproteins (Benditt, 1976). This would provide a mechanism for introduction to the vessel wall of a mutagen responsible

for induction of proliferation. Also, Mustard and Murphy (1963) proposed that cigarette smoke damages the endothelium.

Another hypothesis of major interest in atherogenesis is the monoclonal hypothesis. Benditt and Benditt (1973) discovered that the cellular composition of atherosclerotic plaques is monoclonal, with only a small percentage polyclonal, as characterized by enzyme analysis. Similar findings have been described by Pearson <u>et al</u>. (1977). According to this hypothesis atherosclerosis is initiated by a mutation caused by chemicals, possibly cigarette smoke, radiation or viruses. The cells with a selective advantage then proliferate. Further development of the disease can occur when platelets then form thrombi on these thickened regions. This hypothesis also proposes 'injury' as an initiator and includes platelet involvement in its development.

The earlier theories concentrated on the lipid nature of the lesions, rather than their cellular nature. In most of the current theories, lipid accumulation is regarded more as a secondary event. More emphasis is placed on the origin of cellular proliferative stimuli. Research is concentrated on the possible biochemical alterations in the endothelium and the stimuli responsible for modification of the smooth muscle cells. The importance of platelets in atherogenesis lies in their relationship to endothelial and smooth muscle cell behaviour.

The work reported in this thesis concentrates on the mechanisms of platelet activation keeping in mind the involvement of platelets in thrombosis and atherosclerosis.

1.4 PLATELET INTERACTIONS

The studies of platelet interactions with various blood and vessel wall components, have been mostly carried out, for convenience, <u>in vitro</u>. Biogenic amines, thrombin, connective tissue and ADP are <u>in vivo</u> compounds which are known to interact with platelets.

1.4.1 Catecholamines

Baseline plasma levels of catecholamines have been estimated at 3.4nM for noradrenaline and 0.64nM for adrenaline (Dimsdale and Moss, 1980). During public speaking, adrenaline levels increase twofold, and during moderate physical excercise, noradrenaline levels increase threefold (Dimsdale and Moss, 1980). Roizen <u>et al</u>. (1975) found catecholamine levels to change dramatically with changes in body temperature in rats. Catecholamine levels are capable of changing rapidly to meet acute body reactions to environmental and activity changes.

Catecholamines accumulate in pig platelets (Born <u>et al.</u>, 1958) against a concentration gradient, but the affinity of this transport system is unknown. Interpretation of data on the subject is difficult. Experiments performed on plasma without pH control may have artificially affected uptake results. Similarly, data on

washed platelets for accumulation of catecholamines may be artifactual.

The stimulation of platelets by catecholamines produces a series of reactions. It is difficult to separate pure catecholamine responses from those which may be induced via secondary stimulation by, for example, ADP. Adrenaline has been shown, however, to interact with its own specific cellular receptor (Scrutton and Wallis, 1981), whereas released ADP may mediate secondary aggregation. Studies using selective agonists and antagonists have demonstrated that stimulation of an α -adrenoceptor is responsible for initiation of aggregation and secretory responses of human platelets to adrenaline. Radioligand and physiological response studies suggest that both the aggregatory and pro-aggregatory responses of human platelets to adrenaline are mediated by α_2 -adrenoceptors (Grant and Scrutton, 1980b; Clare et al., 1984). More recent studies, however, suggest that the present understanding of the relationship between receptor occupation and functional responses of platelets may be incorrect (Swart et al., 1984). Although inhibition of adenylate cyclase is widely accepted as the typical stimulus-response coupling mechanism for α_2 -adrenoceptors, this cannot explain induction of platelet aggregation by adrenaline and therefore unique α_2 -adrenoceptors may be involved in the aggregatory response to adrenaline (Clare et al., 1984).

There has been some disagreement as to whether adrenaline causes platelets to change shape. Mills (1973)
based on observations on ATP consumption in combination with aggregation tracings, found no shape change-associated decrease in the platelet adenylate energy charge and concluded that adrenaline does not induce a shape change in human platelets. Patscheke (1980) observed an adrenaline shape change in recordings for washed platelets suspended in a medium containing 1mM calcium. The so-called shape change due to adrenaline certainly differs from that caused by ADP.

Adrenaline, when capable of causing aggregation in vitro, induces two waves of aggregation in human platelets: a primary and a secondary response. The secondary response is associated with the release reaction and a minimal amount of thromboxane production. Arachidonic acid, released from phospholipids, is metabolized via the endoperoxides to thromboxanes and several other products. Thromboxane A_2 is a potent aggregating compound. Arachidonic acid is also metabolized by the enzyme lipoxygenase but the roles for the products of this pathway are uncertain. There are several questions so far unanswered in regard to the mechanism of induction of these responses to adrenaline, and the connections between the two phases.

One point of confusion arises from the choice of anticoagulant for blood collection for the preparation of the platelets. Primary aggregation occurs in both citrated and heparinized plasma, while it is increased in the latter (O'Brien et al., 1969). Secondary aggregation is the

predominant response to adrenaline in citrated plasma, and is dependent on arachidonic acid metabolism by cyclooxygenase to form thromboxane A_2 (Best <u>et al</u>., 1980). Both secretion of serotonin (5-hydroxytryptamine) and thromboxane $B_2(TXB_2)$, the stable end-product of TXA_2 , are inhibited by 100 μ M acetylsalicylic acid (aspirin) which irreversibly acetylates cyclooxygenase. The second phase of the adrenaline response is therefore totally abolished by aspirin while the small primary response remains unaffected.

It is not known whether this secondary aggregation is important in vivo as it could be purely an artifact of citrated plasma (see 2.1.1.2.). (See Chapter 3 for a comparison of platelet reactivity in heparin vs. citrate.) The involvement of adrenaline and noradrenaline in the reactivity of platelets and thus thrombosis and atherosclerosis is an important area of investigation. It is possible that the major importance of adrenaline lies in its preparation of platelets to respond to other agonists. Adrenaline and noradrenaline are known to potentiate the aggregation response to other agonists in vitro. Ardlie et al. (1966) first demonstrated this and it was studied in detail by Mills and Roberts (1967). In particular they studied adrenaline potentiation of the response to nucleotides. Since then, several authors have observed adrenaline potentiation of the response to thrombin, collagen, serotonin and vasopressin (Baumgartner and Born, 1968; Thomas, 1967; Grant and Scrutton, 1980a).

Of further interest is the inability of aspirin to prevent adrenaline potentiaton of ADP-induced aggregation. (Rao et al., 1980a). The alteration of the platelet involved in potentiation is at the receptor-membrane level (Rao et al., 1980b; Rao et al., 1981a). It was thought that the effects may be due to changes in calcium mobilization because adrenaline potentiation was inhibited by verapamil, a calcium 'antagonist' (Owen and Le Breton, 1980). It has subsequently been shown, however, that verapamil is an a-adrenergic antagonist(Barnathan et al., 1982) with no receptor sub-type specificity as studied in other tissues (Motulsky et al., 1983). Adrenaline and noradrenaline depress the basal activity of adenylate cyclase in human platelet lysates (Moskowitz et al., 1971; Salzman and Levine, 1971) and lower prostaglandin E_1 induced increases in cyclic AMP in intact platelets (Harwood et al., 1972; Haslam and Taylor, 1971; Robinson et al., 1969) but adrenaline does not depress the level of cyclic AMP in resting platelets nor can the adrenaline response be enhanced by addition of an inhibitor of adenylate cyclase (Haslam et al., 1978a,b). An alternative mechanism of adrenaline potentiation in vitro must be sought.

Adrenaline is also of interest because platelets from human donors exhibit wide variability in their response to this agonist. Sensitivity to adrenaline could conceivably be related to thrombotic tendencies, but such a correlation needs to be established. There is also a need to establish

the physiological role of adrenaline in platelet reactivity. Detailed reviews of recent work on the mechanisms of adrenaline action will be given in later chapters, appropriate to the particular study undertaken. See also Section 1.5 for discussion on calcium,c'AMP and phospholipids in relation to mechanisms of platelet responses.

1.4.2. Serotonin (5-hydroxytryptamine, 5HT)

Serotonin is actively and passively taken up by platelets and stored in the dense granules, where it is inaccessible to the destructive enzyme monoamine oxidase, im preparation for release (Paasonen, 1965). These granules also contain ADP, ATP, magnesium, calcium and other biogenic amines. The interaction of platelets <u>in</u> <u>vitro</u> with serotonin commences with a shape change, first observed by O'Brien and Heywood (1966) although aggregation is not necessarily a consequence of this event.

Human platelets, prepared in either citrate or heparin, can be aggregated by micromolar concentrations of serotonin (O'Brien <u>et al</u>., 1969). This aggregation is generally small and reversible, and high concentrations are self-inhibitory (Baumgartner and Born, 1968). Variations in ionized calcium levels can, under certain conditions have dramatic effects on the platelet responsiveness to serotonin. Addition of calcium to citrated human platelets increases the initial velocity of serotonin-induced aggregation but does not affect the maximal response

(O'Brien et al., 1969). Aggregation to serotonin is calcium dependent to this extent.

A small percentage of human donors is found to exhibit a secondary response to serotonin with an associated release of adenine nucleotides. Some donors have serotonin -insensitive platelets.

Serotonin, like adrenaline, can potentiate the platelet response to other agonists (Baumgartner and Born, 1968). There are serotonin receptors on the platelet of heterogeneous types, either related to shape change or serotonin uptake (Drummond, 1976).

1.4.3. Thrombin

Since only a few platelets would be in contact with collagen at a wound surface, other agents are expected to be involved in consolidating a platelet mass in the formation of a haemostatic plug. Thrombin is formed at the sites of injury in blood vessels. Thrombin is therefore of interest in in vitro studies.

The effects of thrombin on platelets include an increase in phosphatidylinositol (PI) turnover and an increase in protein phosphorylation (Lyons <u>et al.</u>, 1975). Diacylglycerol production from PI hydrolysis has been reported to be coupled to a protein kinase thought to participate in the release reaction through phosphorylation of a 40,000dalton M.W. polypeptide (Kawahara <u>et al.</u>, 1980). Breakdown of phosphatidylinositol by phospholipase C due to thrombin stimulation, is associated with arachidonate release (Rittenhouse-Simmons, 1979; Bell et al., 1979). If the metabolism of this arachidonic acid is blocked by indomethacin, aggregation induced by thrombin still occurs but with an increased production of phosphatidic acid, an intermediary in the phosphatidylinositol cycle (Lapetina <u>et al.</u>, 1978). Indomethacin also inhibits diglyceride lipase responsible for arachidonic acid release (Rittenhouse-Simmons, 1980). Thrombin also stimulates arachidonic acid release from phosphatidylcholine via the activity of phospholipase A_2 (Bills <u>et al</u>., 1977) so that there are two potential sources of arachidonic acid release due to thrombin. Prostaglandin synthesis occurs subsequent to thrombin activation of arachidonate release (Hamberg et al., 1974).

Thrombin inhibits adenylate cyclase by three concurrent mechanisms: by interaction with its receptor, by mediation via thromboxanes or endoperoxides and by an increase in the level of free calcium in the cytosol(Feinstein <u>et al</u>., 1981). Thrombin binding to its receptor behaves more like a hormone than an enzyme, expressing negative cooperativity for further binding (Majerus <u>et al</u>., 1976). These highly specific receptors bind thrombin with high affinity, ensuring platelet involvement in clot formation. In addition, some thrombininduced responses, aggregation and dense granule secretion do not require the continuing presence of thrombin on its receptor (Holmsen <u>et al</u>., 1981).

1.4.4. Connective Tissue

A wound in a blood vessel exposes tissues not normally in contact with the blood. Platelets have been found to adhere to connective tissue underlying the endothelium, the first step in haemostatic plug formation. The major components of the connective tissue are collagen, elastin, glycoprotein and proteoglycan. Collagen achieves aggregation of platelets <u>in vitro</u> while elastin has been found not to induce aggregation (Hoffman, 1975). Glycosaminoglycans may enhance aggregation (Muir and Mustard, 1968). The enormous amount of work on plateletcollagen adhesion and aggregation will not be included in this introduction.

1.4.5. Adenine Nucleotides

Adenosine diphosphate (ADP) and adenosine triphosphate (ATP) are stored within the platelet in dense granules for release during the release reaction. ADP and ATP located elsewhere in mitochondria, the cytosol and membranes are available for metabolic processes. Platelets challenged with radioactive adenosine, exhibit the label in this metabolic pool, while no label is found in the releasable nucleotides (Holmsen <u>et al</u>., 1969).

There is a critical level of metabolic ATP required for the platelet shape change, aggregation and the release reaction (Holmsen <u>et al</u>., 1974; Holmsen, 1975). Stimulated platelets have an increase in the activity of key enzymes of glycolysis and oxidative phosphorylation as energy is consumed.

The release of nucleotides serves to consolidate a thrombus, as new platelets are challenged and activated by ADP. Studies using ADP <u>in vitro</u> show that platelets undergo a shape change almost immediately in response to ADP (Born, 1970) and in the absence of external calcium. The platelets are then able to aggregate, with the release reaction occuring secondarily.

Little is known of the mechanism of induction of shape change, aggregation and release due to ADP. <u>In vitro</u> studies illustrate that the shape change is immediate and rapid, with a maximum change occurring in less than thirty seconds. The contractile elements of the platelet immediately beneath the plasma membrane can be rapidly activated following calcium loss from the dense tubular system or other membranes (Behnke, 1970). The work of Lloyd <u>et al</u>. (1973a,b) indicates that ADP-induced shape change is associated with phosphatidic acid production.

Calcium is mobilized during shape change, a process which includes psuedopod formation and granule centralization (Le Breton <u>et al.</u>, 1976). The calcium may control contraction in platelets through calcium-dependent myosin phosphorylation reactions (Hathaway and Adelstein, 1979) involving ATP bound to actin (Crawford, 1976). The concentric ring of microtubules becomes disorganized, but can become prominent again, extending into the pseudopodia which form during activation. The platelet is then ready for platelet-platelet adhesion.

ADP can induce the release reaction and prostaglandin production but it is not clear whether these events occur in parallel with secondary aggregation, or are the cause of secondary aggregation. Experiments with platelets from patients with storage pool deficiency, deficient in dense granule ADP, show that secondary aggregation is abolished in the absence of endogenous ADP (Weiss, 1975). Lages and Weiss (1980) reported two patients with storage pool deficiency who exhibited biphasic aggregation responses to ADP and adrenaline but these patients could also secrete ADP, despite the deficiency. These authors suggested that secondary aggregation responses may be mediated by both thromboxanes and ADP secreted from platelets.

1.5 MECHANISMS OF PLATELET RESPONSES

1.5.1. The Necessity for Calcium

Platelet activation involves a shape change, the release reaction, prostaglandin metabolism and aggregation. These steps require energy, membrane changes and contractile function, all of which have been thought to be calcium-dependent processes. The manner in which calcium is utilized is less certain.

The platelet shape change induced by ADP or the ionophore A23187 (an agent known to facilitate divalent cation transport across membranes) is associated with a redistribution of intraplatelet divalent cations, as shown by the use of a fluorescent-chlortetracycline probe (Le

Breton <u>et al</u>., 1976). The contractile mechanism involved in constricting the band of microfilaments during the shape change is activated by calcium (Hanson <u>et al</u>., 1973). Calcium is thought to participate in the disappearance of the equatorial bundle of microtubules. Calcium controls the platelet contractile protein, thrombasthenin, through its effect on ATPase activity in a similar way to muscle proteins.

Calcium ionophores also stimulate aggregation and release (White <u>et al.</u>, 1974). The regulation of cytoplasmic calcium is thought to control release of arachidonic acid from phospholipid substrates by phospholipase A_2 and thus calcium governs the generation of endoperoxides and thromboxanes from these sources (Rittenhouse-Simmons and Deykin, 1978; Gerrard <u>et al.</u>, 1978a). On the other hand, phospholipase C does not have the same requirement for calcium (Billah <u>et al.</u>, 1980). The phosphatidate specific phospholipase A_2 described by Billah <u>et al</u>. (1981) requires calcium for the release of arachidonic acid but not to the same extent as the phospholipase A_2 which degrades phosphatidylethanolamine and phosphatidylcholine.

Rittenhouse (1984) concluded that phospholipase C is not activated by calcium and suggested that stimulation is totally dependent upon a receptor-coupled event, possibly via cyclooxygenase products and ADP. The subsequent events of this calcium-independent pathway of platelet activation involves protein kinase C (Simon <u>et al</u>.,1984). Calcium may

influence the phosphatidylinositol cycle initiated by phospholipase C by inhibiting the resynthesis of phosphatidylinositol, causing the accumulation of phosphatidic acid (Lapetina <u>et al</u>., 1981b). Cyclic AMP directly opposes this action.

Dibutyryl cyclic-AMP(Bt₂c'AMP) and 8-(N,Ndiethylaminooctyl-3,4,5-trimethoxybenzoate (TMB-8) oppose arachidonate release. Bt₂c'AMP is thought to compartmentalize calcium and TMB-8 to immobilize calcium, both restricting the availability of calcium for enzyme activation, the latter's specificity albeit doubtful (Simpson <u>et al.</u>, 1984). These agents also cause inhibition of thrombin-induced release of serotonin, which can be overcome by the addition of calcium. Perhaps the enzymes of arachidonic acid conversion to thromboxanes are also calcium-dependent since they are inhibited by TMB-8 (Shaw,1981).

The work of White <u>et al</u>.(1981) demonstrated a calciumdependent calcium-binding regulator protein, calmodulin, in platelets. Localization using immunofluorescence was inconclusive, giving general staining. It has been suggested that calmodulin controls phospholipase A_2 activity, and thus arachidonate release (Rao <u>et al</u>., 1980b; Vanderhoek and Feinstein, 1979) but the phenothiazines used as a basis for these studies inhibit other enzymes (Schatzman <u>et al</u>., 1981) and may be α -receptor blocking agents (Cocks <u>et al</u>., 1981). More recent evidence indicates that calcium regulation of phospholipase A_2 is independent of calmodulin (Withnall et al., 1984).

The development of the fluorescent intracellular calcium indicator, quin 2, has provided an important tool to examine in detail the role of calcium in platelet responses to stimuli (Tsien et al., 1982). Quin 2 is added as an ester that readily penetrates platelets, is hydrolized intracellularly and binds calcium. The fluorescence of quin 2 increases when it is complexed with calcium. Studies with quin 2 show that several natural agonists, in the presence of 1mM external calcium, cause cytoplasmic free calcium to increase by both stimulated calcium entry across the plasma membrane and discharge of calcium from intracellular stores (Rink et al., 1982; Hallam et al., 1983a; Feinstein et al., 1983; Kawahara et al., 1983a; Hallam et al., 1984a; Hallam et al., 1984b). When calcium was omitted from the suspending medium, thrombin and other agonists evoke responses when the cytoplasmic free calcium remains near basal levels (Rink et al., 1982; Hallam et al., 1984a; Hallam et al., 1984b). The occurrence of shape change and secretion at basal levels of calcium has been attributed to the ability of diacylglycerol, a product of phosphoinositide hydrolysis, to initiate responses in platelets (Rink et al., 1983) (see 1.5.4). More recently, the photoprotein, aequorin, has been utilized to measure ionized calcium in platelets (Johnson et al., 1985) and this has drawn attention to some potential problems with quin 2 which nonetheless remains a useful indicator of calcium elevation in platelets.

Calcium modulation at the membrane level has been suggested as the means by which adrenaline acts as a potentiator of platelet reactions (Rao et al., 1981a). It has been proposed that calcium participates in the initiating events of adrenaline stimulation of platelets (Owen et al., 1980). With the ⁴⁵Ca-assay technique, Owen and LeBreton (1980) found that low concentrations of ADP promote the ability of adrenaline to stimulate an increase in membrane permeability to calcium. If, on the other hand, adrenaline is added to the platelets first, no increased calcium uptake is measured. These workers then proposed that some kind of calcium involvement in the potentiated ADP-response on the basis of inhibition of the response by verapamil. As has been mentioned, however, verapamil also acts as a α_2 -adrenoceptor inhibitor (Barnathan et al., 1982). (See Section 1.4.1)

Clare and Scrutton(1984) concluded that calcium uptake did occur but as a consequence of secretion due to adrenaline and other agonists. They found that blockade of secretion by aspirin prevented radioactively labelled calcium uptake. Hallam <u>et al</u>. (1983b) also found that adrenaline induces no increase in cytosolic calcium concentration in platelets treated with aspirin. On the other hand, Brass and Shattil(1982) did find that both ADP and adrenaline increased the amount of calcium bound to the platelet surface membrane, despite no net influx of calcium, and without the occurrence of secretion. Hence, there is conflicting evidence as to whether adrenaline induces an

aggregatory response by enhancing calcium influx into human platelets. The involvement of calcium in the proaggregatory response to adrenaline has not been investigated in any detail.

1.5.2. Control by c'AMP(cyclic adenosine 3',5'monophosphate)

The actions of several inhibitors of platelet aggregation and release are mediated by c'AMP (Haslam <u>et al.</u>, 1978a). c'AMP is thought to decrease the concentration of calcium ions in the platelet cytosol, probably by activation of a calcium pump. Käser-Glanzman <u>et al.(1977)</u> reported that under certain conditions, addition of c'AMP with platelet c'AMP-dependent protein kinase, can increase the initial uptake of calcium by a platelet membrane fraction containing both plasma and intracellular membranes. c'AMP promotes movement of calcium ions into the dense tubular system or across the plasma membrane or both, and this can be mimicked by dibutyryl cyclic AMP(Bt₂ c'AMP).

Both PGE_1 and PGD_2 were used in early studies to increase adenylate cyclase activity, but PGI_2 has subsequently been found to be far more potent, inhibiting platelet aggregation at nanomolar concentrations (Moncada <u>et al.</u>, 1976). As PGI_2 is produced by the endothelial lining of blood vessels, it is thought to be a major regulatory component of the haemostatic process.

Isoproterenol causes increased adenylate cyclase activity by a weak β-adrenergic effect. Platelet

aggregation is further inhibited by c'AMP phosphodiesterase inhibitors e.g. papaverine, which prevent breakdown of c'AMP (Mills <u>et al</u>., 1970).

Adenylate cyclase activity is itself modulated by released calcium (Kume and Kariya, 1980). Thus, the attenuation by ADP and adrenaline of increased c'AMP by PGE, is likely to be due, not only to receptor-mediated inhibition, but also to calcium-mediated inhibition of adenylate cyclase.

Cyclic guanosine 3',5'-monophosphate (c'GMP) has been found to increase during platelet activation by adrenaline (Jakobs <u>et al.,1974</u>) and by other agonists(Haslam <u>et al.,</u> 1978b). Kume and Kariya (1980) found c'GMP had an inhibitory effect on c'AMP-dependent phosphorylation of platelet proteins which could regulate platelet function. Takai <u>et al.(1981</u>) suggested that increased c'GMP levels, caused by thrombin stimulation, lead to feedback inhibition of phosphatidylinositol hydrolysis, presumably through protein phosphorylation.

The interrelationships of the cyclic nucleotides are obviously complex. For example, Steer and Wood (1979) suggested from their observations, that guanine nucleotides mediate either stimulation or inhibition of adenylate cyclase, depending on which binding site is involved. One of these guanine nucleotide sites is thought to interact with the α -receptor to result in enzyme inhibition, while the other interacts with the PGE1-receptor to cause stimulation. Steer and Wood (1979) also found while adrenaline and GTP individually inhibited

ARACHIDONIC ACID MELADOLE

adenylate cyclase slightly in isolated membranes, adrenaline and GTP together act synergistically to cause 50% inhibition of cyclase activity.

Membrane phospholipids may also be involved. In hamster adipocytes, phosphatidic acid, a product of phosphatidylinositol turnover, inhibited methyl xanthinestimulated c'AMP formation (Schimmel <u>et al</u>., 1980). Phosphatidic acid is formed in platelet membranes upon stimulation by thrombin (Rittenhouse-Simmons, 1979) and is perhaps a means by which thrombin inhibits adenylate cyclase. It is in this area of assigning roles to phospholipids in platelets that attention is currently focussed (See 1.5.4).

1.5.3. Prostaglandins (PG) and the Release Reaction

Secondary aggregation is associated with both the secretion of dense and α -granule constituents, including the aggregating agents 5HT and ADP, and the activation of a pathway for conversion of arachidonic acid to prostaglandins (Smith <u>et al</u>., 1973). Cyclooxygenase converts arachidonic acid to endoperoxides which form thromboxanes (Fig 1.1). The precise role of endoperoxides and thromboxanes so formed is still a matter of controversy. Thrombin and collagen may cause secretion in the absence of thromboxane synthresis whereas ADP induced secretion is dependent on thromboxane synthesis.

Best et al. (1980) concluded that the aggregation process can occur independently of secretion, depending on

ARACHIDONIC ACID METABOLISM



Figure 1.1 Arachidonic acid metabolism.

The combined activity of cyclooxygenase and thromboxane synthetase to produce endoperoxides and thromboxane A₂, a potent platelet aggregator is thought to be the most important pathway of arachidonic acid metabolism in platelets. The reactions are controlled by calcium and c'AMP: calcium (Ca²⁺) promotes the formation of thromboxanes, while c'AMP has the reverse effect, largely through their respective effects on arachidonic acid release. PG = prostaglandin HHT = heptadecatrienioc acid HETE = 12L-hydroxy-5,8,10,14-eicosatetraenoic acid \neq = inhibition. the stimulus used. Thrombin and collagen each appear to activate platelets by two possibly independent mechanisms. If this were so, this would mean that thromboxane synthesis could be stimulated coincidently with, but separately from, platelet activation by a thromboxane-independent pathway. Stimulation by ADP or adrenaline of thromboxane biosynthesis occurs as a consequence of the primary events in aggregation. The major aggregation response to adrenaline, which is of the secondary type, can be blocked by inhibitors of thromboxane production, and is therefore dependent on the small amount of thromboxane found. Meyers et al. (1979) demonstrated that secondary aggregation in cat platelets can be mediated through thromboxane formation but without participation of secreted ADP. The evidence from studies on blood from patients with storage pool disease (see 1.4.5) indicates that some ADP is necessary for the human platelet release reaction. It appears that the products of secretion may amplify aggregation, but for some agonists, an alternate pathway may obviate the need for such stimulation.

Best <u>et al</u>. (1980) also showed with the thromboxane synthetase inhibitor, 1-N-butyl imidazole, that prostaglandin endoperoxides alone contribute little toward platelet activation. Contrary to this, Flower and Cardinal (1979) found 1-N-butyl imidazole only partially effective in blocking platelet aggregation, and suggested that endoperoxides have significant activity. The advantage of endoperoxide formation as an intermediary could be in the

facility for transformation to alternative prostaglandins, depending on the enzyme stimulated (Samuelsson, 1977). Endoperoxides have been found to inhibit PGE_1 -induced increases in c'AMP but not as effectively as thromboxanes (Gorman <u>et al</u>., 1978). Through the use of inhibitors of thromboxane synthase, it appears that the effect of the endoperoxides on c'AMP levels is due to their conversion to TXA₂ (Gorman <u>et al</u>., 1977b).

Gerrard <u>et al</u>. (1978a;1978b) reported indirect evidence that endoperoxides and TXA_2 could act as intracellular calcium ionophores within platelets. Platelet activation induced by these prostaglandins is thought to be mediated by calcium mobilization from the dense tubular system (Gerrard and White, 1978) where thromboxane synthesis occurs (Gerrard <u>et al</u>., 1976). In this fashion, TXA_2 can act in direct opposition to the function of c'AMP which decreases calcium in the cytosol. Recently, however, Menashi <u>et al</u>.(1984) reported that neither thromboxanes, endoperoxides nor phosphatidic acid can act as ionophores.

Siess <u>et al</u>.(1983) suggested that endoperoxides or thromboxane A_2 are responsible for their finding that exogenous arachidonic acid stimulates the phospholipase C pathway. Formation of diacylglycerol and phosphatidic acid are important for platelet activation. Rittenhouse (1984) found also that phospholipase C activation by the ionophore A23187 was dependent on cyclooxygenase products and ADP.

Inhibition of platelet aggregation by prostaglandins

may also occur, without measurable increases in c'AMP levels, although this may represent the limit of discrimination of the assays. Sinha and Colman (1978) reported that PGE, induces an alteration or formation of an inhibitory macromolecule in plasma which can regulate platelet function independently of c'AMP. Rao et al. (1981a) used PGI, at levels which caused inhibition of arachidonate-induced aggregation without raising c'AMP levels. The inhibition was correctable with adrenaline, again independently of c'AMP. No inhibitors of adenylate cyclase were used to eliminate an effect of c'AMP. These workers were investigating the influence of catecholamines on prostaglandin receptors and concluded that stimulation of adrenergic receptors by adrenaline uncouples PG-receptor activity through close physical association of the two receptors. In experiments with washed platelets, where platelet receptors may be altered, small concentrations of PGI2 were found to stimulate aggregation due to ADP and thrombin (Jorgensen et al., 1980). PG-receptor interaction is not a straightforward system leading to adenylate cyclase activation.

1.5.4 Phospholipid Turnover

Phospholipids were initially of interest as a source of arachidonic acid for thromboxane production but phospholipid breakdown has also been associated with platelet aggregation independently of arachidonic acid conversion (Lapetina <u>et al</u>., 1978). Arachidonic acid can be released from several phospholipid classes by the action

of phospholipase A_2 , which cleaves the acyl group in the 2position. Phosphatidylcholine (PC), phosphatidylethanolamine (PE) and phosphatidylserine (PS) are the preferred substrates of phospholipase A_2 and are present in the platelet membrane (see 1.2.1) (Billah <u>et al</u>., 1980). Phosphatidylinositol (PI) is broken down specifically by phospholipase C to diacylglycerol (Rittenhouse-Simmons, 1979). By the action of a lipase, arachidonic acid can be released from diacylglycerol (Bell <u>et al</u>., 1979). Arachidonic acid release is not, however, necessary for platelet activation, and other products of membrane phospholipase activity could be important (Lapetina <u>et al</u>., 1979).

The action of thrombin on PI turnover in human platelets has been extensively studied. A recent report (Lapetina <u>et al</u>., 1981b) describes the conversion of PI to phosphatidic acid which cycles back to PI via four consecutive enzymes (Fig 1.2). Calcium controls this cycle, in that it promotes the degrading enzymes and inhibits resynthesis, resulting in an accumulation of phosphatidic acid. c'AMP increases the resynthesis resulting in a decrease in phosphatidic acid. Lapetina <u>et</u> <u>al</u>. (1981a) conclude that arachidonic acid release occurs via conversion of phosphatidic acid to lysophosphatidic acid. They do not see the action of a lipase on the diglyceride as important, but rather platelet diglyceride kinase is more active, yielding phosphatidic acid. Again these studies are largely concerned with arachidonic acid

PHOSPHATIDYLINOSITOL CYCLE



Figure 1.2 The phosphatidylinositol cycle.

This four enzyme cycle is controlled by Ca²⁺ and c'AMP. Increased Ca²⁺ levels favour the production of phosphatidic acid, while increases in local c'AMP favours the reconversion to phosphatidylinositol. Diacylglycerol occupies a significant position in being able to participate in three possible reactions. release, rather than focussing on agents responsible for an initial and cyclooxygenase-dependent reaction within the platelet.

Much other work is concerned with the debate on the relative importance of phospholipase C or phospholipase A2 as the enzyme responsible for arachidonate release. For example, McKean et al. (1981), using thin layer chromatography (TLC), measured increases in lysophosphatidylcholine in response to thrombin, indicating stimulation of phospholipase A₂ in preference to stimulation of phospholipase C. Also using TLC, Billah et al. (1980) claimed, however, that no lysophospholipid accumulated during platelet activation, but this study utilized deoxycholate treated horse platelets. Uncoupling of normal membrane structure with deoxycholate could lead to different substrate availabilities. Phospholipase activity would depend on which phospholipid was accessible as well as calcium concentration and pH. Billah et al. (1980) held the view that phospholipase C is calcium-sensitive but requires less calcium than phospholipase A₂ and could be stimulated immediately in the cytosol to act on exposed PI. Phosphatidic acid so produced could participate in the transfer of arachidonic acid to other phospholipids. There is also evidence for lysophosphatidic acids being able to act as calcium ionophores (Gerrard et al., 1979). Kannagi et al. (1980) published evidence supporting the idea that arachidonic acid needed to be mobilized within the

phospholipid classes during platelet aggregation. Rabbit platelets, after thrombin stimulation, exchanged arachidonic acid from the PE fraction to the PC fraction which is preferentially hydrolyzed. The idea of there being a store of arachidonic acid is attractive in terms of the benefits of control mechanisms for its release. Later work by Kannagi <u>et al</u>. (1981) modified their conclusions. They proposed domains of phospholipase-susceptible phospholipids, the mechanism of formation of which is unknown.

Thus far, phosphatidic acid has been postulated to act as a substrate itself for arachidonic acid release, but has not yet been given a role in prostaglandin-independent reactions. Considerable recent evidence indicates that diacylglycerol formed from PI may be a transmembrane signal for protein phosphorylation necessary for the release reaction (Kawahara <u>et al</u>., 1980). Diacylglycerol activates protein kinase C which is then capable of phosphorylating specific cellular proteins, especially those required for calcium independent reactions or those requiring very low levels of calcium (Naka <u>et al</u>., 1983). The action of activated protein kinase C, therefore, provides a possible mechanism for stimulation of release without an increase in resting cytosolic $[Ca^{2+}]$.

Michell (1975) interprets a role for PI-turnover in opening cell-surface calcium-gates based on work with guinea pig ileum smooth muscle cells. The 'gate' is an ill-defined physiological system but implies control of calcium

availability, either from calcium uptake or release from membrane calcium-chelators. PI breakdown plays a role in the coupling between receptor-agonist interaction and the opening of the gates. It is of interest to note here, also, that a_1 -receptor activation is thought to involve PI turnover and the release of bound intracellular calcium as well as the entry of extracellular calcium (Fain and Garcia-Sainz, 1980). This conclusion is based on examination of various tissues but not platelets which may not possess a_1 -receptors (see Chapter 4). It would certainly be of interest to associate catecholamine activity and PI turnover even though this would be unusual.

There are three species of phosphoinositide found in cell membranes: phosphatidylinositol and two polyphosphoinositols. The actual levels of any of these lipids may be small but their interconversion enzymes may be quite active. Broekman (1983) observed that the thrombin-induced loss of phosphatidylinositol-4,5bisphosphate in platelets coincides with a decline of membrane-found calcium as monitored by chlortetracycline fluorescence. Hydrolysis of phosphatidylinositol-4,5bisphosphate yields inositol-1,4,5-trisphosphate which seems to act by mobilizing intracellular calcium (Berridge,1984).

treated therefore, the combine : add vittes ab the second

A simple measurement of increased PI-turnover does not indicate which phospholipid products are formed, nor which would be responsible for specific activities within the platelet. A study of the action of one agonist, thrombin, does not allow for the possibility of alternate actions by other agonists on the platelet membrane. Billah and Lapetina (1982a) found that thrombin induces a rapid loss of phosphatidylinositol-4,5-bisphosphate but not phosphatidylinositol-4-monophosphate. This thrombinstimulated loss is linked to receptor activation and is insensitive to calcium mobilization as shown by independence from calcium inhibitors (Billah and Lapetina, 1982b). In contrast to thrombin, activation by the ionophore, A23187, is totally dependent on the formation of cyclooxygenase products and ADP which form as a consequence of a receptor coupled event(Rittenhouse, 1984). The addition of A23187 to human platelets induces a transient drop in the amount of phosphatidylinositol-4,5bisphosphate, a decrease in the amount of PI, and the formation of diacylglycerol and phosphatidic acid, producing lysophosphatidylinositol and free arachidonic acid.

The current understanding of the mechanism of activation of platelets via phosphatidylinositol turnover involves, therefore, the combined activities of two second messengers, diacylglycerol and inositol-1,4,5-

trisphosphate. Agonist receptor interaction induces the formation of these messengers from the hydrolysis of phosphatidylinositol-4,5-bisphosphate formed from part of the pool of PI (Berridge, 1984). Inositol-1,4,5trisphosphate mobilizes calcium which then promotes protein phosphorylation via a calmodulin-dependent kinase. Diacylglycerol, in addition to forming phosphatidic acid and free arachidonic acid, may also be involved in calciumindependent phosphorylation via protein kinase C. Either protein phosphorylation may lead independently to the platelet physiological response of shape change and secretion, but a synergistic interaction between the two may be involved in the full physiological response(Kaibuchi et al., 1983).

1.5.5. Platelet Activating Factor

Platelet activating factor (PAF), is a glycerophospholipid with the proposed structure of 1-0alkyl-2-acetyl-sn-glyceryl-3-phoshorylcholine (Demopoulos <u>et al.</u>, 1979). PAF is released from various cells, including basophils, macrophages and platelets upon immune and non-immune challenge (Vargaftig <u>et al.</u>, 1980a). It is one of the most potent aggregating substances, since concentrations as low as 1nM activate washed rabbit platelets. Plasma reduces its activity and human platelets are less sensitive.

Cazenave <u>et al</u>. (1979) studied the type of aggregation induced in rabbit platelets by PAF. They found the aggregation to be cyclooxygenase-independent and to occur

independently of serotonin release. PAF-induced aggregation can be inhibited by the prostaglandins, PGE₁ and PGI₂, possibly via c'AMP-induced reduction in available calcium ions. The strongest inhibitors of PAF-induced platelet aggregation are drugs that are known to be lipid soluble, and therefore membrane soluble. The mechanism of this inhibition is unknown and the drugs used are not specific.

PAF will induce aggregation in thrombin-degranulated platelets and in previously PAF-degranulated platelets(Cazenave et al., 1979). Thrombin can also activate PAF-degranulated platelets. A23187, thrombin and collagen cause release of PAF from rabbit platelets but ADP and arachidonic acid do not (Chignard <u>et al</u>., 1979; 1980). This indicates that PAF formation is not a general consequence of platelet activation but rather is part of a specifically induced pathway.

Kloprogge et al.(1983) found that the first wave of aggregation due to PAF is independent of cyclooxygenase products and ADP, but the second wave, accompanied by secretion was sensitive to cyclooxygenase inhibition and ADP removal. Despite several conflicting reports on the importance of platelet PAF, these authors suggest it plays only a minor role, if any, in aggregation.

Investigations of a mechanism for PAF induced aggregation have revealed that it stimulates PI turnover in human platelets (MacIntyre and Pollock, 1983), stimulates metabolism of phosphoinositides in horse platelets (Billah

and Lapetina, 1983) and that PAF promotes an early degradation of phosphatidylinositol-4,5-bisphosphate in rabbit platelets (Mauco <u>et al</u>., 1983).

1.6 AIMS OF THIS PROJECT IN RELATION TO PRESENT MODELS OF PLATELET AGGREGATION.

Mechanisms of platelet aggregation can be viewed in terms of pathways. A pathway is a series of events which lead to the same outcome in the platelet, such as aggregation or secretion. Since ADP can induce aggregation independently of thromboxane production and further release, ADP has been seen as having its own pathway for aggregation. Thromboxane mediated aggregation by collagen, thrombin, A23187 and adrenaline has been seen as another pathway. Since aggregation is still possible in the presence of both prostaglandin synthesis inhibitors and ADP degrading enzymes, a third pathway was proposed. Platelet activating factor was thought to be the mediator of this third pathway, but seems to have been excluded. Activation of protein kinase C by diacylglycerol may occur independently of both ADP and calcium. Formation of ADP and release of calcium augment the platelet response. Calcium may stimulate phosphorylation of contractile proteins and thromboxane formation which in turn may stimulate phospholipase C.

This thesis is concerned with examining further, various aspects of the enhancement of human platelet responses by adrenaline. Adrenaline may participate in platelet activation as a potentiator for other agonists.

As mentioned (1.4.1), it is possible that the importance of adrenaline, <u>in vivo</u>, is in its sensitization of platelets to respond more effectively to other agents. Adrenaline needs to be considered when studying platelet reactivity, since psychological and other forms of stress may be risk factors for coronary artery disease, and can produce raised levels of this catecholamine. For example, even in the mildly stressful situation of public speaking, adrenaline levels have been reported to rise 3-fold (Dimsdale and Moss, 1980).

The mechanism of adrenaline potentiation has been investigated by others but not yet clarified. Although adrenaline-induced secondary aggregation depends on the cyclooxygenase pathway of arachidonic acid metabolism, adrenaline potentiation of ADP- and arachidonate-induced aggregation is not prevented by cyclooxygenase inhibitors (Rao et al., 1980a) and neither is the first phase. It has been suggested, particularly with the initial stimulation by ADP, that adrenaline increases calcium uptake from plasma (Owen and Le Breton, 1980), that adrenaline releases membrane-associated calcium, or that it directly promotes deacylation of phospholipids (Rao et al., 1980b). There is, however, no evidence that adrenaline increases intracellular calcium prior to secretion, although Brass and Shattil(1982) found that adrenaline can increase the amount of calcium bound to the platelet surface membrane. With the use of the chlortetracyline fluorescent probe, Owen and Le Breton(1981) found that adrenaline induced

significant mobilization of intracellular calcium but this was reduced in the presence of EDTA. Adrenergic involvement in phospholipid turnover has been reported in other tissues (Abdel-Latif, 1974; Michell, 1975; Fain and Garcia-Sainz, 1980). The mechanism of adrenaline potentiation warrants closer study. In particular, the phospholipids of membranes which could lead to calcium ion mobilization or redistribution across membranes, independently of thromboxane synthesis are of interest in relation to adrenergic interaction.

No one has been able to associate phospholipid changes with the initiating events of adrenaline activation whereas changes have been associated with the later event, secretion (Deykin and Snyder, 1973). Adrenergic effects on phospholipid turnover are usually associated with α_1 receptor activation(Fain and Garcia-Sainz, 1980). α_1 receptors have been reported to be present on human platelets(Grant and Scrutton, 1979) but only with a frequency of 20%(Scrutton, 1981). α_1 -receptors are less effective in inducing aggregation than are α_2 receptors(Grant and Scrutton, 1979), the α_2 -receptor being the primary mediator of the proaggregatory response to adrenaline (Grant and Scrutton, 1980b).

The expected response following α_2 -receptor interaction is inhibition of adenylate cyclase(Fain and Garcia-Sainz, 1980). Primary aggregation is dependent on stimulus-response coupling following α_2 -receptor activation by adrenaline(Grant and Scrutton, 1979) but the events of

this coupling are not precisely known. No one has been able to show that adrenaline inhibits adenylate cyclase in resting, intact platelets.

ADP-induced aggregation can be potentiated by subthreshold concentrations of several agonists, including adrenaline, and this is examined in Chapter 3. ADP-induced aggregation provides a thromboxane-independent system when cyclooxygenase or thromboxane synthetase inhibitors are used. ADP does not release PAF (Chignard <u>et al</u>., 1980), but does cause an increase in membrane phosphatidic acid (Lloyd <u>et al</u>., 1972; Lapetina and Cuatrecasas, 1979). The possibility of α -adrenergic activity also leading to increased phospholipid turnover suggests a mechanism for adrenaline potentiation of ADP-induced aggregation. Adrenaline potentiation of ADP-induced aggregation was therefore studied in the presence of various inhibitors.

Cyclooxygenase inhibitors are the subject of Chapter 4. Adrenaline potentiation without arachidonic acid metabolism is compared with adrenaline potentiation when arachidonic acid metabolism is stimulated, both subsequent to ADP-activation and as a result of exogenous arachidonic acid. This aspect of the project highlights the involvement of adrenaline in non-thromboxane initiated events under conditions of potentiation, even though the usual response to threshold levels of adrenaline results in thromboxane-dependent aggregation and release. An attempt was also made to examine adrenaline potentiation at stages prior to cyclooxygenase activation by blocking postreceptor events with different types of inhibitors.

In Chapter 5 a study is reported of the use of putative inhibitors of phospholipase A_2 which inhibit the release of arachidonic acid. The aim was to observe aggregation in the absence of phospholipid changes arising from the action of phospholipase A_2 . The results suggest that adrenaline affects the phospholipase C pathway of phosphatidylinositol turnover. To explore this further, the direct effects of these inhibitors on exogenous phospholipase C were studied and compared with their effects on adrenaline potentiation (Chapter 6).

The first studies were based on observations of aggregation, release of ATP and thromboxane B_2 production. To answer questions on associated phospholipid turnover, phospholipid analysis by Whatman SG-81 paper chromatography was undertaken (Chapter 7). With this method, preliminary observations of phospholipid metabolism were made in relation to α -receptor activation.

With these combined studies a model for the mechanism of adrenaline potentiation has been examined. This model includes the interaction of adrenaline with a-receptors, and subsequent events at the membrane level involving phospholipid turnover, which lead to protein phosphorylation and calcium movement. Subsequently, thromboxane synthesis occurs and serves to amplify the original signal.

CHAPTER 2

METHODS AND MATERIALS

2.1 METHODS

2.1.1. Aggregation Studies

2.1.1.1 Advantages and Disadvantages of Studying Platelet Aggregation in vitro.

Platelet aggregation studies are performed <u>in vitro</u> since it is much easier to obtain a blood sample from a human donor than it is to examine blood and vascular tissue, interactions <u>in situ</u>. Setting up a system in the laboratory enables parameters to be defined and standardized. Several different environmental factors can be controlled simultaneously. There is the additional advantage that potentially toxic substances can be used <u>in</u> vitro.

There are, of course, several disadvantages in attempting to simulate the physiological environment. Platelets studied in isolation, are not influenced by vascular tissue which is multifactorial. Anticoagulants have to be used in the preparation of platelets from blood. These compounds very likely affect the reactivity of the platelets. Platelet responsivness alters with time once blood is removed from the circulation. Blood flow dynamics are reproduced by a simple procedure only, that of stirring magnetically.

The physical handling of platelets gives rise to major problems, particularly if washed platelets are required. Release of granular products, if not actual aggregation, occurs. Extraordinary lengths have to be taken to ensure no release of PF_4 and βTG and other products from the α granules. The washing process may also remove a necessary platelet atmosphere containing fibrinogen, thrombin and calcium. All conclusions based on <u>in vitro</u> studies should therefore be made with an awareness of these differences from the in vivo situation.

2.1.1.2 Platelet Preparation in Citrated Blood Versus Heparinized Blood.

The choice of anticoagulant best suited to aggregation studies of platelets has been a matter of controversy for some years. It has been claimed that the release reaction induced by ADP is an artifact in citrated plasma brought about by the reduction in available calcium but is independent of such a reduction in magnesium (Mustard et al., 1975; Macfarlane et al., 1975). According to their reports platelets in plasma which is anticoagulated by heparin or hirudin, or platelets washed and suspended in a medium which contains 1-2mM Ca²+ do not undergo the release reaction when they are aggregated by ADP. The concentration of calcium in the circulation is about 1mM (Hansen and Theodorsen, 1971), but the optimum concentration of calcium for aggregation is below this (Heptinstall, 1976).

EGTA (ethylene-glycol-bis (β-amino-ethyl ether) N,N'tetra-acetic acid) behaves in the same manner as citrate in its effect on aggregation due to ADP, confirming that a reduction in calcium ions promotes the release reaction (Huzoor-Akbar and Ardlie, 1978). A direct effect of
citrate is therefore not the cause of secondary aggregation. ADP-induced biphasic platelet aggregation will occur in the absence of citrate. Huzoor-Akbar and Ardlie (1976; 1977; 1978) proposed that thrombin is necessary for the release reaction which would explain the inhibitory effect of heparin and hirudin.

Recently a detailed comparison was made of the reactivity of platelets in plasma anticoagulated with citrate, heparin or hirudin (Lages and Weiss, 1981). Variable responses between different donors were observed for heparinized platelet-rich plasma, with respect to secretion, but not the extent of aggregation by ADP. Adrenaline-induced aggregation showed variability in both secretion and extent of aggregation in heparin compared to citrate. Hirudin consistently prevented the release reaction which supports earlier evidence that thrombin is involved in platelet reactions caused by ADP, adrenaline and collagen (Ardlie and Han, 1974; Huzoor-Akbar and Ardlie, 1976; 1978).

Hwang and Le Blanc (1981) proposed that heparin has aspirin-like activity in that it inhibits formation of endoperoxide metabolites. If this were so, lack of thromboxane formation and secondary aggregation could be an artifact of heparinized plasma. Heparin, added to citrated plasma inhibited thromboxane production. Collagen stimulation of thromboxane production was reduced in heparinized plasma compared to citrated plasma.

Heparin itself is reputedly an aggregating agent in some circumstances but the effects are variable and inconsistent (Wissler and Gitel, 1976). Heparin was shown to induce aggregation at 4°C and at 37°C of washed platelets derived from patients exposed to severe stress (Mims <u>et al</u>., 1977). Other authors claimed that storage at 37°C protects the platelets from aggregation due to heparin (Han and Ardlie, 1974). Heparin has also been reported to potentiate the platelet response to ADP (Thomson <u>et al</u>., 1973). Lages and Weiss (1981) could induce aggregation with 2.5U/ml to 100U/ml heparin.

Reches <u>et al</u>. (1979) showed that heparin (0.4-1.62U/ml) inhibits by 50%, PGE_1 -sensitive adenylate cyclase and antagonizes PGE_1 antiaggregating effects on platelets. These authors claim that their observations demonstrate that heparin induces a secondary phase of aggregation but neither release of serotonin, nor ATP, nor thromboxane production was investigated in their study. Their definition of secondary phase seems to depend only on the lack of reversal of aggregation.

Certainly, the anticoagulant has to be specified to enable any interpretation to be made of an aggregation study. For one of the studies in this thesis, low concentrations of heparin were chosen in order to more closely reflect the situation in circulating blood. This practice incurred several difficulties, including insufficient anticoagulation of the blood from some donors.

For other aggregation studies, where maximal platelet sensitivity was not required, and a stable system was desired for comparisons to be made between experiments, citrate was used as the anticoagulant. Although trisodium citrate (10mM) chelates calcium ions, it does not, unlike EGTA, leach calcium out of the platelets.

2.1.1.3 Preparation of Platelet-Rich and Platelet-Poor Plasma from Fresh Whole Blood.

Blood donors were healthy human volunteers who had not taken any aspirin-containing drugs in the previous fortnight. Donors taking steroids, antihypertensive drugs or antiasthmatic drugs were also avoided.

Anticoagulant was mixed gently with the blood which was then contrifuged at 100g for 15 mins. The plateletrich plasma (PRP) was aspirated off, avoiding the layer near the white cells, using a siliconized-glass Pasteur pipette. This plasma was pooled in a polypropylene vessel, then taken up in a syringe from which air was subsequently expelled. The syringe was tightly capped and placed in a 37°C water bath for the duration of the experiment. Under these storage conditions the aggregation response remains constant for 2.5-3 hours. The platelets were not used during the first ten minutes of storage. Platelet counts were determined using a Coulter Counter, Model ZF.

A sample of PRP was spun in an Eppendorf microcentrifuge at 8000g for 2 mins. to sediment the platelets. The platelet-poor plasma (PPP) so formed was

then used to establish 100% light transmission using an aggregometer.

2.1.1.4 Preparation of Washed Platelets.

Method A. PRP was prepared from blood anticoagulated with acid citrate. The PRP was centrifuged at 1100g for 15min to form a platelet pellet. The plasma was poured off, the tube allowed to drain and washing fluid (pH6.5) was added. The platelets were resuspended in this washing fluid by pipetting up and down and a further 2ml of acidcitrate dextrose was added per 8ml fluid prior to further centrifugation. The platelets were resuspended in suspending fluid (pH7.4)(see below) and stored in an airtight syringe at 37°C.

Method B. PRP was obtained from the Red Cross Blood Bank, derived from a unit of blood. The PRP was spun at 100g in the presence of 2ng/ml prostacyclin (PGI₂) to remove more red blood cells. The supernatant PRP was then centrifuged with 300ng PGI₂/ml at 1100g for 15min to form a platelet pellet. These platelets were resuspended in washing fluid (pH7.4) with 150ng PGI₂/ml and apyrase (1ml/50ml plasma)(see below for preparation of apyrase). The platelets were again centrifuged at1100g for 15min and the resultant pellet resuspended in suspending fluid (pH7.4). The platelets were stored in an airtight syringe at 37°C. A recovery time of 30min was allowed to reverse the effects of the PGI₂. This method is a modification of Vargas et al.(1982).

2.1.1.5 Observation of Aggregation using an Aggregometer and Lumi-aggregometer.

A turbidometric method for observation of platelet aggregation was devised by Born in 1962. This widely accepted technique was adopted, using a Payton dual channel Aggregation Module to measure changes in light transmission through stirred samples of PRP in siliconized glass cuvettes.

Changes in light transmission were recorded with a Riken-Denshi SP-H6P recorder with a paper speed of 600mm/hour. A baseline was established for unstimulated swirling platelet-rich plasma (zero transmission) and 100% transmission was set for PPP. This system allowed for observation of shape change when the turbidity increased slightly and subsequent aggregation associated with an increase in light transmission.

A Chrono-Log Lumi-Aggregometer was used to measure the release of ATP associated with platelet aggregation Firefly lantern extracts were prepared by homogenizing 30mg Sigma FFT in 1ml of 100mM HCl, filtering through gauze and finally centrifuging at 1,000g for 5 mins. 25µl of this preparation was added to 500µl PRP containing 20µl of 154mM MgCl₂.

Standard solutions of ATP were prepared freshly on the day of the experiment in Reagent Diluent (RD, see Section 2.2). One test for each of a range of concentrations of ATP was used to determine the number of divisions of

recorder deflection for that concentration. In this manner the ATP assay was semi-quantitative.

2.1.2 Radioimmunoassay of Thromboxane B2

2.1.2.1 Antibody Binding and Crossreactivity

TXB₂ antiserum (#280 D21) was a kind gift from Dr L. Levine, Waltham, Mass. USA. With the number of counts of the tracer set at 4,000 cpm per 100µl, 50% binding was achieved with a final dilution of antiserum of 1 in 20,000. A dilution of 1 in 10,000 shifted the standard curve to the right for no increase in binding. As binding displacement was required for standards in the lower range and, of course for conservation of antibody, the 1 in 20,000 dilution was chosen. There was less than 1% crossreactivity with PGF₁ a and PGF₂ a.

2.1.2.2. ³H-TXB₂ Characteristics

The radioactive tracer, ³H-TXB₂, was obtained from New England Nuclear in different lots which varied in radiochemical purity, as assessed by thin layer chromatography, between 98 and 98.5%. Each lot had a different specific activity but was counted prior to use and diluted to achieve a final dilution of approximately 4,000 cpm per incubation tube.

2.1.2.3 The Assay

In polystyrene, Wasserman tubes, the radioimmunoassay was set up with each incubate consisting of 100µl each of sample or standard antibody and tracer, added in that order. The tubes were incubated at 4°C for at least 18 hrs and no longer than 2 days. An intra-assay control consisted of placing standard TXB₂ at various positions throughout the one assay set of tubes. Inter-assay variability was controlled by placing in each assay a standard set of tubes from a batch kept frozen.

The standard curve was formed with the dilution from a $2\mu g$ aliquot to make a series of 8 standards from 8pg to $1000pg/100\mu l$. The samples assayed contained plasma which was found to interfere with antibody binding so each incubate contained $10\mu l$ of sample only. Consequently each standard TXB₂ tube had $10\mu l$ PRP (inactivated with $100\mu M$ indomethacin) added. Each tube was vortexed and then incubated.

2.1.2.4 Separation, Counting and Calculations

Separation of bound from unbound tracer was achieved by the addition of 300μ l of dextran-coated charcoal. The mixture of charcoal, dextran and albumin was stirred for 10 min. on ice and the aliquot was rapidly added to each tube which was immediately placed in a centrifuge carrier and shaken.

5 mins after the addition of charcoal to the last tube centrifugation for 10 mins at 1500g on a MSE Coolspin was commenced. The supernatant fluid was decanted into a scintillation vial containing 5ml of Xylene/TritonX100 scintillant. The vials were capped, shaken vigourously and counted in a Packard scintillation counter.

Raw counts were converted to percentage binding by the formula %Bound = Counts - NSB

Total Counts

NSB = non-specific binding, that is, the counts remaining in the supernatant fluid without the presence of antibody. A standard curve of percentage bound radioactivity versus amount of TXB_2 (pg) was plotted on semi-logarithmic paper. The unknown samples were then read from this curve and concentrations in pg/ml were converted to pmole/10⁷ platelets for each experiment.

2.1.3 Phospholipid Analysis

2.1.3.1 Extraction of Membrane Lipids

Platelet reactions in a volume of 900μ l were stopped by the addition of 1.25ml CH₃OH:10M HCl(50:1). The cuvette was rinsed out with 3.1ml CHCl₃OH (25:32) and added to reaction mixture. This was partioned using 1.4ml CHCl₃ and 1.4ml water which was mixed and the mixture centrifuged at 1000g for 10min. The lower layer was dried under nitrogen at 45°C. The phospholipids were redisolved in CHCl₃:CH₃OH:10M HCl (200:100:0.1) containing 0.05% butylated hydroxytoluene (BHT) and stored at -20°C prior to chromatographing. This method is modified from Bligh and Dyer (1959) as used by Lloyd <u>et al.(1972).</u>

2.1.3.2 Paper Chromatography

The lipid extract was separated into different polar lipids in a two dimensional system on Whatman SG-81 chromatography paper. Two different solvent systems were tried with the second one being adopted for experiments. Both were taken from Wuthier (1976).

System A.

1st dimension (long) chloroform: diisobutylketone: methanol: acetic acid: water (45:30:15:20:4)

2nd dimension (short) chloroform: diisobutylketone: pyridine: methanol: water (30:25:35:25:8). System B.

1st dimension (short) chloroform: methanol: 5M NH₄OH (64:34:4).

2nd dimension (long) chloroform: diisobutylketone: pyridine: methanol: water: acetic acid: formic acid (23:20:25:20:3:8:1).

46x57cm sheets of paper were cut into 9 equal-sized sheets 15.3x19.0cm and stored in a dessicator. Diisobutylketone was vacuum distilled prior to use. Solvent mixtures were prepared fresh daily and 100ml added to each tank at least 10min prior to introduction of the paper. TLC tanks were used into which four rolled cylinders of paper could stand side by side.

A spot of lipid extract was made in the lower left hand corner, applying 20-30µg P in about 20µl. The spot was made in the form of a streak about 3x7mm in the direction of the first run. The spot and labels were marked with a pencil. The paper was rolled and hand-sewn with a whip-stitch to form a cylinder. The paper was then placed in the chromatography tank after no longer than fifteen minutes. The solvents were allowed to run to about 1cm from the top which took 2 hr for the first dimension (short) and 3 hr for the 2nd dimension (long). The paper was dried for 30min between the two runs by hanging in a fume cupboard. The cylinder was then cut open and resewn at right angles to the first.

After the second run, the paper was thoroughly dried again in the fume cupboard and prepared for staining which was done in a metal tray. The chromatogram was washed with distilled water and rinsed with 0.5M acetic acid. The staining was then achieved by soaking in rhodamine 6G (0.0012%) for 10min with gently rocking to ensure uniform contact.

The stained chromatogram was viewed under UV while still wet and the spots marked with a dull pencil. When the chromatogram had dried, it was again viewed under UV and the spots checked for colour changes. Lysolipids were detected by a change from pale yellow to blue on drying. 2.1.3.3 Phospholipid Estimation by Phosphorus Analysis.

(Rouser et al., 1966)

Each spot was cut out, weighed and placed in a 15x1cm pyrex test tube. To each tube was added 750µl 70% perchloric acid containing 1ml/litre of 10% ammonium molybdate. The tubes were heated in a radiant heat apparatus at 180-200°C until the colour disappeared. To the cooled tubes 3ml of water was added and mixed. Next, 0.5ml 2.5% ammonium molybdate then 0.5ml 10% ascorbic acid was added to each tube, mixing between additions. The tubes were incubated at 56°C for 30min. The silica was removed by centrifugation and the supernatant fluid read on a Beckman DU-8 spectrophotometer. A zero standard was used as a blank for the instrument. A standard curve was prepared using standard inorganic phosphorus solutions in the range 0.1 to 10µg/ml. Each unknown phosphorus was estimated from the standard curve allowing for the phosphorus content of the paper itself. The paper phosphorus was calculated from non-stained areas of the chromatogram which were weighed and digested as were the lipid spots. The estimation of phosphorus per mg paper allowed an estimation of phosphorus due to the paper alone from the weight of the spot.

2.1.3.4 Phospholipid Turnover as Indicated by ³H-arachidonic Acid Incorporation

³H-arachidonic acid was incorporated in the membranes of platelets by incubating at 37°C for 1 hr with 5µl ³Harachidonic acid (Amersham), specific activity 137Ci/m mol per 100ml of original PRP volume. (5µl=5µCi). The incubation was done in washing fluid, ensuring prostacyclin 150µg/ml was added for the incubation and again immediately prior to centrifugation. The platelets were then resuspended as usual and allowed to recover for 15min at 37°C.

Platelet reactions, extraction of lipids and chromatography were carried out as described above. The

spots were cut out and placed in scintillation vials to which 5ml of scintillant was added. The vials were counted in a Packard Tri-carb 460 scintillation counter which recorded counts as disintegrations per min. Different quenching due to the different sizes of stained paper was found to be insignificant.

2.2 MATERIALS

2.2.1 Buffers and Solutions.

Reagent Diluent (RD) is a modified Tyrode's solution with the following composition:8.0g/1 NaCl, 0.2g/1 KCl., 1.0g/1 NaHCO₃, 0.05g/1 NaH₂PO₄ at pH 7.4. This solution was prepared weekly and stored at 4°C. It was used to dilute reagents used in aggregation studies.

Washing fluid was prepared on the day required to give a final composition of: NaCl 8.0g/1,KCl 0.2g/1,NaHCO₃ 1.0g/1,NaH₂PO₄ 0.05g/1, albumin 3.5g/1, glucose1g/1; pH7.4 or 6.5 at 37° C. Apyrase was added (Ardlie <u>et al</u>., 1970; 1971) immediately prior to addition to platelets or 2ml extra ACD was added per 8ml fluid. The solutions were kept at 37° C prior to use. ACD = acid citrate dextrose which consists of 25g trisodium citrate, 14g citric acid and 20g dextrose per litre.

Suspending fluid was prepared on the day required and differed from the washing fluid in containing, in addition, MgCl₂ 0.009g/l and CaCl₂ 0.022g/l;pH7.4,37⁰C.

Stock fibrinogen for adding to washed platelets in order that adrenaline and ADP may cause aggregation. It was prepared from Kabi human fibrinogen, 1g was dissolved in 20ml normal saline, and placed on gentle agitation. 2ml of 20% Al(OH)₃ was added when fibrinogen dissolved. The solution was left for 10min at room temperature for adsorption and centrifuged at 2000g for 10min. The adsorption was repeated with another 2ml of 20% Al(OH)₃. The fibrinogen solution was then dialyzed against normal saline at room temperature with four changes overnight. The dialyzed solution was ultracentrifuged at 37,000g for 20min ambient temperature on a Beckman J2-21 centrifuge. The protein concentration was estimated by a modified Lowry method (Peterson, 1977). Aliquots of 100µl were stored at -20°C and thawed on the day required.

The stock solution was estimated to be 1.2g%. An aliquot was thawed by warming and used at 20µl per 750µl platelet suspension. This gave a final concentration of approximately 0.3mg/ml.

Apyrase was prepared from potatoes and stored in 1ml aliquots at -20°C. Peeled, sliced potatoes were homogenized in a Waring blender $(10 \text{kg/litre d.H}_20)$. The suspension was stirred a further 30min at room temperature and then pressed through 6 layers of cheese cloth. The effluent was centrifuged at 900g R.T. for 10min. The supernatant fluid was refiltered and centrifuged. Further steps were carried out at 0-10°C. The supernatant fluid was brought to 25mM CaCl₂ and stirred 15min at 4°C. The solution was allowed to settle for 1 hr at 4°C, then centrifuged 3500g for 20min at 4°C. The precipitate was resuspended in 1.0M CaCl₂, stirred 60min 0-4°C and centrifuged as before. The supernatant fluid was then dialyzed against 201 0.1M KCl for 24hr 4°C. The supernatant fluid was combined with saturated ammonium sulphate and stirred for 40min, then again centrifuged. The precipitate was redissolved in 25ml distilled water and dialyzed twice against normal saline overnight.

For platelet counting, 5µl of platelets were added to 15 mls of Isoton II (azide free) balanced electrolyte solution.

TRIS-gel buffer for radioiminunoassay was a 15mM TRISbase (Sigma) solution with 500mg/l gelatin (Ajax) added. A stock solution of 150mM TRIS-HCl, pH 7.4 was kept at 4°C. As required this was diluted one-tenth, gelatine dissolved in it by warming, then adjusted to pH 7.4 at 4°C.

2.2.2 Aggregating Agents

Adenosine diphosphate (ADP)(Sigma) was made up in 100µl aliquots of 10^{-2} M and stored at -20° C. Just prior to an experiment this solution was further diluted with RD.

Adrenaline bitartrate (Calbiochem San Diego, CA, USA) was prepared and stored the same way as ADP, except that adrenaline was kept in the dark.

Arachidonic acid was converted to the potassium salt by dissolving 100mg in 11.25ml ethanol containing 18.42mg KOH. This was stored in 1ml aliquots in the dark at -20°C. Immediately prior to each experiment, the potassium

arachidonate was dried down under nitrogen and redissovled in RD.

5 hydroxytryptamine (serotonin) from Sigma was prepared as required.

A 20U/ml ampoule of Vasopressin (Pitressin(R), Parke-Davis) was opened for each experiment and diluted as required in RD.

Noradrenaline (Sigma) was prepared and stored the same way as adrenaline.

Clonidine (Catapres pure substance) a gift from Boehringer Ingelheim, was prepared and stored the same way as adrenaline.

Phenylephrine-HCl(Sigma) was kept dry at 4°C and prepared as required.

Phospholipase C (Sigma) of specific activity 20U/mg protein or 8U/mg solid was stored at -20°C and prepared in RD as required.

Thrombin (Parke Davis, bovine origin) was diluted in RD to 1000U/mland stored in 100µl aliquots at -20° C.

2.2.3 Inhibitors

Acetysalicylic acid, aspirin (ASA) (BDH,general purpose reagent) 50mM was made freshly each day with 90mg per 10ml. It was dissolved by stirring and adding NaOH until the pH was approx. 7.3. This solution was diluted as required and used at one-tenth volume of PRP.

Indomethacin (Sigma) was prepared in absolute ethanol at a concentration of 10mM to be used at one-hundredth the volume of PRP at a final concentration of 100µM. BW755C; 3-Amino-1-[M-(trifluoromethyl)-phenyl]-2-

pyrazoline, a gift from Burroughs-Wellcome, was dissolved at a concentration of 100μ g/ml (380μ M) in RD and used at one-tenth volume in PRP.

n-butyl-imidazole (Koch-Light), a liquid stored at 4°C was freshly diluted in RD to obtain a concentration of 50mM and was used at one-tenth volume in PRP. p-Bromophenacyl bromide, BPB (Koch-Light) was made freshly in dimethylsulphoxide (DMSO) at a concentration of 5mM to be used at one-hundredth volume in PRP.

Tosylphenylalaninechloromethylketone (TPCK) (Sigma) was also dissolved in DMSO.

Dibenamine and phenoxybenzamine were both gifts from Smith, Kline and French, Frenchs Forrest, NSW. Both were prepared freshly in ethanol and added at one-hundredth volume or diluted further in saline.

Hirudin (Sigma Chemical Co.) was dissolved freshly in RD.

Isoptin (verapamil)(Knoll, 2.5/ml ampoules) was stored at 4°C and diluted in RD as required.

Largactil (chlorpromazine)(May and Baker, 25mg/ml ampoules) was stored at 4°C and diluted as required. This preparation also contained sodium sulphite, sodium metabisulphite and sodium citrate.

Phenergan (promethazine)(May and Baker, 25mg/ml ampoules) was stored at 4°C and diluted as required as was trifluoperazine. Mepacrine(quinacrine)(May and Baker) was prepared freshly in DMSO.

Prostacyclin, a gift from the Upjohn Company, Kalamazoo, Michigan, U.S.A. was prepared in 0.05M TRIS buffer, pH9.3 freshly each week.

Trimetoquinol (S-TMQ)was a kind gift from Yoshio Iwasawa, Tanabe Sekako Co.Ltd.Japan. and was stored at -20°C dissolved in DMSO as required

Spermine (N,N-bis[3-aminopropyl]-1,4 butanediamine) Sigma was dissolved in RD.

2.2.4 Anticoagulants.

Citrate, (trisodium salt) was kept as a stock solution of 100mM and used in a ratio of 1ml to 9ml of blood, giving a final concentration of 10mM.

Heparin (Injection B.P., Allen and Hanburys) was used freshly from an unopened ampoule, diluted in RD to give desired concentration from a one-in-ten dilution. Dried sodium heparin (mucous) from Commonwealth Serum Laboratories was dissolved freshly as required. Heparin from Evans Medical Ltd., Liverpool, England was used for some experiments.

Acid citrate dextrose (ACD) was used as anticoagulant for some preparations of washed platelets. It was prepared from 25g trisodium citrate, 14g citric acid and 20g dextrose per litre.

2.2.5 RIA Reagents

Antibody against thromboxane B_2 #280 D21 courtesy Dr T. Levine, Waltham, Mass. U.S.A. was divided into 5µl aliquots which were freeze dried. As required, each aliquot was reconstituted to 1ml with TRIS-gel buffer pH 7.4 and diluted on the day of assay. Anything remaining of the reconstituted aliquot was stored at 4°C, not frozen, and discarded after one week if not used.

³H-TXB₂ (New England Nuclear) was stored at -20°C with the addition of 5ml of ethanol. On the day of assay the appropriate volume was removed and diluted with TRISgel buffer pH 7.4.

Dried TXB₂ (U51541) for use as a standard was a gift from The Upjohn Companay, Kalamazoo, Michigan, USA. 0.569mg was weighed on a Perkin-Elmer Autobalance AD-27, weighed by difference. This sample was then dissolved in 50ml water in a standard flask from which 200 aliquots were freeze dried and kept at -20°C.

Dextran coated charcoal was prepared freshly by combining 10mg dextran T-10, 10mg bovine serum albumin and 60mg charcoal (Norrit A) per 10ml TRIS-gel at 4°C.

2.2.6 Solvents for Paper Chromatography

The solvents used for the paper chromatograph were analytical reagents, except for diisobutylketone (2,6dimethyl-4-heptanone, Ega-Chemie) which was redistilled by vacuum distillation. The other solvents were chloroform AR (Ajax), methanol AR (Ajax), pyridine (Ajax), 5N ammonia (NH₄OH)(Commonwealth Ammonia Co.), acetic acid (Ajax) formic acid (BDH) and nanopure distilled water.

2.2.7 Reagents for Phosphorus Assay

Digestions were performed in 70% perchloric acid (G.

Frederick Smith Co., Columbus, Ohio). Colour development in the assay was produced by ammonium molybdate (Baker Chemical Co.) and ascorbic acid (Calbiochem). Nanopure distilled water was used for all solutions. CHAPTER 3

MULTIPLE AGONISTS IN VITRO

3.0 SUMMARY

This study examined the interaction between several agonists of human platelet aggregation in blood collected into low concentrations of heparin, in an attempt to reproduce the in vivo situation. These conditions were chosen because multiple agonists are present in circulating blood. Another characteristic of the haemostatic mechanism may be the physiologic regulation of blood coagulation by vascular endothelial heparan sulphate. Irreversible aggregation was observed in vitro at pathological levels of adrenaline, when used in combination with ADP and vasopressin. Even lower levels, in the physiological range, caused aggregation when four agonists were combined. In vivo aggregation might therefore be expected to occur under certain circumstances. Concentrations of adrenaline, vasopressin and ADP which together caused irreversible aggregation in heparinized PRP were insufficient to cause this response in citrated PRP from the same donor, in some cases, indicating a necessity for calcium in the potentiation mechanism. Any increase in aggregation in heparinized plasma was reduced by hirudin, indicating that thrombin can act as an additional agonist. It is suggested that the ability of adrenaline to potentiate platelet aggregation by physiological agents such as ADP and thrombin provides a mechanism for facilitating platelet activation at sites of vascular injury.

3.1. INTRODUCTION

The study of the effects of several different agonists on platelet aggregation is of interest because it reflects more closely the <u>in vivo</u> platelet environment in two aspects. Firstly, the circulating blood contains several agonists which can influence the platelet at any one time. Secondly, most of these agonists may not be raised above normal levels in the circulation in a given situation. Use of multiple agonists allows lower concentrations of each agonist to be tested for effects on platelet aggregation. In studies using single agonists the concentrations required are well in excess of physiological concentrations.

The phenomenon of positive interaction of low concentrations of agonists with platelets was first shown by O'Brien (1964). Ardlie <u>et al</u>.(1966) first described the effect of prior exposure to catecholamines, adrenaline, noradrenaline and dopamine. Catecholamines potentiated the response to ADP <u>in vitro</u> which suggested that the role of catecholamines in the initiation and subsequent development of thrombi deserved further investigation.

Aggregating agents were studied further in paired combination by Niewiarowski and Thomas (1966), Mills and Roberts (1967), Baumgartner and Born (1968), Nakanishi <u>et</u> <u>al.(1971), Packham et al.(1973), Michal and Motamed (1976)</u> and Grant and Scrutton (1980a). No multiple agonist studies have been undertaken.

The work of Huang and Detwiler (1981) contributed little to our understanding of the mechanisms of potentiation by particular agonists. Their work was based on varying the concentrations of different pairs of either weak or strong agonists. They concluded that the concentration of agonist determined the characteristics of the response regardless of the order of addition of the agonists. Their conclusion differed from Grant and Scrutton (1980a) and they claimed that this was due to the fact that the latter workers always added the higher concentration of agonist second. This was not, in fact, the case and where the concentration of the second agonist was lower than the first, the aggregation tracing still showed more of the characteristics of the second agonist.

Further evidence that the order of addition is significant comes from Owen and Le Breton (1980) who showed that calcium uptake differed, depending on the order of addition of adrenaline and ADP. When ADP was added first there was measurable ⁴⁵Ca uptake but not when adrenaline was added first. The concentrations of each agonist were kept constant and the final extent of aggregation was the same, regardless of order of addition.

Two agonists investigated in the present study may be increased in the circulation under conditions of stress. Stress may be an important risk factor in coronary artery disease. Baseline levels of adrenaline reported by Dimsdale and Moss (1980) were 0.64nM (range 0.48-1.2nM) which could rise 3-fold during public speaking.

Noradrenaline could rise 3-fold during vigorous exercise and increased 50% during public speaking from baseline levels of 3.4nM (range 2.2-4.3nM). Pathological stress could increase adrenaline levels to 10nM (Grant and Scrutton, 1980a).

Grant and Scrutton (1980a) reported that vasopressin increased 10-fold in adrenocortical insufficiency, from normal levels of 0.1nM but Johnston et al.(1981) quote normal levels of 1-5pM. The level of 5-hydroxytryptamine in blood is 0.1-10µM (De Clerck and Herman, 1983) which would be locally increased once platelet release has been activated. ADP circulating levels are unknown and it is also not known what a pathological level of ADP would be. The work on positive interaction of platelet agonists has, so far, only reported effects at concentrations higher than pathological levels. The lowest concentrations at which adrenaline and vasopressin interacted postively were at least two orders of magnitude greater than the blood concentrations of these hormones achieved in pathological states (Grant and Scrutton, 1980a). To be able to achieve potentiation of aggregation with at least pathological levels would be a major step in accepting in vitro aggregation as a proper reflection of in vivo phenomena.

The present work extends the study of the interaction of aggregating agents by challenging platelets with up to four agonists in immediate succession. It was suspected that an additional agonist, thrombin, would be present as a very low concentration of heparin was used as anticoagulant. In these ways, it was anticipated that the study approximate more closely the situation in circulating blood. Heparan sulphate associated with vascular endothelium may be a physiologic regulator of blood coagulation (Rosenberg, 1977) so that heparin can be regarded as a more 'natural' anticoagulant than citrate (see section 2.1.1.2 for a comparison of the use of heparin versus citrate).

The study does not attempt to uncover a mechanism for the potentiation by any of the agonists. The aim of the study was to find out how low a concentration of an agonist can be used <u>in vitro</u>. In particular, a verification of the importance of circulating levels of catecholamines in platelet activation was sought. The lowest possible concentration of heparin was used to achieve anticoagulation. This might allow thrombin generation. The platelets sometimes showed increasing sensitivity during the course of a single experiment but each result was controlled by observing, simultaneously, the response obtained in the absence of the final agonist.

3.2 METHODS AND MATERIALS

Blood was obtained and plasma prepared as described in Chapter 2. 1.3U/ml heparin was used as the anticoagulant for these experiments. The stress of venepuncture was minimized by allowing the donor to lie down and by the use of a winged infusion set (21Gx3/4") with a 12" (305mm) extension tube. The first two drops

were discarded and withdrawal of the blood with a syringe was continued without a tourniquet.

Initially heparin (CSL) (161.8U/mg) was used but was found to allow microaggregates to form after centrifugation with some donors. This type of heparin was reported (Han and Ardlie, 1974) as being pro-aggregatory at 8.3U/ml when added to washed platelets. Thereafter, liquid heparin from Allen and Hanbury's was used in this study with more satisfactory anticoagulation.

Since the stimulation by each agonist is not instantaneous and is only transitory, agonists were added at 15sec. intervals for an optimal rate of primary response for the range of concentrations used(Grant and Scrutton, 1980). The order of addition of agonists was kept constant,firstly adrenaline, then serotonin(5hydroxytryptamine),noradrenaline or vasopressin and lastly ADP.

Thrombin in PRP was determined using the chromogenic substrate Chromozym TH. 770µl TRIS imidazole buffer (pH 8.4; 0.2M in 0.125M NaCl),10µl PRP (or thrombin) and 20µl Chromozym TH (187.5µM) were incubated in a cuvette at 37°C and the absorption per min (at 405nm) was recorded.

ADP, noradrenaline, adrenaline, 5-hydroxytryptamine (5HT), vasopressin, heparin and hirudin were obtained and stored as described in 2.2.

3.3 RESULTS

3.3.1 The Use of Lower Concentrations of Heparin to Prepare Platelet Rich Plasma.

As discussed in Chapter 2, the characteristics of platelet aggregation induced in heparinized plasma have been found to differ by different authors. Observations were therefore made with different concentrations of heparin and these were compared with citrate.

3.3.1.1 Effect of Heparin Concentration on Biphasic Aggregation due to ADP

As the heparin concentration for anticoagulation was increased, so the threshold concentration of ADP required to cause a biphasic response increased(Fig 3.1). This reduced sensitivity to ADP was further illustrated when the concentration of heparin was increased such that only a single phase occurred in response to 5μ M ADP(Fig 3.2). 3.3.1.2 Comparison of Citrate and 1-2U/ml Heparin

Heparinized platelet rich plasma was not always more sensitive to ADP than was citrated plasma. The pattern of response varied from donor to donor. Further, not all donors exhibited a biphasic ADP response in either citrate or heparin. Of those that did, the threshold concentration for induction of biphasic aggregation in response to ADP was often the same for citrate and 1-2U/ml heparin. Of eleven donors tested, four exhibited greater sensitivity in heparin (Table 3.1). For each of these donors, the extent of primary aggregation in response to ADP was greater in PRP containing heparin (1-2 units/ml) than in PRP anticoagulated with citrate. Furthermore, vasopressin (4µM) caused maximal aggregation in heparinized PRP, but almost no aggregation in citrated PRP. These differences could not be accounted for by differences in platelet count, since these were not significant.

Table 3.1 Comparison Between Threshold Concentrations of ADP for Induction of Biphasic Aggregation in Citrated and Heparinized (1-2U/ml) PRP.

Donor No	ADP threshold (µM)	
	citrated PRP	Heparinized PRP
1	2.7	1.8
2	1.8	1.8
3	5.4	5.4
4	7.2	7.2
5	8.1	7.2
6	1.8	1.8
7	3.6	1.8
8	5.4	5.4
9	4.5	4.5
10	4.5	4.5
11	4.5	2.7



Figure 3.1 The effect of heparin concentration on ADP threshold.

The biphasic aggregation responses to adenosine diphosphate (ADP) are shown. Platelet rich plasma (PRP) was anticoagulated with 1.3U/ml and 4.0U/ml heparin, the latter concentration requiring more ADP to achieve the biphasic response. The arrow indicates the time of addition of the ADP.



Time(min)

Figure 3.2 <u>High concentrations of heparin prevent</u> biphasic aggregation.

Platelet rich plasma from the same donor was anticoagulated with different concentrations of heparin (as shown) or citrate (10mM). ADP was added at increasing concentrations until biphasic aggregation was achieved. For concentrations of heparin of 5U/ml and higher, biphasic aggregation was inhibited for ADP (5μ M). The threshold heparin concentration for this inhibition varied between donors. Traces B, C and D were associated with release of ATP. The arrow indicates the time of addition of ADP. Platelet counts varied from 290,000-330,000/µl. 3.3.1.3 Effect of Added Heparin on the Interaction between Adrenaline, Vasopressin and ADP.

Plasma was anticoagulated with 1.3U/ml heparin, and aggregation was achieved with a combination of subthreshold concentrations of adrenaline, vasopressin and ADP. Heparin was added at two different concentrations three minutes prior to the addition of these agonists (Fig 3.3). Doubling the heparin concentration did not alter the response significantly and tripling the heparin concentration increased the response by only 5%. 3.3.1.4 Thrombin as a Candidate for Explaining any

Increased Sensitivity of Plasma Anticoagulated with 1.3U/ml Heparin.

Thrombin was shown to potentiate the response to adrenaline and ADP in citrated plasma (Fig 3.4).

Hirudin, a specific thrombin inhibitor, reduced the response to the combined addition of adrenaline, vasopressin and ADP to heparinized plasma (Fig 3.5). No significant differences were observed, however, in the thrombin levels (0-0.06U) using the chromogenic substrate, Chromozym TH in citrated or heparinized PRP from the same donor.

3.3.1.5 Discussion on the Use of Heparin.

Higher concentrations of heparin tend to increase the threshold for ADP which could be explained by inhibition of any thrombin which may be present. The low concentration of 1.3U/ml heparin sometimes increased sensitivity when compared to responses in citrate, possibly due to a proaggregatory effect. It is probable that citrate inhibits vasopressin by reducing the concentration of calcium ions



Figure 3.3 The effect of added heparin on the response to the combined addition of adrenaline, vasopressin and ADP.

The platelet count was $295,000/\mu$ l in PRP anticoagulated with 1.3U/ml heparin. In another experiment 3.5U/ml extra heparin did not alter the response at all. The arrows indicate the time of addition of :

- 1. adrenaline (22nM)
- 2. vasopressin (2nM) and
- 3. ADP (435nM).
- A. no added heparin (RD).
- B. 1.7U/ml extra heparin.
- C. 3.5U/ml extra heparin.



Time(min)

Figure 3.4 The potentiation of adrenaline and ADP by thrombin in PRP anticoagulated with 10mM citrate.

RD + 4nM adrenaline + 0.35µM ADP Α. RD + 2.6mU/ml thrombin Β. 2.6mU/ml thrombin + RD + 0.35µM ADP C. 2.6mU/ml thrombin + 4nM adenaline + 0.35µM ADP D. The arrows indicate the time of addition in the order: 1. thrombin or RD adrenaline or RD 2.

3. ADP.



Figure 3.5 The effect of pre-incubation with hirudin (1U/ml), a specific thrombin inhibitor, on the response to adrenaline, vasopressin and ADP.

PRP was anticoagulated with 1.3U/ml heparin. The arrows indicate the time of addition of:

- 1. adrenaline 18nM
- 2. vasopressin 0.8nM
- 3. ADP 0.18µM.

(Haslam and Rosson, 1972). For most donors, 1.3U/ml heparin did not increase the sensitivity to ADP at all. When three agonists were used, added heparin at 3 times the anticoagulating concentration caused a small increase in response in one experiment. Heparin at low (1-1.3U/ml) concentrations was therefore considered to be the most suitable choice for a multiple agonist study where the lowest possible agonist concentrations of agonists was to be tested. It is clear that biphasic aggregation is not an artifact of citrated plasma and with most donors 1.3U/ml heparin did not inhibit the second phase with associated release caused by ADP.

 3.3.2 <u>Agonist Interactions Involving Adrenaline, ADP,</u> <u>Vasopressin, Noradrenaline and 5-Hydroxytryptamine</u>.
3.3.2.1 Interaction Between Two Agonists.

The paired interaction of adrenaline and ADP or vasopressin and ADP resulted in irreversible aggregation for subthreshold concentrations of each agonist (Fig 3.6) 3.3.2.2 Interaction Between Three Agonists

When three agonists were added in combination, lower concentrations of each agonist were required to achieve irreversible aggregation. Fig 3.7 illustrates an example of the responses when 1U/ml heparin (CSL) was used, and Fig 3.8 illustrates an example of the responses when 1.3U/ml heparin (Allen and Hanburys) was used. 3.3.2.3 Interaction Between Four Agonists.

Noradrenaline and 5HT are also present in the circulation in addition to vasopressin, adrenaline and ADP.

With the addition of noradrenaline to the three agonists already used, even lower concentrations could achieve irreversible aggregation (Fig 3.9). The result was variable, however, and as much as 43nM noradrenaline was required for irreversible aggregation when using adrenaline (4.3nM), vasopressin (0.5 - 1nM) and ADP (0.2 - 0.4μ M) in some donors.

5HT added in combination with adrenaline, vasopressin and ADP also showed variable responses. For some donors aggregation was always reversible, no matter how much 5HT added. The lowest concentration of 5HT capable of causing irreversible aggregation in combination with the other three agonists was 120nM (Fig 3.10). The lowest concentrations of each agonist which interacted positively when combined with other agonists are shown, together with their physiological and pathological levels, where known, in Table 3.2.
Table 3.2 Comparison of lowest concentration of agonist causing in vitro aggregation with reported physiological and pathological levels in circulating blood.

Agonist	Physiological Level	Pathological Level	Level to cause in vitro aggregation in this study
Adrenaline	1nM*	10nM*	2.6nM
Noradren.	2.2-4.3nM@		4.3nM
Vasopressi	n 1-5pM#	1nM*	0.5nM
5-HT	0.1-10µM ⁰		120nM

* Grant and Scrutton (1980). @ Dimsdale and Moss(1980). # Johnston et al. (1981). ⁰ De Clerck and Herman(1983) The levels given causing in vitro aggregation are those achieved by the addition of agonists. No assay was done on the existing levels in the blood. The stress of venepuncture was minimized as described.

3.3.2.4 Effect of Citrate on the Response to Multiple

Agonists.

The response to multiple agonists was decreased in PRP obtained from blood anticoagulated with citrate compared with PRP obtained from blood anticoagulated with heparin in several donors. The concentration of adrenaline needed to enhance the aggregation response in citrated plasma was at least two orders of magnitude greater than the minimum concentration which achieved amplification in plasma containing a low concentration of heparin. Less adrenaline was required as the concentration of ADP was increased.



Figure 3.6 The positive interaction of adrenaline and ADP, vasopressin and ADP.

The PRP was anticoagulated with 1.2U/ml heparin. Neither 18nM vasopressin nor 42nM adrenaline caused aggregation when added by themselves. These responses were obtained 120 min after blood collection time. Platelet count was 300,000/µl. Agonists were added at the times indicated by the arrows which are 15sec. apart. The order of addition was as indicated below. RD = reagent diluent

Agonists

2.

Α.	RD	800nM ADP
Β.	18nM vasopressin	800nM ADP
С.	42nM adrenaline	800nM ADP

1.



Figure 3.7 The positive interaction of adrenaline, vasopressin and ADP.

The PRP was anticoagulated with 1U/ml heparin (CSL). 2.5nM adrenaline did not cause aggregation by itself. These responses were obtained 200 mins after blood collection. Platelet count was $277,000/\mu$ l. Agonists were added at the times indicated by the arrows which are 15sec. apart. The order of addition was as indicated below. RD = reagent diluent

Agonists

2.

3.

	1.	2.	3.
A.	RD	0.2nM vasopressin	200nM ADP
B.	3.5nM adrenaline	0.2nM vasopressin	200nM ADP



Time(min)

Figure 3.8 The positive interaction of adrenaline, vasopressin and ADP.

The PRP was anticoagulated with 1.3U/ml heparin (A + H). 21nM adrenaline did not cause aggregation by itself. These responses were obtained 70 mins after blood collection. Platelet count was $307,000/\mu$ l. Agonists were added at the times indicated by the arrows which are 15sec. apart. The order of addition was as indicated below. RD = reagent diluent

Agonists

1.		2.	3.	
A.	RD	1.0nM vasopressin	200nM ADP	
B.	21nM adrenaline	1.0nM vasopressin	200nM ADP	



Time(min)

Figure 3.9 The positive interaction of adrenaline, noradrenaline, vasopressin and ADP.

The PRP was anticoagulated with 1U/ml heparin (CSL). 2.6nM adrenaline did not cause aggregation by itself. These responses were obtained 130 mins after blood collection. Platelet count was $319,000/\mu$ l. Agonists were added at the times indicated by the arrows which are 15sec. apart. The order of addition was as indicated below. RD = reagent diluent

Agonists

	1.	2.	3.	4.
Α.	RD	4.3nM noradrenaline	0.5nM vasopressin	400nM ADP
B.	2.6nM	4.3nM noradrenaline	0.5nM vasopressin	400nM ADP



Figure 3.10 The positive interaction of adrenaline, vasopressin, 5-hydroxytryptamine and ADP.

The PRP was anticoagulated with 1.3U/ml heparin (A + H). 4.3nM adrenaline did not cause aggregation by itself. These responses were abtained 140 mins after blood collection. Platelet count was 291,000/ μ 1. Agonists were added at the times indicated by the arrows which are 15sec. apart. The order of addition was as indicated in the right hand columns.

RD = reagent diluent

Agonists

	1.	2.	3.	4.
A.	RD	1.0nM vasopressin	120nM 5HT	200nM ADP
в.	4.3nM adrenaline	1.0nM vasopressin	120nM 5HT	200nM ADP

3.4.3 DISCUSSION

This study has demonstrated that enhancement of aggregation of human platelets is possible <u>in vitro</u> using concentrations of catecholamines which could be achieved <u>in</u> <u>vivo</u>. This is in contrast to previously published minimum levels (Grant and Scrutton, 1980a) which have been considerably higher than the levels which occur in pathological states. If the results using low concentrations of heparin can be extrapolated to the <u>in vivo</u> situation, then platelet aggregation may occur in certain states. An example may be the emotional stress of public speaking which is associated with a more than doubling of the plasma level of adrenaline (Dimsdale and Moss, 1980). The potentiating properties of adrenaline may provide a mechanism for facilitating platelet activation at sites of vascular injury.

The method used here for examining agonist interaction differs from others in that a much lower concentration of heparin was used to anticoagulate blood. With this method of collection, platelets were at least as sensitive as those collected in citrate, and often more sensitive to aggregating agents. Those platelets which were found to be more sensitive responded to multiple agonists at concentrations obtained in physiological and pathological states.

The reduction in the sensitivity of the heparinized PRP caused by hirudin, a specific inhibitor for thrombin, suggests that thrombin participates in the aggregatory response. This has been suggested by earlier workers, Ardlie and Han (1974) and Huzoor-Akbar and Ardlie (1976; 1977) who particularly emphasized a role for thrombin in the release reaction. The lack of difference in thrombin levels in PRP anticoagulated with either citrate or heparin seems to negate this theory. Nevertheless, a small amount of thrombin may act as an additional agonist, inhibitable by hirudin. MacIntyre <u>et al</u>.(1981) reported that heparin potentiates aggregation, but in the present study heparin had to be increased 3-fold above the anticoagulant concentrations used to cause any enhancement of aggregation caused by three combined agonists.

The reduced effectiveness of adrenaline in the presence of citrate is consistent with calcium fluxes being involved in the mechanism of adrenaline potentiation (Owen <u>et al</u>., 1980). Others have obtained evidence, however, to show that adrenaline does not cause net influx of calcium into human platelets unless secretion occurs(Brass and Shattil, 1982; Clare and Scrutton, 1984). An alternative explanation for the effect of citrate may have to be sought, not necessarily, however, excluding effects on membrane bound calcium. In addition, and consistent with observations made by Haslam and Rosson (1972), citrate appears to inhibit vasopressin which therefore, probably requires calcium for its action.

There are possible similarities between the test system described for this study and physiological

anticoagulation. Anti-thrombin is considered to be an important natural anticoagulant and its interaction with clotting factors is significantly enhanced by heparin (Rosenberg, 1977). Heparan sulphate, which is associated with endothelial cells, may be a normal cofactor for antithrombin in regulating coagulation. In both the test system described in the present study and in the haemostatic response (Shattil and Bennett, 1980) platelet activation is enhanced by thrombin. Furthermore, the interaction of platelets with a combination of agonists probably more closely reflects platelet activation <u>in vivo</u>.

CHAPTER 4

MECHANISM OF THE FACILITATING EFFECTS OF ADRENALINE ON PLATELET AGGREGATION

that adreading stinuisted a cyclonyymessindependent pathway resulting is potentiation of platelet aggregation. Adreading also stinuisons the cyclonayyonass jathway to an extent sufficient to generate 178, when platelets were under the inititory influence of aspirit, but only upon the addition of large concentrations of assentidence. Adreading was not alle to cause this stimulatum is the presence of indometratic and 507550 o dual contentions of generate and lipoxygeness insibilitor. At the contentiones of generates and lipoxygeness insibility, the was unable to potentions of generations and lipoxygeness insibility as unable to potentions of a state insite and history used, adrensities was unable to potentions of generations attentions to agregation. There appears he he as bitelet Agregation by solar adrensities can potentiate platelet Agregation, one of these being cyclonayyons

4.0 SUMMARY

Adrenaline, by itself, can aggregate human platelets <u>in vitro</u>, and also has the property of being able to potentiate the aggregation response to subthreshold concentr ations of other platelet agonists. Since it is largely inhibited by aspirin, the ability of adrenaline, alone, to cause platelet aggregation in citrated plasma is believed to depend on the cyclooxygenase pathway of arachidonic acid metabolism. This chapter examines whether the potentiating effects of adrenaline also depend on this mechanism.

This report describes the effects of several inhibitors of arachidonate metabolism on adrenalinepotentiated aggregation and TXB2 production. It was shown that adrenaline stimulated a cyclooxygenase-independent pathway resulting in potentiation of platelet aggregation. Adrenaline also stimulated the cyclooxygenase pathway to an extent sufficient to generate TXB2 when platelets were under the inhibitory influence of aspirin, but only upon the addition of large concentrations of arachidonate. Adrenaline was not able to cause this stimulation in the presence of indomethacin and BW755C, a dual cyclooxygenase and lipoxygenase inhibitor. At the concentrations of these inhibitors used, adrenaline was unable to potentiate arachidonate-induced aggregation, but was able to potentiate ADP-induced aggregation. There appears to be at least two mechanisms by which adrenaline can potentiate platelet aggregation, one of these being cyclooxygenase-

independent, the relative importance of which may depend on the function of this pathway in the aggregation response of the principal agonist. In the absence of thromboxanes, more adrenaline is required to potentiate aggregation.

The facilitating effect of adrenaline on a cyclooxygenase-independent pathway was mimicked by clonidine, a partial a_2 -receptor agonist but not by phenylephrine an a_1 -agonist. This indicates that these effects of adrenaline on the platelet occur via stimulation of the a_2 -receptor. Dibenamine and phenoxybenzamine, post-receptor inhibitors, inhibited even high concentrations of adrenaline and inhibited the second phase of the ADP-response. Not only did these inhibitors prevent adrenaline potentiation of ADP, but the response of the combined agonists in their presence, was less than the ADP response alone. The biochemical changes subsequent to receptor occupation are investigated further in later chapters.

4.1 INTRODUCTION

Potentiation of human platelet aggregation by adrenaline may possibly represent a prime role for this catecholamine in the haemostatic response to injury of a blood vessel. Increased levels of this catecholamine may enable platelets to aggregate more quickly and irreversibly to plug a wound. It is not known whether adrenaline, by itself, can cause aggregation <u>in vivo</u> and it is disputed whether the observed <u>in vitro</u> aggregation in citrated PRP is an artifact or not. Adrenaline has been reported, however, by several workers as being able to potentiate the platelet aggregation response to other agonists in both citrated and heparinized plasma (O'Brien, 1964; Ardlie <u>et al</u>., 1966; Mills and Roberts, 1967; Grant and Scrutton, 1980a).

The ability of adrenaline alone to cause platelet aggregation in vitro, is reported by Best (1980) to be largely dependent on the metabolism of arachidonic acid via the cyclooxygenase pathway to thromboxane $A_2(TXA_2)$, the first phase, which is independent of thromboxane, being very small. Both aspirin and n-butyl imidazole, which inhibit cyclooxygenase and thromboxane synthetase respectively, prevent the second phase of adrenalineinduced aggregation. It is of interest to determine whether the ability of adrenaline to potentiate aggregation by other agonists also depends on the metabolism of arachidonic acid.

Rao <u>et al</u>. (1980a) using a thin layer chromatography method, have previously shown that adrenaline can overcome aspirin inhibition of ADP-, and arachidonate-induced aggregation without utilizing the cyclooxygenase pathway, as shown by lack of TXA₂ production. This is not unexpected since aspirin is believed to irreversibly acetylate the enzyme cyclooxygenase, but the mechanism by which adrenaline achieves potentiation independently of prostaglandin synthesis remains largely unexplored.

This study sought to verify first of all that the mechanism of adrenaline potentiation could, in part, be

thomboxane independent. The effects of different cyclooxygenase inhibitors on adrenaline potentiation of ADPand arachidonate-induced aggregation was examined. The effectiveness of these various inhibitory agents on thromboxane synthesis was determined using a radioimmunoassay technique. Clonidine, a partial a_2 -receptor agonist, and phenylephrine (a partial a_1 -agonist) were also studied to determine whether they were able to potentiate aggregation in a similar manner to adrenaline in the absence or presence of these inhibitors.

Following the establishment of the involvement of a thromboxane-independent pathway in adrenaline potentiation, certain aspects of the α -receptor system itself were considered to be worthy of further study. Dibenamine and phenoxybenzamine are thought to be drugs which act on the a-receptor system but by a mechanism independent, to some extent, of agonist binding (Lullman et al., 1969; Mitchelson, 1975). Their actions have been found to be different, despite similarities in structure between the two drugs (Jafferji and Michell, 1976; Cockcroft et al.,1980). Dibenamine seems to inhibit calcium mobilization without altering phosphatidylinositol turnover which is, however, affected by phenoxybenzamine. It was thought, therefore, that the effects of these two drugs on the adrenaline-potentiated response in platelets might reveal more of the events following a-receptor activation by adrenaline. The results indicate that further attention

needs to be focussed on platelet membrane events subsequent to adrenaline a-receptor interaction.

4.2. METHODS AND MATERIALS

Platelet aggregation, release of ATP and TXB₂ formation were assessed utilizing techniques described in Chapter 2. Blood was anticoagulated with citrate.

The agonists used were adrenaline (Calbiochem), adenosine diphosphate (ADP)(Sigma), arachidonic acid (Sigma), clonidine (Boehringer-Ingelheim) and phenylephrine (Sigma).

The inhibitors used were acetylsalicylic acid (ASA)(aspirin) (BDH), indomethacin (Sigma),3-amino-1-[m-(tri-fluoromethyl)-phenyl]-2-pyrazoline (BW755C)(Wellcome), n-butyl-imidazole (Koch-Light), phenoxybenzamine and dibenamine (Smith, Kline and French).

Aspirin,100µM, unless otherwise stated was preincubated with PRP in a tightly stoppered syringe at 37°C for at least 10min prior to the test. The other inhibitors were added to the PRP in the cuvette 1min prior to the addition of agonists.

4.3. RESULTS

4.3.1 <u>Potentiation of ADP-induced Aggregation by</u> Adrenaline

The synergistic effect on agregation of subthreshold concentrations of ADP and adrenaline was associated with a small increase in TXB_2 (Fig 4.1). The appearance of the potentiated aggregation response varied from donor to donor. The tracing took on the appearance of an adrenaline-induced aggregation response in some cases. That is, the "first phase" was small and there was a lag before a second phase. Nevertheless, there was also a shape change. This tended to occur when low concentrations of both ADP and adrenaline were used, with the order of addition being adrenaline before ADP.

Pre-incubation with 100µM aspirin did not prevent adrenaline potentiation of ADP-induced aggregation but did prevent TXB₂ formation (Fig 4.2). In the presence of aspirin, higher concentrations of adrenaline were required to acheive a similar degree of potentiation observed without aspirin. For example, the addition of 0.8µM adrenaline and 1.6µM ADP in the presence of aspirin (Fig4.2,tracingD) resulted in the same extent of aggregation as 0.08µM adrenaline and 1.6µM ADP in the absence of aspirin in the same donor(not shown). Nonetheless, in the presence of aspirin, 0.08µM adrenaline clearly potentiated 1.6µM ADP(Fig.4.2,tracingC).

When PRP was pre-incubated with $10-100\,\mu$ M indomethacin for 1 min, potentiation of the first phase of the ADP response was also possible(Fig 4.3.) In the presence of $100\,\mu$ M indomethacin, the addition of $8\,\mu$ M adrenaline was necessary to obtain marked potentiation of aggregation but without TXB₂ formation. Similarly, when the dual cyclooxygenase and lipoxygenase inhibitor, BW755C ($38\,\mu$ M) was used, enhancement of ADP-induced aggregation by



Figure 4.1 Potentiation of ADP-induced aggregation by adrenaline.

```
A. RD + 0.8µM ADP
B. 0.08µM adrenaline + RD
C. 0.08µM adrenaline + 0.8µM ADP.
Aggregation tracings are shown for PRP prepared from
citrated blood. The arrows indicate the points of addition
of the agonists in the order given above. Similar results
were obtained with four other donors whose TXB<sub>2</sub> levels were
0.4-2.0 pmol/10<sup>7</sup> platelets for irreversible aggregation.
ND = not detected.
```



Figure 4.2 The influence of aspirin on the potentiation of ADP-induced aggregation by adrenaline

Aggregation tracings for the responses in the presence of 100 μ M ASA are shown. The synergistic effect of ADP and adrenaline in the presence of ASA was not associated with the production of TXB₂.

- A. 0.08µM adrenaline + RD.
- B. $RD + 1.6\mu M ADP$.
- C. 0.08µM adrenaline + 1.6µM ADP.
- D. 0.8µM adrenaline + 1.6µM ADP.
- ND = not detected.

• •



Figure 4.3 The influence of indomethacin on the potentiation of ADP-induced aggregation by adrenaline.

PRP was pre-incubated for 1 min with 100μ M indomethacin. A. RD + 0.8μ M ADP. B. 8μ M adrenaline + RD. C. 8μ M adrenaline + 0.8μ M ADP. Similar results were obtained with two other donors using indomethacin concentrations as low as 10μ M. ND = not detected. adrenaline occurred again with no increase in TXB_2 formation. When the concentration of BW755C was decreased to 3.8µM, potentiation occurred with a small increase in TXB₂ formation. The thromboxane synthetase inhibitor, nbutyl imidazole (5mM) also behaved like indomethacin and did not inhibit a prostaglandin-independent ADP-response. Adrenaline overcame the inhibition of ADP-induced aggregation by 1mM n-butyl imidazole with the production of a small amount of TXB₂.

In addition to the assay of thromboxane formation, aggregation and release of nucleotide were assessed simultaneously in a lumi-aggregometer. PRP was preincubated with each inhibitor of arachidonic acid metabolism and adrenaline was added at concentrations which, by itself, were insufficient to overcome the inhibitors alone. When ADP-induced aggregation was enhanced in this manner without an increase in TXB₂ formation, there was no release of nucleotide. When TXB₂ formation occurred, this was accompanied by release of nucleotide.

4.3.2 <u>Potentiation of Potassium Arachidonate (KA)-induced</u> Aggregation

Potentiation by adrenaline of KA-induced aggregation in the absence of inhibitors was associated with an increase in TXB₂ formation (Fig 4.4).

When PRP was preincubated with aspirin (100µM) potentiation of KA-induced aggregation could only be achieved with high(45µM) concentrations of adrenaline(Fig



Figure 4.4 The potentiation of KA-induced aggregation by adrenaline

A. RD + 160µM KA.
B. 0.08µM adrenaline + RD.
C. 0.08µM adrenaline + 160µM KA.
In another experiment where 160µM KA was the subthreshold concentration, TXB₂ levels were 1.6 p mol/10⁷ platelets for KA only and 6.6 p mol/10⁷ platelets for adrenaline-potentiated KA-induced aggregation.
ND = not detected.



Figure 4.5 The influence of aspirin on the potentiation of KA-induced aggregation by adrenaline.

PRP was preincubated with 100µM ASA as described. A. RD + 400µM KA. B. 45µM adrenaline + RD. C. 45µM adrenaline + 400µM KA. ND = not detected.

4.5). This potentiation was small and occurred without any increase in TXB₂. If the concentration of KA was increased approximately 10-fold, such that a small amount of aggregation occurred in the presence of 100μ M aspirin, adrenaline could further increase this aggregation with an increase in TXB₂(Table 4.1). This was possible even if the aspirin concentration was doubled, but again, only at high concentrations of KA. In addition, 5mM n-butylimidazole could not prevent adrenaline potentiation of arachidonateinduced aggregation and TXB₂ formation if the arachidonate concentration was sufficiently high (Fig 4.6).

Table 4.1 Aggregation and TXB₂levels associated with adrenaline potentiation of arachidonate-induced aggregation in the presence or absence of aspirin.

	No Inhibitor		Aspirin	
AGONISTS	%aggreg	TXB ₂	%aggreg	TXB ₂
#1.RD + 0.4mM KA	90	7.6	0	0
45µMadren + RD	85	0.37	10	0
45µMadren+0.4mM KA	100	9.6	16	0
RD + 5mM KA	100	35	10	0.26
45µMadren+ 5mM KA	100	35	50	4.0
#2.RD + 0.4mM KA	0	0.25	0	0
45μ Madren + RD	15	0.92	10	0
45μ Madren+0.4mM KA	90	29	10	0
RD + 3.6mM KA	100	45	8	2.2
45μ Madren+3.6mM KA	100	45	50	7.2

The data for two separate experiments are shown. 100μ M aspirin was used for exp#1 and 200μ M aspirin was used for exp#2. The abbreviations used are: KA = potassium arachidonate, adren = adrenaline, RD = reagent diluent, TXB₂ = thromboxane B₂(pmol/10⁷platelets).

When PRP was preincubated with 10-100µM indomethacin, however, potentiation of KA-induced aggregation by



Figure 4.6 The influence of n-butylimidazole on the potentiation of KA-induced aggregation by adrenaline

PRP was preincubated with 5mM n-butylimidazole for 1 min. A. 0.8μ M adrenaline + 400 μ M KA. B. 0.8μ M adrenaline + 600 μ M KA. Each agonist at these concentrations was inhibited when added separately. ND = not detected. adrenaline (up to $8\,\mu$ M) was not achieved. Some potentiation by adrenaline was achieved with an increase in TXB₂ formation when the concentration of indomethacin was decreased to $5\,\mu$ M which was still sufficient to inhibit each agonist alone. Similarly, when PRP was pre-incubated with $38\,\mu$ M BW755C, no potentiation was achieved, whereas a reduction in the concentration of BW755C to $3.8\,\mu$ M did allow adrenaline potentiation but not to the same extent as in the absence of the inhibitor. With each of these cyclooxygenase inhibitors, when adrenaline did overcome the inhibition of KA-induced aggregation, release of nucleotide was also observed.

4.3.3 Potentiation of ADP-induced and Arachidonate-

induced Aggregation by the a-adrenoceptor Agonists,

Clonidine and Phenylephrine

Clonidine a partial α_2 -agonist behaved in a similar way to adrenaline in being able to potentiate subthreshold concentrations of ADP and KA. Clonidine was able to potentiate responses to ADP in the presence of the cyclooxygenase inhibitors without TXB₂ formation. Clonidine, therefore, facilitated cyclooxygenaseindependent aggregation.

Clonidine usually did not cause aggregation by itself even at concentrations as high as 800µM. However, two patients with thrombotic disorders were found to respond to 0.8µM and 800µM clonidine respectively. The concentrations of clonidine which potentiated aggregation due to ADP in the presence of aspirin were comparable to those of adrenaline, that is in the range 0.8 - 8µM. Nevertheless, in any one donor, a concentration of clonidine which potentiated an ADP reponse was not as effective as the same concentration of adrenaline (Fig 4.7).

Some donors exhibited potentiation of the ADP response by the α_1 -agonist, phenylephrine. Those donors who did, also exhibited an additive response to the combination of clonidine and phenylephrine (Fig 4.7). Much higher concentrations of phenylephrine than of clonidine were required to potentiate ADP. For the donor illustrated in Fig 4.7, 8.0µM phenylephrine potentiated 0.4µM ADP with an associated TXB₂ production of 0.46 pmole/10⁷platelets (2.4ng/ml).

4.3.4 Post-Receptor Inhibitors

As discussed previously, there is some evidence that dibenamine and phenoxybenzamine may be acting on the stimulus-response coupling sequence distal to the receptor. These drugs were therefore used to probe post-receptor events. Both dibenamine (0.2mM) and phenoxybenzamine (0.2mM) inhibited the aggregation response of platelets to adrenaline even when the concentration of adrenaline was increased to $80\,\mu$ M. Adrenaline induced TXB₂ formation was also inhibited by these drugs. These drugs also prevented the second phase of the ADP response with a reduction of the first phase, but had no effect on the ADP-induced shape change. Inhibition of the second phase of the ADP response was associated with an absence of TXB₂ formation.



Time(min)

Figure 4.7 Potentiation of ADP-induced aggregation by adrenaline, clonidine and phenylephrine.

The concentration of ADP for all tracings was 0.4µM. The additional agonists (added at the first arrow) were: A. RD.

B. 0.8µM phenylephrine.

C. 0.008µM clonidine.

D. 0.008µM adrenaline.

E. 0.8µM phenylephrine + 0.008µM clonidine.

This donor was one of a small percentage of donors whose ADP response was potentiated by phenylephrine in addition to clonidine. This donor is extremely sensitive to adrenaline as indicated by the very low concentration $(0.008\mu M)$ which potentiated the ADP response. ND = not detected. Phenoxybenzamine, and sometimes dibenamine, also prevented adrenaline potentiation of ADP. In some donors, the addition of either of these drugs with adrenaline actually diminished the aggregation response to ADP alone. There were no observable differences with these inhibitors when experiments were carried out with platelets from donors exhibiting α_1 -receptors, as detected by the ability of phenylephrine to potentiate the ADP response, and platelets from donors that did not exhibit α_1 -receptors.

4.4 DISCUSSION

These data support the theory that adrenaline can be pro-aggregatory through a thromboxane-independent mechanism. In the presence of cyclooxygenase inhibitors, adrenaline can potentiate aggregation due to other agonists. Adrenaline can also increase thromboxane production despite the presence of aspirin, when high concentrations of arachidonate are added. In the absence of cyclooxygenase inhibitors, adrenaline probably acts through both mechanisms and these are expected to involve activation of α_2 -receptors which is supported by the potentiating effects of clonidine described in this chapter.

Grant and Scrutton(1979) identified the α_2 adrenoceptor as being primarily responsible for mediating the response to adrenaline. They showed that the potentiation of ADP-induced aggregation by clonidine was inhibitable by yohimbine, a selective α_2 -antagonist. In

contrast, indoramin, a selective α_1 -antagonist, was relatively ineffectual as an inhibitor of clonidinestimulated responses. Inhibition of responses to clonidine by yohimbine occurred over a range of concentrations significantly higher than that required for inhibition of adrenaline-induced responses. Clonidine is therefore regarded as a partial α_2 -agonist. Since the possible α_1 -receptor is not exhibited on platelets from all donors as shown by methoxamine potentiation and specific [³H]-prazosin binding (Scrutton, 1982), it cannot be the predominant receptor involved in activation of platelets by adrenaline.

Some conversion of arachidonate to TXB_2 is possible despite irreversible acetylation of cyclooxygenase by aspirin if sufficiently large amounts of exogenous arachidonate are added. Whatever the mechanism of this reaction, high concentrations of adrenaline can enhance it. Concentrations of indomethacin and BW755C less than that of aspirin, inhibited the cyclooxygenase pathway no matter how much adrenaline was added. The chemical nature of the interference of the inhibitor with the enzyme active site may determine whether adrenaline can increase cyclooxygenase activity (Humes <u>et al</u>., 1981). Another explanation for the greater effectiveness of indomethacin may be that this inhibitor also reduces the amount of available endogenous arachidonate (Erman <u>et al</u>., 1980).

Indomethacin was effective in preventing adrenaline potentiation of KA-induced aggregation but not when the inhibitor concentration was reduced to 5µM. Similarly, BW755C, which inhibits both cyclooxygenase and

lipoxygenase, prevented adrenaline potentiation of KAinduced aggregation at a concentration of 38µM, but not at one-tenth this concentration. Indomethacin(100µM) inhibits lipoxygenase by 20% (Higgs <u>et al</u>.,1979) and can therefore also be a dual inhibitor like BW755C. It is possible that adrenaline achieves potentiation by overcoming inhibition of the lipoxygenase pathway of arachidonate acid metabolism, although no role has yet been assigned to the products of the lipoxygenase pathway in platelet activation.

ADP-induced aggregation can still be enhanced by adrenaline or clonidine under the influence of these inhibitors at concentrations which completely prevent thromboxane production. Adrenaline promotes this thromboxane-independent pathway of ADP-induced aggregation without causing release of nucleotide. Higher concentrations of adrenaline were required to achieve the same extent of aggregation as in the absence of thromboxane inhibitors and much higher concentrations than found <u>in</u> <u>vivo</u>. Without the amplifying effect of thromboxanes and nucleotide, a larger stimulus seems to be required to achieve platelet activation by the thromboxane-independent pathway.

This thromboxane-independent aggregation is also irreversible and can appear indistinguishable from the pattern of irreversible aggregation which is accompanied by release, discerned from recorder tracings only. Additional information on thromboxane formation and the release

reaction are needed to characterize the aggregation response in more detail. What is usually referred to as first phase aggregation could also be referred to as cyclooxygenase-independent aggregation.

There could be several mechanisms by which adrenaline promotes thromboxane-independent aggregation. It has recently been proposed that adrenaline by binding to areceptors, enhances calcium (Ca^{2+}) mobilization and thereby promotes aggregation (Rao et al., 1981a). Stimulation of the α -receptor also increases Ca²⁺ uptake (Owen et al., 1980) but further work (Owen and Le Breton, 1980) showed that potentiation of the ADP response by subthreshold levels of adrenaline was not associated with measurable Ca²⁺ uptake. Calcium redistribution was considered to be involved since the Ca²⁺ channel blocker, verapamil, inhibited potentiation, but it is now acknowledged that verapamil achieves an inhibitory effect at the level of the⁺ a-adrenergic receptor (Barnathan et al., 1982). Brass and Shattil(1982) were unable to find an enhancement of net Ca^2 uptake by ADP or adrenaline although both these agonists increased the extent of binding of Ca^{2+} to the surface and also the rate of exchange into the intracellular pool. The biochemical changes leading to this possible redistribution. of Ca²⁺ are not understood and probably only some of the many effects of Ca²⁺ in platelets have been so far revealed Inositol-1,4,5-trisphosphate is a possible mediator of calcium redistribution(Berridge, 1984).

 a_2 -receptor stimulation is thought to be universally associated with inhibition of adenylate cyclase (Fain and Garcia-Sainz, 1980), reducing c'AMP production which could be related to calcium redistribution (Kaser-Glanzman <u>et</u> <u>al.</u>, 1977). Rao <u>et al</u>. (1981a) reported, however, that adrenaline can potentiate platelet responses without measurable changes in c'AMP levels. No net decrease in c'AMP results on exposure of resting platelets to adrenaline (Haslam, 1978a, b). Furthermore, the induction of aggregation by adrenaline has been shown to be dissociated from its effect as an inhibitor of adenylate cyclase, by removal of extracellular Na⁺ (Connolly and Limbird, 1983) or by the use of certain antagonists (Clare et al., 1984).

There could also be other membrane changes in response to a-receptor stimulation. For other tissues, besides platelets, it has been suggested that α_1 -receptors mediate phosphatidylinositol (PI) turnover and hence the release of bound intracellular Ca²⁺, as well as the entry of extracellular Ca²⁺ (Fain and Garcia-Sainz, 1980; Jafferji and Michell, 1976). These Ca²⁺ responses are precisely the reported effects of adrenaline on platelets. The effects of adrenaline on platelets, however, are believed to be mediated by α_2 -receptors, not α_1 -receptors (Grant and Scrutton, 1979). It is also not known whether adrenaline, while potentiating ADP, affects PI turnover. Very recently, Clare and Scrutton(1984) failed to show PI breakdown due to adrenaline alone in the absence of

secretion. In either case polyphosphoinositides have not been examined.

The fact that phenoxybenzamine totally inhibits the adrenaline response in the platelet is consistent with PI turnover being important in mediating the adrenergic effects. The results are obviously difficult to interpret, however, since phenoxybenzamine is an α -receptor(α_1 greater than α_2) antagonist in addition to possessing post-receptor inhibitory activity (Jafferji and Michell, 1976). There is no need to necessarily assume that PI breakdown would depend on α_1 -receptor activation since the particular effects associated with a particular receptor demonstrated in other tissues may not apply to the platelet. The effects of phenoxybenzamine on adrenaline stimulation and potentiation require further study, possibly in association with analysis of membrane phospholipids.

The inhibition by dibenamine is consistent with Ca^{2+} mobilization being a prerequisite for expression of adrenaline stimulation of α -receptors. Even though there is some evidence that calcium uptake may not to be involved in the mechanism of adrenaline activation, calcium movement within the membrane may still be involved. It is possible that effects on calcium may be independent of the PI effect, since dibenamine seems to act independently of PI turnover. The present findings concur with those of Rao et al.(1980a) in establishing the existence of a thromboxane-independent pathway of adrenaline potentiation and both studies show the ability of adrenaline to influence the

aggregation response of aspirin-treated platelets to arachidonate. Adrenaline, together with higher concentrations of arachidonate, partially restores the activity of cyclooxygenase, under conditions appropriate for total enzyme inhibition.

Following cyclooxygenase inhibition, much higher concentrations of adrenaline are required to achieve the same potentiation of ADP-induced aggregation as that observed when the thromboxane pathway is operating. It is unlikely that such concentrations would ever be achieved in the circulation, and therefore thromboxane formation is probably essential for the complete pro-aggregatory effect of adrenaline <u>in vivo</u>. The influence of other agonists present in the circulation on the pro-aggregatory effect of adrenaline has been shown(Chapter 3). Therefore the effective concentration of adrenaline with or without cyclooxygenase inhibition, <u>in vivo</u> is expected to be much lower than the concentrations used <u>in vitro</u>. A thromboxane-independent pathway is involved in adrenaline potentiation the mechanism of which remains unresolved.

CHAPTER 5

THE INFLUENCE OF PHOSPHOLIPASE A₂ INHIBITORS ON ADRENALINE POTENTIATION

5.0 SUMMARY

To further characterize adrenaline potentiation of ADP-induced aggregation, the influence of phospholipase A_2 inhibitors was examined. The inhibitors chosen were bromophenacyl bromide (BPB), a widely reported inhibitor of phospholipase A_2 ; tosylphenylalaninechloromethyl ketone (TPCK), a chymotrypsin inhibitor with reported phospholipase A_2 inhibitory properties; and phenothiazines which inhibit several enzymes including calmodulindependent enzymes and protein kinase C.

BPB and TPCK shared in common the ability to enhance thromboxane production when adrenaline potentiated the ADPresponse. Adrenaline therefore enhances arachidonic acid release by a mechanism independent of phospholipase A₂ activity in the presence of these two inhibitors. The phenothiazines, chlorpromazine and trifluoperazine inhibited thromboxane production but adrenaline potentiation of thromboxane-independent aggregation was still observed.

It is suggested that adrenaline promotes diacylglycerol production from which arachidonic acid release is promoted via lipase activity, in the presence of BPB or TPCK, but not in the presence of phenothiazines, in which case the kinase may be more active.
5.1 INTRODUCTION

In the previous chapter, the mechanism of adrenaline potentiation of ADP and arachidonate-induced aggregation was examined. While adrenaline was found to stimulate a non-prostaglandin pathway there was also evidence of some stimulation of arachidonic acid metabolism. Perhaps adrenaline has a direct effect on cyclooxygenase or thromboxane synthetase activity. It was therefore decided to inhibit the prostaglandin pathway by inhibition of arachidonate release because it was thought that inhibition of arachidonate release might allow further characterization of the effects of adrenaline leading to both prostaglandin-dependent and independent pathways of activation.

As discussed in Chapter 1, there are several ways in which arachidonate might be released from membrane phospholipids(Fig 5.1). Phospholipase A₂ enzymes could release the fatty acid from phosphatidic acid or otherwise from the phospholipids, phosphatidylcholine, phosphatidylethanolamine, phosphatidylserine or phosphatidylinositol(PI) (Billah <u>et al</u>., 1980; 1981). Alternatively, phospholipase C and diacylglycerol lipase activity can lead to the release of arachidonic acid from PI (Bell <u>et al</u>., 1979). The respective contributions of these different phospholipases to arachidonic acid release has not been resolved.

It has been difficult to find specific inhibitors of either phospholipase A_2 or phospholipase C. Vallee <u>et al</u>.

POSSIBLE SOURCES OF ARACHIDONIC ACID



+ lysophosphatidic acid

Figure 5.1 The release of arachidonic acid from platelet phospholipids.

A simplified diagram of the reactions and enzymes involved in the possible mechanisms of arachidonate release. The proposed influence of adrenaline under conditions of potentiation of ADP-induced aggregation on these reactions is shown.

(1979) assessed four different phospholipase A_2 inhibitors. Mepacrine and papaverine were found to be indirect inhibitors, probably inhibiting through effects on calcium, while bromophenacyl bromide (BPB) and 2,3-dibromo (4'cyclohexyl-3'chloro)-phenyl-4-oxo-butyric acid (CB874) were thought to have direct inhibitory properties. None of these inhibitors was found to be specific for phospholipase A_2 . BPB was thought to alkylate the histidine residue of the active site of phospholipase A_2 while the mechanism of action of CB874 is still under investigation.

Phenothiazines have been shown to inhibit both phospholipase A2 and phospholipase C at different concentrations. Wightman et al. (1981) studied several phenothiazines for their inhibition of phospholipase C in resident mouse macrophages. Chlorpromazine was found to have an I_{50} of 6.3µM for phospholipase C. This is in direct contrast to the observations of Walenga et al. (1981) who investigated platelets, and found that $10 \mu M$ chlorpromazine caused a 57% increase in phosphatidic acid. Increasing the concentration of chlorpromazine to 30 µM, however, decreased the amount of phosphatidic acid formed. Phenothiazines cannot be regarded as specific calmodulin antagonists since selective receptor blockade, or inhibition of a pathway stimulated by receptor activation, has also been shown by Cocks et al. (1981) for liver and smooth muscle. Chlorpromazine has been shown to inhibit protein kinase C in some circumstances (Mori et al., 1980). Lullman et al. (1981) reported that phenothiazines can

displace bound Ca^{2+} from PI monolayers at an I_{50} of approximately $50\mu M$. Phospholipase C activity requires calcium (Billah <u>et al</u>.,1980) and at certain concentrations, phenothiazines provide calcium for phospholipase C in the cytosol. Receptor affinity and calcium availability could determine whether phenothiazines behave as inhibitors or stimulators.

BPB and tosylphenylalanine-chloromethylketone (TPCK) were shown by Vargaftig et al. (1980b) to inhibit aggregation caused by thrombin or ionophore A23187. Both were shown to inhibit phospholipase A2 activity. BPB was not specific for phospholipase A2 since it also reduced the yield of thromboxane A2 from exogenously added arachidonic acid, indicating an inhibition of cyclooxygenase or synthetase. Short incubation times, however, showed a preferential inhibition of phospholipase A2 over cyclooxygenase or thromboxane synthetase. TPCK caused a reduction in thromboxane release but this was shown to not be significant. BPB can also inhibit acetyl and acyltransferases which could affect the synthesis of platelet activating factor (PAF) (Wykle et al., 1980). BPB can also presumably inhibit the putative acyltransferase enabling mobilization of arachidonic acid between membrane phospholipids (Billah et al., 1980). It has also been shown with crude platelet extracts that BPB is an inhibitor of the PI-specific phospholipase C-diglyceride lipase pathway (Hofmann et al., 1982). Another cautionary note in the use of BPB comes from the work of Smolen and Weissman

(1980) who showed that BPB also inhibits superoxide anion generation in human polymorphonuclear leucocytes, a normal response to surface stimuli. What effect such inhibition may have in the platelet is unknown.

Serine esterase inhibitors have been shown to inhibit PI and phosphatidylcholine (PC) metabolism (Walenga <u>et al</u>., 1980). These authors concluded that serine esterase inhibitors are therefore inhibitors of phospholipase C and possibly, of phospholipase A_2 . Their evidence for inhibition of PI turnover induced by A23187 could actually be due to inhibition of phospholipase A_2 . It has subsequently been shown by Rittenhouse-Simmons (1981) that the action of phospholipase A_2 is responsible for the release of arachidonic acid from PI by A23187. Serine esterase inhibitors therefore appear to be inhibitors of both phospholipase C and A_2 .

In this chapter the effects of BPB, TPCK and phenothiazines on adrenaline potentiation of ADP is examined.

5.2 METHODS AND MATERIALS

Blood collection in citrate, platelet aggregation and thromboxane B_2 assay were carried out as before.

Bromophenacyl bromide (Koch Light) and tosylphenylalaninechloromethylketone (Serval) were dissolved in dimethylsulphoxide (DMSO) and added at one-hundredth of the plasma volume. The phenothiazines, clorpromazine and

trifluoperazine (Smith, Kline and French) were dissolved in reagent diluent (RD).

5.3 RESULTS

5.3.1 Potentiation of the ADP-response by Adrenaline in the Presence of Bromophenacyl Bromide (BPB).

BPB(10 μ M) completely inhibited the response to 8 μ M adrenaline which would otherwise cause irreversible aggregation. BPB also inhibited thromboxane B₂ formation caused by adrenaline. BPB did not completely inhibit the response to threshold ADP, but inhibited the second phase along with TXB₂ production and also reduced the first phase. BPB did not affect the shape change due to ADP. BPB also inhibited the response to threshold exogenous arachidonate, reducing but not totally inhibiting thromboxane formation. Table 5.1 summarizes an experiment exemplifying these properties of BPB. The amount of TXB₂ formed when the arachidonate response was inhibited by BPB was still in excess of that produced in response to adrenaline.

Table 5.1 The ability of bromophenacyl bromide to inhibit aggregation due to adrenaline, ADP and potassium arachidonate.

Agonist	Inhibitor	TXB ₂	Aggregation
Adrenaline(8µM)	DMSO BPB(10µM)	1.3 0	irreversible no aggregation
ADP(2.5µM)	DMSO BPB(10µM)	0.3	biphasic shape change and small first phase
Arachidonate(0,12µM)	DMSO BPB(10µM)	11 1.9	complete no aggregation

The inhibitor was added 1min prior to each agonist. TXB₂ levels are expressed as pmol/10⁷platelets. DMSO = dimethylsulphoxide(1%).

Despite these inhibitory effects on single agonists, BPB (10 μ M) did not inhibit the ability of adrenaline to potentiate ADP or arachidonate. Indeed, when potentiation occurred in the presence of BPB, thromboxane formation was increased (Fig 5.2). Not every donor exhibited the dramatic reduction in the first phase of aggregation in the presence of BPB as shown in Fig. 5.2. In all donors, however, there was a 2-3fold increase in TXB₂. BPB (100 μ M) prevented adrenaline potentiation of the ADP response, as did 50 μ M BPB for some donors, but 20 μ M BPB did not inhibit potentiation nor TXB₂ formation.

A time-course experiment revealed that the increase in TXB_2 in the presence of BPB commenced sooner than in the



Figure 5.2 Potentiation of the response to ADP by adrenaline in the presence of bromophenacyl bromide (BPB).

The agonist concentrations were

1. 0.8µM adrenaline and

2. 0.8µM ADP.

The PRP was prepared from blood anticoagulated with 10mM citrate and the platelet count was $516,000/\mu$ l. BPB or DMSO 1% was added 1 min. before the first agonist, at one-hundredth volume to achieve a concentration of 10 μ M.

absence of this inhibitor. There was measurable TXB₂ during the first phase of aggregation in the presence of BPB but not in its absence.

When high concentrations of ADP alone were used, a similar increase in TXB₂ was observed in the presence of BPB comparable to that observed when adrenaline potentiated a low concentration of ADP.

5.3.2 Potentiation of the ADP-response by Selective αreceptor Agonists in the Presence of Bromophenacyl Bromide.

Clonidine, a partial α_2 -receptor agonist, behaved in a similar manner to adrenaline and was able to potentiate the ADP-response in the presence of BPB. Thus, at concentrations of BPB, which were inhibitory for threshold concentrations of adrenaline alone, potentiation by clonidine was observed with an increase in TXB₂.

Phenylephrine, a partial α_1 -agonist, in donors showing the capacity for α_1 -potentiation, showed weak potentiation in the presence of BPB. Those donors who did not exhibit phenylephrine potentiation of the ADP-response in the absence of BPB, showed no potentiation in the presence of BPB. Adrenaline was able to potentiate TXB₂ formation in the presence of BPB in donors irrespective of platelet α_1 -receptor status.

5.3.3 <u>A Comparison of the Effects of Tosylphenylalanine</u>chloromethylketone (TPCK) and Bromophenacyl Bromide (BPB) on Adrenaline Potentiation.

TPCK was also tested as an inhibitor of adrenaline

poentiation. TPCK was found to behave in a similar manner to BPB in that increased concentrations of TXB_2 were produced when adrenaline potentiated ADP in the presence of the inhibitor. The result differed in that, at the same concentrations of inhibitors in the same donor, more TXB_2 was produced in the presence of TPCK than BPB (Fig 5.3). Comparable amounts of TXB_2 were produced in the presence of 20μ M TPCK or 50μ M BPB (in donors in whom this was not inhibitory). It can be seen that the extent of aggregation was not altered very much by these inhibitors at these concentrations, but the first phase of aggregation was decreased and this was associated with earlier TXB_2 formation.

As with BPB, increased TXB₂ production with high concentrations of ADP alone was observed in the presence of TPCK.

5.3.4 Influence of Phenothiazines on the Potentiation of the ADP-response by Adrenaline.

The phenothiazines, chlorpromazine, and trifluoperazine were examined for their influence on adrenaline potentiation. These phenothiazines inhibited thromboxane production due to ADP in a dose-dependent fashion. Adrenaline potentiation of the ADP-response occurred with very little increase in TXB₂ formation in the presence of these drugs; chlorpromazine did not inhibit adrenaline potentiation of the thromboxane-independent pathway (Fig 5.4). Aggregation by adrenaline alone at concentrations which potentiated ADP was completely



Time(min)

Figure 5.3 Potentiation of the response to ADP by adrenaline in the presence of BPB or TPCK.

The agonist concentrations were 1. 0.8µM adrenaline and 2. 0.8µMADP. The PRP was prepared from blood anticoagulated with 10mM citrate and the platelet count was 448,000/µl. Similar results were obtained with five other donors. The inhibitor concentrations were A 50µM BPB B 20µM TPCK C DMSO1% D 50µM TPCK.



Figure 5.4 Influence of chlorpromazine (CPZ) on adrenaline potentiation of the ADP-response.

The agonist concentrations were A. RD + 4µMADP. B. 0.8µMadrenaline + 4µMADP. The platelets were treated with 50µMCPZ for 1min prior to the addition of the agonists. The PRP was anticoagulated with 10mM citrate and the platelet count was 322,000/µl. In the absence of CPZ the adrenaline potentiated response produced 3.8pmol TXB₂/10⁷ platelets.

inhibited by chlorpromazine. In the presence of trifluoperazine, adrenaline also potentiated the ADPresponse independently of thromboxane formation.

5.4 DISCUSSION

The observations reported in this chapter are based on the use of phospholipase A2 inhibitors none of which, however, are specific for this enzyme. Any conclusions therefore, must be tentative. BPB has been shown to have a preferential effect on phospholipase A2 over cyclooxygenase/synthetase when incubated for 10 minutes or less with the platelets (Vargaftig et al., 1980b). Fifteen minute incubations, however, significantly affect thromboxane production from exogenously added arachidonic acid. In the present experiments however, TXA2 production from arachidonic acid was shown to be reduced after less than 10 min incubation with BPB. The fact that the thromboxane production was still in excess of that produced by adrenaline, indicates that cyclooxygenase is far from completely inhibited by BPB and the reduction may be due to other effects of BPB. It is noteworthy also that despite this amount of TXA₂ produced, aggregation was completely inhibited by BPB.

It is not known how preferential BPB might be for phospholipase A_2 over other membrane effects at these incubation times. Hofmann <u>et al</u>. (1982) reported that BPB inhibits both PI-specific phospholipase C and diglyceride lipase. Maximum inhibition of phospholipase C was achieved

with 100 μ M BPB after 15min incubation. After 1min incubation, approximately 20% inhibition was observed. 50 μ M BPB would presumably cause even less inhibition. These workers used crude platelet extracts exposing phospholipase C, otherwise in the soluble fraction of intact platelets. The efficiency of BPB in inhibiting phospholipase C and diglyceride lipase would be expected to be reduced in intact platelets. On the other hand, phospholipase A₂, being in the membrane, may be more accessible to BPB. Experiments examining the effects of BPB on exogenous phospholipase C are presented and discussed in Chapter 6. Phenothiazines are not specific, and may also affect α -receptors or α -receptor linked events (Cocks <u>et al</u>., 1981) and thus other enzymes will be inhibited.

Bearing these cautionary remarks in mind, it is of interest that BPB or TPCK not only did not prevent TXB_2 formation, but rather, stimulated an even greater increase in thromboxane production when adrenaline potentiated the ADP response, or when high concentrations of ADP alone were used. According to other reports thromboxane production should be reduced in the presence of these concentrations of BPB since it affects cyclooxygenase and/or synthetase in addition to its direct action on phospholipase A_2 . The assumption that phospholipase A_2 is the only source of arachidonic acid would lead one to predict a complete inhibition of thromboxane production by BPB.

There is, however, more than one source of arachidonic acid in the platelet. PI breakdown by phospholipase C and the action of a lipase on the resultant diglyceride can produce arachidonic acid (Bell <u>et al</u>., 1979). Phosphatidic acid could also act as a source of arachidonic acid but would also require a phospholipase A_2 for its release. It is possible that in the presence of BPB or TPCK, stimulation of phopholipase C and/or diglyceride lipase may still occur. This stimulation could be caused by potentiation of ADP by adrenaline, probably involving the a_2 -receptor, or could be caused by high concentrations of ADP alone. With the blockade of other phospholipid changes, arachidonate release could actually be stimulated via the phospholipase C-diglyceride lipase pathway(Fig5.1).

Vargaftig <u>et al</u>. (1980b) noted that TPCK did not lower thromboxane production as expected and postulated an alternative pool of arachidonic acid. Vargaftig (1976; 1977) also reported potentiation of TXA_2 formation by TPCK and by BPB under certain conditions. BPB and TPCK have similar chemical properties in that both can alkylate a histidine residue, accounting for the direct inhibition of phospholipase A_2 . The two inhibitors may also share nonspecific effects. Nevertheless, it is interesting that they both have the same effect on adrenaline potentiation of ADP-induced aggregation.

This chapter contributes the further information that the addition of adrenaline to subthreshold ADP leads to potentiation of TXA₂ formation in the presence of TPCK or

BPB. Further, clonidine, a partial α_2 -receptor agonist but not phenylephrine a partial α_1 -receptor agonist, mimics this property of adrenaline. It could be argued that adrenaline might overcome the inhibition of phospholipase A_2 directly by perhaps increasing Ca²⁺ availability. However, as BPB is a direct enzyme inhibitor, binding to the active site, it would seem more likely that adrenaline stimulates an alternative mechanism for arachidonate release such as the phospholipase C/lipase pathway (Fig.5.1). The early formation of TXA₂ would support this theory since phospholipase C is thought to be more rapidly activated than phospholipase A₂ (Billah <u>et al.</u>, 1980).

The phenothiazines, chlorpromazine and trifluoperazine exhibited characteristics which differed from BPB and TPCK with respect to adrenaline potentiation. The phenothiazines inhibited arachidonic acid release from all sources. At 50 μ M, chlorpromazine was expected to inhibit the second wave of aggregation stimulated by ADP or adrenaline and to inhibit thromboxane-dependent potentiation of ADP by adrenaline. Although this was indeed confirmed, thromboxane-independent potentiation was observed in the presence of these inhibitors. The mechanism of this thromboxane-independent aggregation would therefore seem to be independent of certain events subject to α -receptor interaction and calmodulin-dependent enzymes.

It is believed that the contribution of the diglyceride lipase pathway for arachidonate release is minor while phospholipase A₂-activated release predominates in response to thrombin (Imai <u>et al.</u>, 1982). While the predominant mechanism of arachidonate release caused by ADP is not known, it has been shown that ADP increases phosphatidic acid turnover (Lloyd <u>et al.</u>, 1973a,b). In the presence of phenothiazines, it might be possible that adrenaline is able to potentiate diglyceride kinase activity, which would increase phosphatidic acid formation due to ADP, since there is no evidence of arachidonic acid release due to diglyceride lipase activity. This must await biochemical analysis to be confirmed. Alternately, effects of adrenaline on calcium influx or mobilization need to be considered.

Rao <u>et al</u>. (1980b) reported on the influence of trifluoperazine and BPB on platelet aggregation and disaggregation. They concluded that adrenaline could correct the inhibition of phospholipase A_2 by BPB. They were, however, studying re-aggregation of dissociated platelets which is a different situation to an adrenalinepotentiated ADP response as studied here.

Despite the very different results obtained with different types of phospholipase A_2 inhibitors, some general conclusions can be drawn. Firstly, adrenaline has effects on ADP-induced aggregation which lead to increased thromboxane production, possibly through increased arachidonic acid release. Secondly, adrenaline enhances platelet responses by a pathway independent of thromboxane production as indicated in the previous chapter. In the presence of phenothiazines the potentiation was obviously

independent of thromboxane production. Stimulation of arachidonic acid release from diacylglycerol may be possible in the presence of BPB or TPCK. Diversion of diacylglycerol to arachidonic acid may account for the decrease in the extent of the primary phase and the early onset of the second phase.

An association between adrenergic receptors and phospholipase C activation has been shown in other cells. Abdel-Latif(1974) for example, found that catecholamines caused a stimulation of ³²Pi incorporation into PA and PI of the iris muscle of the rabbit. The relationship between α_1 -receptor activation and the PI effect has been discussed in Chapter 4. It is, however, the α_2 -receptor which seems to be involved in the adrenaline-platelet interaction.

Adrenaline has been shown to cause changes in phospholipids in association with release (Deykin and Snyder,1973) and more specifically, adrenaline and ADP have been found to increase production of diacylglycerol, but only after 120sec (Kawahara <u>et al</u>.,1983b). Very recently, it has been found that adrenaline alone fails to cause breakdown of PI independently of release(Clare and Scrutton,1984) but the potentiating effects of adrenaline on ADP have yet to be analysed in this manner. Since ADP alone can stimulate the PI cycle, adrenaline may potentiate ADP by enhancing the effect of ADP on PI turnover or alternately, by enhancing events consequent upon PI metabolism caused by ADP. Finally, the combined effects of

adrenaline and ADP on calcium fluxes may be involved and need to be investigated.

CHAPTER 6

THE EFFECTS OF EXOGENOUS PHOSPHOLIPASE C ON PLATELETS.

6.0 SUMMARY

Aggregation and release due to exogenous phospholipase C were examined in the presence of several inhibitors used in the previous studies of adrenaline potentiation. Aggregation was found to be prostaglandin-independent whereas a second wave of release was found to be thromboxane-dependent. Aggregation by exogenous phospholipase C in the presence of BPB, however, was inhibited by indomethacin. Calcium and c'AMP appear to regulate the response to phospholipase C.

Phospholipid analysis of platelets treated with exogenous phospholipase C in the presence or absence of the inhibitors BPB or verapamil showed that PC could be broken down without an increase in PA production. It is possible that under these conditions, diglyceride lipase is more active, releasing arachidonic acid which accounts for increased TXB₂ production in the presence of BPB. Diacylglycerol could have a pivotal role in the release mechanism, and if unavailable for the protein phosphorylation reaction (Kawahara <u>et al</u>., 1980) release could be inhibited. The first phase of release caused by exogenous phospholipase C in resuspended platelets could be due to the formation of diacylglycerol from PC.

Spermine was found to cause inhibition of the action of exogenous phospholipase C itself, preventing diacylglycerol production and hence phosphatidic acid production. Trimetoquinol did not inhibit phosphatidic acid production but caused inhibition of phospholipase Cinduced aggregation.

The results using exogenous phospholipase C in the presence of BPB or a combination of BPB and verapamil are consistent with diglyceride lipase being more active in the presence of BPB. It remains to be shown that this is the pathway by which ADP, and adrenaline potentiated ADP, trigger responses in the presence of BPB.

6.1 INTRODUCTION

The activity of endogenous phospholipase C in platelets is significant because of its role in the conversion of inositol phospholipid to diacylglycerol in stimulated platelets (Lloyd <u>et al</u>., 1972; 1973a,b; 1974; Lapetina and Cautracasas, 1979; Rittenhouse-Simmons, 1979). Diacylglycerol may be metabolized by two alternative pathways and may also be involved in protein phosphorylation (Kawahara <u>et al</u>., 1980). Through the action of a kinase, diacylglycerol can be converted to phosphatidic acid. Alternatively, the action of a lipase can supply arachidonic acid from this same intermediary (Bell <u>et al</u>., 1979). The relative importance of these two enzymes in normal platelet function is under debate.

Exogenous phospholipase C also increases the amount of phosphatidic acid in the platelet membrane (Chap <u>et al</u>., 1979). The substrate is not inositol phospholipid as in thrombin-treated platelets, but is believed to be phosphatidylcholine (PC), the most predominant phospholipid after sphingomyelin. Platelet membrane phospholipids are distributed asymmetrically. PS,PI and PE are present largely on the interior of the membrane of inactivated platelets. PC on the exterior of the membrane is an obvious site of attack. Although the phospholipid substrates for endogenous and exogenous phospholipase differ, investigation of the action of exogenous phospholipase C on platelets could still be worthwhile because of its ability to produce diacylglycerol and phosphatidic acid.

Chap <u>et al</u>. (1971) reported that phospholipase C derived from <u>Clostridium welchii</u> induces aggregation of human platelets with the production of ceramides and diaclyglycerols. Cell lysis follows slowly as indicated by lactate dehydrogenase release. Schick and Yu (1974) also studied the effects of this phospholipase C on platelets. They showed that PC and SPH were hydrolyzed, and platelet release of serotonin, ADP and platelet factor 4 occurred. These platelets were found to be intact by ultrastructural analysis after treatment with phospholipase C. It was also verified that the enzyme hydrolysis was due to the phospholipase and not due to any bacterial contaminant. PE and PS were degraded when more extensive hydrolysis occurred but PI was not found to be affected.

It is difficult to investigate endogenous phospholipase C activity since agonists known to stimulate phospholipase C also stimulate phospholipase A_2 . Thrombin (Lapetina <u>et al.</u>, 1981a) and collagen (Broekman <u>et al.</u>, 1980) for example, stimulate both phospholipases. ADP has also been shown to stimulate phospholipase C activity

within 30sec of contact with platelets (Lloyd <u>et al.</u>, 1972) with the activation of phospholipase A_2 occurring secondarily (Best <u>et al.</u>, 1980). The calcium ionophore A23187 also stimulates both phospholipases but is an inefficient stimulus for diacylglycerol generation, and in fact requires the formation of cyclooxygenase products and ADP to stimulate phospholipase C (Rittenhouse, 1984).

Another method of examining the role of phospholipase C activity involves the use of specific inhibitors. Unfortunately, no such specific inhibitor has been found to date. Walenga et al. (1980) demonstrated that several serine esterase inhibitors did, in fact, inhibit PIspecific phospholipase C, but also inhibited phospholipase A2. Since the serine esterase inhibitors blocked arachidonic acid release by A23187, an enzymatic step beyond calcium mobilization is inhibited, which in the case of A23187 is likely to be phospholipase A2 (Rittenhouse-Simmons, 1981). Phenothiazines may inhibit phospholipase C but they also inhibit phospholipase A_2 (Rao et al., 1980b). Trimetoquinol has recently been reported to be a phospholipase C inhibitor (Huzoor-Akbar et al., 1982) as have some polyamines (Nahas and Graff, 1982) but the specificity of these compounds is yet to be determined.

The following studies were carried out to characterize the aggregation of platelets and associated release achieved with exogenous phospholipase C, and examine the effect of several inhibitors which have been previously used with adrenaline. The effects of inhibitors on

phospholipase C treated platelets and adrenaline potentiated responses of platelets are compared.

6.2 METHODS AND MATERIALS

6.2.1 Preparation of Platelets

Blood was obtained and plasma prepared as decribed in Chapter 2, with citrate as the anticoagulant for the preparation of platelet rich plasma (PRP) or acid-citratedextrose for the preparation of washed platelets and resuspended platelets. Some studies were performed with PRP and some with unwashed platelets suspended in a modified Tyrode's solution. This resuspension method was essentially the same as the washing procedure described except that the wash step consisted of rinsing the top layer of the platelet pellet, instead of resuspending the whole pellet in washing fluid, thus eliminating one centrifugation step. The modified Tyrode's solution had the following composition:NaCl 8.0g/1, KCl 0.2g/1, NaHCO₃ 1.0g/1, NaH₂PO₄ 0.05g/1, MgCl₂ 0.009g/1, CaCl₂ 0.022g/1, albumin 3.5g/1, glucose 1g/1; pH 7.4, 37°C.

6.2.2 Platelet Studies

Aggregation, measurement of ATP release, TXB₂ radioimmunoassay and phospholipid analyses were undertaken according to the methods described in Chapter 2.

6.2.3 Materials

Phospholipase C, Type I was obtained from Sigma and had a specific activity of 8.1U/mg solid. Thrombin(Parke-

Davis, bovine origin) was also used as an aggregating agent.

The inhibitors used were aspirin, indomethacin, BW755C, verapamil (Isoptin), p-bromophenacyl bromide, phenoxybenzamine, dibenamine, spermine, chlorpromazine, promethazine and trifluoperazine. Their sources and usage were as described in Chapter 2.

6.3 RESULTS

6.3.1 Characteristics of the Response to Phospholipase C

Exogenous phospholipase C (0.1 mg/ml) caused aggregation in PRP after a variable delay of up to 5min (Fig. 6.1). Resuspended platelets were more sensitive to phospholipase C with a threshold of 5-10µg/ml, and the aggregation response was faster (Fig. 6.1). Release of nucleotides in PRP commenced slowly and reached a peak after aggregation was almost complete (Fig. 6.1). In contrast to PRP, the release reaction in resuspended platelets occurred in two phases. The first phase was very rapid, coinciding with the shape change, and the second phase appeared about one minute after, or later if the overall reaction was slow for that particular donor. The release reaction was more extensive in resuspended platelets than in PRP (Fig 6.1). Aggregation of platelets due to phospholipase C in either medium was associated with the formation of small amounts of TXB2. For example, in PRP, phospholipase C (0.1mg/ml) caused the formation of $0.2-2.0 \text{ pmol TXB}_2/10^7 \text{ platelets}.$



Figure 6.1 Platelet aggregation and release caused by exogenous phospholipase C.

Platelets were suspended in platelet rich plasma(B) or in modified Tyrode's solution (see Methods)(A). Aggregation(upper traces) was acheived with 0.1mg/ml (0.8U/ml)(B) and 0.01mg/ml(0.08U/ml)(A) of phospholipase C Release of nucleotide was assessed in comparison to ATP standards using a lumiaggregometer(lower traces). TXB₂ levels obtained on completion of aggregation were A,0.6pmol/10⁷ platelets and B,1.0pmol/10⁷ platelets.

6.3.1.1 The Effects of Cyclooxygenase/Synthetase Inhibitors.

Neither aspirin (5mM) nor indomethacin (10⁻⁴M) nor BW755C (10µg/ml) by themselves had any effect on aggregation caused by exogenous phospholipase C (0.1mg/ml). These inhibitors were found to be effective in preventing formation of TXB₂. Following the inhibition of cyclooxygenase, however, other inhibitors were found to be more effective as described below for BPB. Indomethacin inhibited the second peak of nucleotide release from resuspended platelets. The first peak was slightly reduced.

6.3.1.2 The Effects of Verapamil, a Calcium Channel Blocker.

In PRP, verapamil (55 μ M) did not inhibit aggregation due to exogenous phospholipase C (0.1mg/ml), but reduced the amount of TXB₂ formed. Higher concentrations of verapamil(110 μ M and 165 μ M) decreased the rate of aggregation and reduced TXB₂ formation. In PRP, verapamil (165 μ M) inhibited the release of nucleotide. With resuspended platelets, the first phase of release was markedly reduced, while the second was increased (Fig 6.2). Nifedipine (10 μ M) had a similar effect to verapamil (165 μ M) (results not shown).

6.3.1.3 The Effects of Bromophenacyl Bromide (BPB)

In both resuspended platelets and PRP, BPB (100µM) partially inhibited aggregation caused by exogenous phospholipase C while increasing TXB₂ formation (Fig 6.3).



Figure 6.2 The effects of verapamil on aggregation and release due to exogenous phospholipase C.

Platelets were suspended in modified Tyrode's solution and were aggregated with 0.05mg/ml phospholipase C without inhibitor(A) or with 75µg/ml verapamil(B)(upper traces). Release of nucleotide was assessed in comparison to standard ATP using a lumiaggregometer(lower traces). TXB₂ levels were A,0.5pmol/10⁷ platelets and B,0.3pmol/10⁷ platelets.



Figure 6.3 The additive inhibitory effect of BPB and indomethacin on aggregation due to phospholipase C.

Platelet rich plasma (400,000 platelets/µl)prepared from citrated blood was incubated for 1min with (A) $10^{-4}M$ indomethacin + 100μ M BPB, (B) 100μ M BPB and (C) DMSO. Phospholipase C(0.1mg/ml) was added at the arrow. ND=not detected.

Preincubation with a combination of indomethacin and BPB completely inhibited TXB_2 formation(Fig.6.3). The combination of verapamil and BPB, however, did not inhibit TXB_2 formation even though verapamil, by itself, reduced TXB_2 formation caused by exogenous phospholipase C. This is in contrast to indomethacin which invariably prevented TXB_2 formation.

BPB decreased the release of nucleotide from resuspended platelets, and altered the pattern of release (Fig 6.4). Only one phase of release occurred and the larger the extent of inhibition of aggregation, the larger was the inhibition of the release reaction.

6.3.1.4 Other Inhibitors

Several inhibitors known to affect calcium dependent reactions were tested for their effects on aggregation caused by exogenous phospholipase C. Mepacrine (quinacrine)(30µM) caused a slight inhibition of aggregation with a complete inhibition of TXB₂ formation. The phenothiazines, chlorpromazine, promethazine and trifluoperazine were not inhibitory at 100µM. There was, however, some reduction in release of nucleotide in PRP in the presence of phenothiazines. Phenoxybenzamine(40µM) and dibenamine (40µM) also reduced release but not aggregation. In resuspended platelets, this reduced release of nucleotide occurred in one peak only, well after the shape change, as aggregation was proceeding. Prostacyclin (10µg/ml) and theophylline(1mM) partly inhibited aggregation if the concentration of phospholipase C was so



Figure 6.4 The effects of bromophenacyl bromide on aggregation and release due to exogenous phospholipase C.

Platelets were suspended in modified Tyrode's solution and were aggregated with 0.05mg/ml phospholipase C without inhibitor(A) and with 50µM BPB(B)(upper traces). Release of nucleotide was assessed in comparison to standard ATP using a lumiaggregometer(lower traces). TXB₂ levels were 0.5pmol/10⁷ platelets for both traces.

reduced that aggregation only occured after a delay of about 4min. The inhibitory effects of these agents, although small when used singly, were additive when combined.

6.3.1.5 Spermine and s-Trimetoquinol

s-Trimetoquinol (1mM) inhibited aggregation due to exogenous phospholipase C (10µg/ml). The polyamine, spermine (10mM) also completely inhibited phospholipase C mediated aggregation. These observations were examined further by the analysis of the changes in phospholipids caused by phospholipase C in the presence of these inhibitors (see below). Both s-trimetoquinol and spermine inhibited thrombin-induced aggregation and the accompanying phospholipid changes caused by thrombin in the presence of these inhibitors were also analyzed.

6.3.2 Phospholipid Changes Due to Exogenous Phospholipase C

Washed platelets were aggregated with phospholipase C and membrane phospholipids were extracted. The phospholipids were separated by paper chromatography and estimated by phosphorus analysis and sometimes following ³H-arachidonic acid incorporation. The phospholipids analyzed were PC, SPH, PI, PE, PS and PA. The distribution of these lipids on a typical chromatogram is given in Fig 6.5. The lysolipids, lysophosphatidylcholine (LPC) and lysophosphatidylethanolamine (LPE) were tentatively identified in addition to these.

The major change caused by phospholipase C was an increase in PA. A typical phospholipid analysis is given



X

X

Figure 6.5 Chromatograms of unstimulated(A) and phospholipase C-activated(B) platelets.

Chromatograms were run on Whatman SG-81 paper, first in chloroform:methanol:5N ammonia(64:34:4) and then in chloroform:diisobuylketone:pyridine:methanol:acetic acid:formic acid:water(23:20:25:20:8:1:3). The phospholipids are identified(see Ch.7) as:1.sphingomyelin SPH 2.phosphatidylcholine PC 3.phosphatidylserine PS 4.phosphatidylinositol PI 5.phosphatidylethanolamine PE 6.phosphatidic acid PA 7.lysoPC 8.lysoPE. in Table 6.1. PA increased 10-fold in 3min compared with the area (unstained) in a corresponding control chromatogram of unstimulated platelets. PC is the lipid measurably hydrolyzed while the other lipids appear unaffected. Mean PA production at 3min and 5min for three experiments is shown in Fig 6.6.

Table 6.1 Phospholipid Analysis by Phosphorus Content for Unstimulated and Phospholipase C-treated Platelets.

% TOTAL PHOSPHORUS

Phospholipid	Unstimulated	Phospholipase C
SPH	16.5	16.3
PC	32.3	27.3
PS	9.1	9.0
PI	2.9	3.6
PE	17.9	19.3
PA	0.34	3.8
LPC	1.7	1.0
LPE	2.3	2.8

These are the results of a single experiment in which 5.7×10^9 platelets in 1500μ l were treated with phospholipase $C(10 \mu g/ml, 0.08 U/ml)$ for 90 sec. Phospholipids were extracted and chromatographed according to the methods described.

6.3.3 The Effects of Inhibitors on Phospholipid Changes

due to Exogenous Phospholipase C

The effects of BPB and verapamil on changes in PC and PA changes caused by phospholipase C are shown in Table 6.2. Only PC and PA are shown as there were no detectable changes in the other phospholipids. The data includes estimation of phospholipids by ³H-arachidonic acid incorporation (2 experiments) and by phosphorus assay (4 experiments). Changes caused by phospholipase C in the



Figure 6.6 Production of phosphatidic acid during platelet aggregation due to phospholipase C.

Platelets were washed (methodB) and resuspended at $2-3\times10^9$ platelets/ml. PA was estimated by phosphorus assay of the spot separated on two-dimensional chromatography and shown in terms of %total phosphorus as the mean of three experiments ±SD depicted as bars.

....
presence of BPB (50µM) or verapamil (50µg/ml) or a

combination of these two inhibitors are shown.

Table 6.2 Changes in PC and PA due to Exogenous Phospholipase C(PLC) in the Presence of BPB and/or verapamil. Z TOTAL PHOSPHORUS

	τ	Jnstim.	PLC		PLC	PLC	PLC
	aledero.				+BPB	+verap.	+BPB/verap
EXD.	PL	c	3 5		3 5	3 5	3 5min
#1	PC	47.4	36.5	2	27.6	36.8	
	PA	0.28	1.94		1.94	3.24	
#2	PC	44.3	41.0 40	.0 3	38.3 35.9	41.7 32.0	42.5 33.5
	PA	0.05	2.17 5	.81	2.32 3.90	2.56 2.84	2.35 3.70
#3	PC	45.0	44.2 40	.1 4	10.3 33.7	38.2 39.7	34.6 33.0
* 3	DA	1.10	4.15 7	.81	4.13 9.98	4.56	5.60 6.83
* 4	PC	40.0	38.0 37	. 4 3	32.9 29.2	30.8 42.3	35.4 34.2
T **	DA	0.46	1.06 5	.48	4.82 5.21	2.18 5.03	3.80 4.73
MEAN	PA DC	42 1	11 1 39	2 :	37.2 32.9	36.9 38.0	37.5 33.6
MEAI	ten	2 7	3.1 1	5	3.8 3.4	5.6 5.4	4.3 0.6
MPA		0.54	2 46 6	36	3.76 6.36	3.10	3.92 4.89
MEAL	ten	0.54	1 56 1	26	1.29 3.20	1.30	2.30 1.70
	-5D	0.55	1	• 20	1.25 5.20		
. 3	ACHTD	ONTO NO	TD				
to n-AK	DC	A2 5	143 0 40	0	43.0 35.0	41.0 34.0	36.0 32.0
# 3	PC	43.5	1 26 2	90	1.50 3.00	1.70 2.00	1.60 2.20
	PA	0.15	1.20 2		1.50 5.00		
		1	51 0 20		52 0 31 0	57.0 33.7	24.2 23.1
# 4	PC	41.0	51.0 25	· cal	1 70 3 68	1.60 3.21	0.58 1.88
	PA	0.12	0.9/ 3	.04	1.70 5.00	1.00 5.2	
			1.2	-	17 5 22 0	10 0 22 0	30.1 27.6
MEA	N PC	42.6	47.0 34	.0 4	41.5 33.0	11 2 0 21	8370
	İSD	1.3	5.7 7	• /	0.4 2.8	1 70 2 61	1 10 2 04
MEA	N PA	0.12	1.20 3	.27	1.60 3.34	0.07.0.00	0 72 0 22
	±SD	0.06	0.21 3	.27	0.14 0.48	0.01 0.86	0.12 0.23

In each experiment, $2-3x10^9$ platelets/ml were treated with phospholipase C(10µg/ml) for 3 or 5min. The last two experiments were analyzed for ³H-arachidonic acid incorporation in addition to phosphorus analysis. Exp.#1 is excluded from the mean±SD because of limited data. BPB(50µM) and verapamil(50µg/ml) were added 1min before phospholipase C.

Using phosphorus estimation, it was found that exogenous phospholipase C achieved a greater hydrolysis of PC in the presence of BPB than in the absence of this inhibitor. This was found in each experiment and is reflected in the mean. The large standard deviations are due to wide differences in responsiveness of different platelet preparations used in each experiment. PA production does not appear to account for this reduction in PC. Breakdown of PC by exogenous phospholipase C was also increased in the presence of verapamil, but not by as much as in the presence of BPB. PA production was not always increased in the presence of verapamil or BPB. In the presence of a combination of BPB and verapamil, a similar pattern emerged as in the presence of each inhibitor alone, namely, increased breakdown of PC but no further increase of PA production.

With ³H-arachidonic acid incorporation, however, it was found that, in the presence of the combination of BPB and verapamil there was a greater loss of arachidonic acid from PC which did not appear in PA, compared to the effects of exogenous phospholipase C in the presence of each inhibitor singly. The changes in ³H-arachidonic acid incorporation due to exogenous phospholipase C in the presence of these inhibitors, in general, revealed a different pattern to the results obtained using phosphorus estimation. No change in ³H-arachidonic acid incorporation in PC was evident after 3min and there was only a small decrease in radioactivity by 5min. Changes in the presence of inhibitors were, therefore more difficult to detect.

Phospholipid analysis of phospholipase C-treated platelets in the presence of trimetoquinol or spermine is

shown in Table 6.3. Spermine, and to a lesser extent, trimetoquinol, inhibited the breakdown of PC by exogenous phospholipase C. The two inhibitors differed in their effect on PA production in that spermine (10mM) inhibited PA production while trimetoquinol (2mM) did not.

Table 6.3 Changes in PC and PA due to Exogenous Phospholipase C in the presence of Trimetoquinol or Spermine.

% TOTAL PHOSPHORUS

PL	Unstimulated	Phospholipase (PLC)	С	PLC + trimetoquinol (2mM)	PLC + spermine (10mM)
PC PA	31.4 0.60	18.2 4.9		24.6 5.3	34.4 0.40

The results shown are from one experiment, and were confirmed in another. 3.4×10^9 platelets were treated with phospholipase C(100µg/ml) for 8min in the presence or absence of each inhibitor, preincubated for 1min.

6.3.4 The Effects of Inhibitors on Phospholipid Changes

due to Thrombin

Washed platelets (method A) were aggregated with thrombin in the presence of the same inhibitors previously examined for their effects on phospholipase C-induced phospholipid changes. Two experiments were performed, one with 0.14U thrombin/10⁸ platelets and another with 0.08U thrombin/10⁸ platelets. Phospholipids were quantitated by ³H-arachidonic acid incorporation and by phosphorus assay. BPB partially inhibited aggregation and verapamil almost totally inhibited aggregation by thrombin. Total inhibition of aggregation by thrombin was achieved with the combination of BPB and verapamil. In Table 6.4 it can be seen that PA production due to thrombin was not prevented by BPB nor by verapamil. Also PI breakdown due to thrombin was not greatly affected by these inhibitors. Since only two experiments were performed, this data cannot be analyzed further.

Table 6.4 Changes in PC, PI and PA due to Thrombin in the Presence of BPB and/or Verapamil.

%³H-ARACHIDONIC ACID

PL	Unstimulated	Thro	ombin	Thrombin		Thrombin		Thrombin	
				+BPB		+verap.		+BPB/verap.	
		3	5	3	5	3	5	3	
PC	55.9	62.0	50.3	55.8	58.1	59.0	51.0	58.4	
PI	8.73	5.6	3.7	6.8	5.9	5.4	4.9	6.8	
PA	0.21	3.6	4.2	2.2	4.2	3.8	4.3	2.6	
						10			

%TOTAL PHOSPHORUS

PC	43.0	44.8	38.9	45.5	39.7	38.8	38.6	42.8
PI	5.0	3.0	3.5	3.4	2.6	3.4	4.4	3.1
PA	0.47	1.9	2.5	2.7	2.0	3.9	2.9	1.8

 3.5×10^9 platelets/ml were treated with thrombin(5U/750µl) and incubated for 3 or 5min, in the presence or absence of BPB(50µM) or verapamil(50µg/ml) or a combination of the two.

Phospholipid analysis was also performed on thrombinactivated platelets (0.4 U/10⁸ platelets) in the presence of trimetoquinol or spermine. The results are shown in Table 6.5. With phosphorus analysis, 100mM spermine prevented PI breakdown and PA formation, whereas 10mM trimetoquinol did not inhibit PI breakdown nor PA formation, even though both caused inhibition of

aggregation.

Table 6.5 Changes in PI and PA due to Thrombin in the Presence of Trimetoquinol or Spermine.

% TOTAL PHOSPHORUS

PL	Unstimulated	Thrombin	Thrombin + trimetoquinol (10mM)	Thrombin + spermine (100mM)
PI	6.8	5.3	5.2	6.3
PA	0.38	1.9	2.5	0.56

2.4x10⁹ platelets/ml were treated with thrombin($8U/750\mu$ 1) and incubated for 3min in the presence of trimetoquinol(10mM) or spermine (100mM).

6.4 DISCUSSION

There have been several reports on the effects of phospholipase C (derived from <u>Clostridium welchii</u>) on human platelets <u>in vitro</u> (Schick and Yu, 1974; Chap <u>et al</u>., 1977; Kawahara <u>et al</u>., 1980; Huzoor-Akbar <u>et al</u>., 1982). In agreement with these reports, the present data shows that phospholipase C causes aggregation and release of nucleotides from platelets. Aggregation was independent of prostaglandin synthesis but was partially inhibited by verapamil, mepacrine, phenothiazines and agents which increase c'AMP. It is suggested, therefore, that calcium and c'AMP may regulate the platelet response to phospholipase C.

Phospholipase C caused aggregation and release in both PRP and resuspended platelets although higher concentrations of the enzyme were required in PRP, and the pattern of release in PRP differed from that in modified Tyrode's solution. The reduced sensitivity of PRP may be attributable to the presence of other lipids and to protein in the plasma. The first and more rapid phase of release of nucleotide from resuspended platelets corresponds in time with 5-hydroxytryptamine release by phospholipase C reported by Schick and Yu (1974). This first phase of nucleotide release was inhibited by verapamil (165μ M) or by BPB (50μ M). It could be speculated that, in the presence of BPB, diacylglycerol is diverted, possibly by the lipase (Bell <u>et al</u>., 1979) such that it is unable to be linked to protein phosphorylation for the release reaction (Kawahara <u>et al</u>., 1980). This is supported by the data showing increased breakdown of PC in the presence of BPB, without conversion to PA.

If diacylglycerol is involved in the action of exogenous phospholipase C it would have to enter the platelet after the action of phospholipase C on externally located PC. Phospholipase C does not enter the cell but instead utilizes the externally located substrate as indicated by the experiments with ³H-arachidonic acid. Phospholipase C caused very little change in ³H-arachidonic acid in PC, while phosphorus analysis indicated significant hydrolysis of PC. These observations indicate that the substrate for phospholipase C is PC located in the outer membrane leaflet. Inner membrane PC contains a higher proportion of arachidonic acid while PC in the outer

membrane leaflet contains oleate and linoleate (Rittenhouse-Simmons and Deykin, 1981).

Both calcium mobilization and diacylglycerol production seem to be necessary for the full physiological response of the platelet(Kaibuchi <u>et al</u>.,1983). Verapamil may inhibit the release reaction by acting as a calcium antagonist which inhibits calcium-dependent reactions including protein phosphorylation.

The second phase of nucleotide release from resuspended platelets is thromboxane-dependent since it was inhibited by indomethacin. Aggregation is not thromboxanedependent. Nevertheless, small amounts produced may contribute towards aggregation associated with the second phase of release. Thromboxane seems to contribute more to aggregation occurring in the presence of BPB since this was more sensitive to inhibition by indomethacin. Nevertheless, increases in TXB2 observed in the presence of BPB or BPB and verapamil combined, were not able to overcome the inhibition of aggregation. A step beyond the formation of thromboxane therefore seems to be inhibited. If diacylglycerol lipase is indeed more active than the kinase in the presence of BPB, with resultant release of arachidonic acid, then the present observations can be explained.

The effects of BPB and verapamil on phospholipid changes caused by thrombin differ from their effects on phospholipid changes caused by exogenous phospholipase C. Thrombin activates endogenous phospholipase C in platelets,

forming diacylglycerol from phosphatidylinositol-4,5bisphosphate (Broekman, 1984). Whereas exogenous phospholipase C was more active in the presence of BPB than in its absence, the activity of endogenous phospholipase C following thrombin activation was unaffected by BPB. The two phospholipases in question act on different substrates: PC in the case of the exogenous enzyme and PI in the case of thrombin activated phospholipase C. Thus there may be important differences in diacylglycerol formation and metabolism when platelets are activated by thrombin or exogenous phospholipase C, the latter enzyme requiring entry of diacylglycerol into the platelet.

Most of the agents examined for their ability to inhibit exogenous phospholipase C were only partially effective. Spermine(10mM) and trimetoquinol(1mM) were the only agents found to be completely inhibitory for exogenous phospholipase C. These two inhibitors differed in their effects on phospholipids. Spermine, but not trimetoquinol inhibited PA formation due to exogenous phospholipase C. Perhaps trimetoquinol inhibits a step in the activation pathway beyond diacylglycerol and PA formation. An interesting possibility is that trimetoquinol inhibits protein kinase C, since it does not inhibit diacylglycerol formation. Inhibition by spermine can be attributed to inhibition of diacylglycerol formation which is important for aggregation and release by phospholipase C.

The results of this chapter suggest that BPB could indeed, be affecting the phospholipase C pathway, perhaps

enabling diacylglycerol lipase to be more active. It would be interesting to examine the effects of BPB on the adrenaline potentiated response to ADP by platelets in terms of such phospholipid changes. If adrenaline can influence this pathway, it may well be of considerable physiological and pathological significance. Chapter 7

THE EFFECT OF ADRENALINE ON PLATELET MEMBRANE PHOSPHOLIPIDS.

7.0 SUMMARY

The mechanism of prostaglandin-independent adrenaline potentiation may involve phospholipid changes, or alternately, could be associated with increased calcium uptake due to adrenaline. Since PI metabolism has been found to alter as a result of stimulation by other agonists, it was thought that adrenaline effects on PI metabolism should be examined.

Two solvent systems for a paper chromatographic technique for the separation and quantitation of PA and PI were compared. The second system was found to give a well resolved, compact spot compared to the more diffuse spot of the first system. This method was used for some preliminary experiments, examining adrenaline potentiation of ADP and thrombin.

The results are preliminary and therefore inconclusive but focus on the PI cycle and mobility within the different polyphosphoinositides as areas worthy of further investigation.

7.1 INTRODUCTION

Adrenaline potentiation of other platelet agonists <u>in</u> <u>vitro</u> may reflect the proposed important biological role of adrenaline in the preparation of platelets to react efficiently where required in wound repair. Besides an effect on arachidonic acid metabolism, the present work indicates that adrenaline potentiation can be independent of prostaglandin synthesis. Adrenaline, by its binding to a₂-receptors (Grant and Scrutton,1979) possibly causes some kind of membrane change involving calcium mobilization(Rao et al., 1981a; Brass and Shattil, 1982). This calcium movement could be associated with phospholipid changes (Michell, 1975) or could be independent of phospholipids. Alterations to phospholipids have been observed during aggregation of human, horse and rabbit platelets induced by thrombin, collagen and ADP(Lloyd <u>et al.</u>, 1973b; 1974; Broekman <u>et al.</u>, 1980; Lapetina <u>et al.</u>, 1981a).

Adrenaline has been found to increase PI synthesis in platelets, but only following completion of the release reaction (Deykin and Snyder, 1973) and not in the presence of aspirin (Clare and Scrutton, 1984). It is, however, the immediate post-receptor events in adrenaline activation which are of interest in the potentiation mechanism since potentiation can be acheived after only fifteen seconds or less prior contact with adrenaline. Both phospholipid turnover and changes in phospholipid proportions need to be Since PI is capable of cycling (Lapetina et assessed. al., 1981b) resynthesis is possible so that while PI can participate in a platelet reaction, its overall level may not alter, as was found for ADP (Lloyd et al., 1972). The evidence reported so far in this thesis indicates that adrenaline may affect phospholipase C-mediated PI metabolism when the ADP response is potentiated and this chapter focusses on the effects of adrenaline on this

aspect. These investigations, however, are only of a preliminary nature. Details of the development of the chromatographic procedure employed are also included.

A paper chromatographic method was chosen for the separation of phospholipids. With this technique, PI and PA could be separated and quantitated, but diacylglycerol could not. PI was considered to be important to assay for reasons which have been discussed previously. PA is a product of PI breakdown and its accumulation has been shown to be important in the response to thrombin (Lapetina <u>et</u> <u>al.</u>, 1981). Increases in PA indicate that the enzyme diglyceride kinase is active as opposed to digyceride lipase. Silica gel-loaded paper has been reported to be suitable for separation of lipids from tissue extracts (Wuthier, 1966).

7.2 METHODS AND MATERIALS

Platelet preparation, aggregation, chromatography, ³Harachidonic acid incorporation and phosphorus assay were carried out according to the methods described in Chapter 2. The platelets were washed using method B with the inclusion of prostacyclin in each centrifugation. Aggregation was performed in a large cuvette in the aggregation module so that a sufficient number of platelets could be used to obtain measurable amounts of phospholipid. At least 1x10⁹ platelets(approx.30µg P) were required per incubate. For most experiments, 750µl platelet suspension containing approximately 3x10⁹ platelets was employed. The lipid extraction procedure was modified from Bligh and Dyer(1959) to allow for the small volumes of each incubate. SG-81 paper from Whatman was used for twodimensional chromatography to separate the phospholipids. The first pair of solvent mixtures employed (System A) used by Kramer <u>et al.(1982)</u>, and modified from Wuthier(1966), was found to be unsuitable, especially for the recovery of measurable phosphatidic acid and was later substituted with a second pair of solvent mixtures (System B). A full comparison of chromatograms obtained with both pairs is reported in the Results section of this Chapter(7.3).

Phospholipids were quantitated by phosphorus assay (Rouser <u>et al</u>.,1966) and expressed as a percentage of total phosphorus extracted from the platelet incubate. Phospholipid turnover was assessed by determining the incorporation of ³H-arachidonic acid.

The aggregating agents employed were adrenaline, ADP, and thrombin. The inhibitor used was BPB. Phosphatidic acid standards used included synthetic dioleoyl phosphatidic acid-Na salt(Sigma), dipalmitoyl phosphatidic acid-free acid(Sigma), lecithin-derived phosphatidic acid(Serdary) and phosphatidic acid(Supelco). Other standard phospholipids, PI(Supelco), PC, PE, PS and SPH(Sigma) were also used to identify chromatographic spots.

7.3 RESULTS

7.3.1 Chromatography with Solvent System A.

Two different solvent systems for two-dimensional chromatography on SG-81 paper were tested. The first system(A) separated platelet membrane lipids according to the pattern shown in Fig 7.1. PI resolved well in this system, but PA could not be observed, presumably because it constitutes such a small proportion of total phospholipids. The maximum lipid phosphorus load of this system is 10µg(Wuthier,1966) of which less than 0.1%(0.01µg) would be PA. This amount would not stain with rhodamine 6G, especially if the spot was diffuse. The localization of PA was therefore attempted with the use of commercial standards in quantities which would stain. Even so, the localization of platelet PA would not necessarily be coincident with PA derived from a source other than platelets(Wuthier,1976).

The mobilities of PA from three different sources in each phase of the solvent system are shown in Fig 7.2. All of these standards were unsatisfactory. The Serdary sample separated into three different spots in the second dimension. Dipalmitoyl-PA showed no movement at all in the second dimension and dioleoyl-PA moved high in both phases. PA moving rapidly in the second phase would overlap with cardiolipin and PE. Platelet PA could not be localized by this method.

Washed platelets(6x10⁹/ml) were treated with thrombin (0.2U/10⁸platelets) for 2min, extracted and chromatographed in parallel with an extract from unstimulated platelets. The PI content of control and thrombin-stimulated platelets



Figure 7.1 Chromatogram of platelet membrane lipids using solvent system A.

The lipids were extracted by a modification of Bligh and Dyer(1959). The chromatogram was run on Whatman SG-81 paper first in chloroform:diisobutylketone:methanol:acetic acid: water(45:30:15:20:4) for the long phase, and secondly in chloroform:diisobutylketone:pyridine:methanol:water(30:25: 35:25:8). The spots were stained with rhodamine 6G and identified as 1.SPH 2.PC 3.PS 4.PI 5.PE and 6.?PA by their colour under UV light and their comparison to standards. x=origin.



Figure 7.2 The mobilities of standard phosphatidic acids in each phase of solvent system A.

The PA samples were: A. synthetic dioleoyl PA-Na salt (Sigma) B. synthetic dipalmitoyl PA-free acid(Sigma) and C. egg lecithin derived PA(Serdary). The effect of the first phase solvents are shown in the upper chromatogram and the effect of the second phase solvents are shown in the lower chromatogram, run separately. The solvents of the two phases are as given for Fig 7.1. is shown in Table 7.1 together with SPH and PC, and a possible PA. The results for the suspected PA spot were highly variable and the expected increase due to thrombin stimulation was not observed. The amount of supposed PA was also much higher than the expected levels. Phospholipase C treated platelets were then used as a source of standard PA together with an alternative solvent system which was expected to yield a more compact, well resolved spot for PA.

Table 7.1 Phospholipid Content of Unstimulated and Thrombin-stimulated Platelets.

Phospholipid	Solvent System A Unstimulated	Stimulated
SPH	18.8%(1.6)	15.5%(2.4)
PC	36.9%(0.78)	37.8%(1.8)
PI	3.8%(0.32)	2.2%(0.71)
?PA	1.2%(1.7)	1.4%(1.2)

Results are expressed as percentage total phosphorus and are the mean(±SD) of three experiments.

7.3.2 Chromatography with Solvent System B.

The chromatograms of platelet membrane phospholipids using solvent system B are shown in Fig 7.3 and Fig 7.4. Chromatography of phospholipids of both thrombinstimulated(Fig 7.3) and exogenous phospholipase Cstimulated platelets(Fig 7.4) revealed the appearance of a spot on the lower right hand corner not present in unstimulated platelets. This spot(numbered 6 in the Fig) is consistent with the position of PA in a chromatogram of rat tissue phospholipids illustrated by Wuthier(1976). A PA standard(Supelco) was subsequently found to



Figure 7.3 Chromatograms of platelet membrane lipids from unstimulated and thrombin-stimulated platlets.

Washed platelets (3X10⁹/ml) were treated with thrombin (0.1U/10⁸platelets) for 5min. The chromatograms were run on Whatman SG-81 paper, first in chloroform:methanol:5N NH₄OH (64:34:4) and secondly in chloroform:diisobutylketone: pyridine:methanol:acetic acid:formic acid:water(23:20:25: 20:8:1:3). The spots were stained with rhodamine 6G and identified as 1.SPH 2.PC 3.PS 4.PI 5.PE 6.PA 7.lysoPC 8.lysoPE. x=origin.



Figure 7.4 Chromatograms of platelet membrane lipids from unstimulated and phospholipase C-stimulated platelets.

Washed platelets($3X10^9/ml$) were treated with phospholipase C($10\mu g/ml$) for 2min. The chromatograms were prepared as described for Fig 7.3. x=origin.

*

chromatograph in this position also. Thrombin treatment gave rise to two additional spots(7,8) considered to be lysolipids:lysoPC and lysoPE resulting from phospholipase A₂ activity.

This solvent system therefore produced a chromatogram with well separated PI and PA spots. The PA spot had the advantage of being compact with little chance of overlap with any other phospholipids. The solvent system B was therefore considered the method of choice. The chromatographs of SPH, PC, PS, PE and PI standards using solvent system B are shown in Fig 7.5 and the identification of each platelet phospholipid was thus verified.

7.3.3 The Effects of Adrenaline on ADP-induced Metabolism of PI.

The results of three experiments, in which phospholipids were analyzed for ³H-arachidonic acid incorporation and phosphorus content, in response to a combination of adrenaline and ADP both in the presence and absence of BPB are shown in Table 7.2. These results are difficult to analyze for several reasons. First of all, the degree of incorporation of arachidonic acid in PA in unstimulated platelets is highly variable and stimulation by agonists causes variable effects. Stimulation may either decrease or increase this basal level of incorporation, or remain much the same, as in experiment #1.

First Solvent Front X X PA PI Second Solvent Front X X SPH PC Χ X PE PS

Figure 7.5 Chromatograms of standard phospholipids using solvent system B.

The standard phospholipids were chromatographed as described for Fig 7.3. The samples were: PI(Supelco)5µg PA(Supelco)5µg PC,SPH,PS,PE,(Sigma)10µgea. x=origin.

Secondly, there is no consistent pattern of changes in PA nor PI. The first two experiments indicate a small increase in ³H-arachidonic acid incorporation in response to a combination of adrenaline and ADP. Experiment #3, on the other hand, shows a decrease, but, in this experiment, the degree of incorporation was high in the unstimulated platelets. Experiments #2 and #3 show an increase in incorporation by adrenaline in the presence of BPB, a result paralleled by increases in amount of PA as estimated by phosphorus content. There is very little change in incorporation in PI, except for a general trend toward increased incorporation in platelets in the presence of BPB, especially when incubated with ADP or adrenaline combined with ADP. Two experiments also showed a small increase in ³H-arachidonic acid incorporation in PI in platelets aggregated with adrenaline combined with ADP in the absence of inhibitor.

The phosphorus estimations for PA are also variable. One experiment showed an increase in PA while the other showed no change, when the platelets were treated with ADP for 30 sec. The phosphorus estimations for PI are difficult to interpret.

Table 7.2 Phospholipid Changes During Adrenaline Potentiation of ADP-induced aggregation in the Presence or Absence of BPB.

%³H-ARACHIDONIC ACID

			N	lo inh	ibito	r	BPE	$(50\mu M$	1)	
Exp.No.	PL	С	Adr. 15s	Adr. 45s	ADP A 30s	dr/ADP 15/30s	Adr. 15s	Adr. 45s	ADP A 30s	dr/ADP 15/30s
#1	PA PI	0.23	0.27	0.32	0.25	0.40 6.32	0.19 6.93	0.21 7.14	0.32 7.82	0.38 9.30
#2	PA PI	0.69	0.57	-	1.40 5.71	1.55 7.11	2.13	-	0.406.08	1.55 7.13
#3	PA PI	2.34 6.38	0.99 6.80	0.457.21	0.90 5.98	0.49 7.38	2.86	0.59 6.17	1.45 8.34	0.41 7.01
%Phosph #2	oru PA PI	s 1.82 3.47	0.14		5.10 3.30	0.65 2.68	2.03		0.38 3.80	1.40 7.20
#3	PA PI	0.10 2.32	0.11	0.09	0.07 2.96	0.10 2.69	0.60 2.18	0.10 2.37	0.08	009 2.89
gr 317	ah i	dania		inco	rnora	tion is	chow	n for	thre	0

%³H-arachidonic acid incorporation is shown for three experiments and %phosphorus for the last two experiments. 1-2x10⁹platelets/ml were treated with adrenaline(Adr.) and/or ADP as indicated at the following concentrations: #1 adrenaline 10⁻⁵M, ADP 10⁻⁵M. #2 adrenaline 5X10⁻⁶M, ADP 10⁻⁵M. #3 adrenaline 10⁻⁴M, ADP 10⁻⁴M. Fibrinogen(0.3mg/ml) was added 1min prior to the agonists. Each figure is the result of a single estimation. PL=phospholipid, s=sec, C=unstimulated platelets.

7.4 DISCUSSION

The results of this chapter, while of a preliminary nature, indicate that it is unlikely that adrenaline influences the total amounts of PI or PA. Some aspects of the data are conflicting, indicating that some aspects of the method could be improved. Separation of PA and PI on Whatman SG-81 paper, using solvent system B, was adequate, provided excessive amounts were not applied, but was subject to variations in behaviour between different batches and small changes in climatic conditions. Platelet preparations were obtained from platelet-rich plasma supplied by the Red Cross Blood Bank. Variability in donors and in handling may have been responsible for the widely different unstimulated levels of arachidonic acid incorporation in PA. Other stimuli associated with handling procedures could account for high incorporation of arachidonic acid observed in PA in apparently unstimulated platelets.

ADP was expected to induce increased PA turnover, without altering the amount of PA (Lloyd <u>et al</u>.,1972), whereas thrombin was expected to increase the amount of PA. In the experiment where there was increased incorporation of arachidonic acid due to exposure to ADP for 30sec, there was also an increase in the amount of PA. On the other hand, where a large incorporation was observed in unstimulated platelets, ADP reduced this incorporation, possibly due to increased turnover. Further experiments are obviously needed to compare a wider range of platelet preparations.

Equally confusing results were obtained in experiments where adrenaline and ADP were added to platelets. It should be noted, however, that the two experiments which did show a rise in arachidonic acid incorporation were performed with platelets in which there was a low initial

incorporation. In the third experiment, where ADP reduced incorporation, adrenaline and ADP in combination reduced this incorporation further. All experiments therefore, show that adrenaline affects phospholipid metabolism, even if only to a small extent. Further replication of all experiments is necessary.

Phospholipase C may be activated in at least two different ways. Thrombin and other agonists interact with their receptors to directly activate phospholipase C which degrades PI (Lapetina <u>et al</u>., 1978) while the ionophore A23187 stimulates phospholipase C via thromboxane and ADP (Rittenhouse, 1984). A23187 mobilizes calcium which activates phospholipase A_2 to release arachidonic acid for thromboxane formation. If adrenaline could be shown to affect calcium mobilization, this may lead to activation of phospholipase C following secretion. When adrenaline potentiates another agonist, the effects of adrenaline on calcium may be sufficient to act synergistically with whatever pathways are triggered by the second agonist.

Increased labelling of PI, which seems to occur in the presence of BPB may indicate an increase in activity of the PI cycle (Lapetina <u>et al</u>., 1981b). Previous work, reported in Chapters 5 and 6, has indicated that BPB may inhibit diglyceride kinase, providing conditions suitable for increased lipase activity if diaclyglycerol production is increased.

The PI spot assayed by this technique may provide

insufficient evidence to determine the effects of adrenaline. It has been shown that both thrombin (Billah and Lapetina, 1982) and ADP (Vickers <u>et al</u>., 1982) cause rapid changes in amounts of phosphatidylinositol-4,5bisphosphate which may chromatograph differently from PI, while the different pools of inositides could participate in reactions necessary for platelet activation in different ways. ADP did not cause any increase in the amount of PA associated with the decrease in PI. It was suggested by these authors that decreases in phosphatidylinositol-4,5bisphosphate result in the mobilization of membrane-bound calcium. An alternative method which could differentiate between the different inositide classes in stimulated platelets might be more revealing in analyses of the effects of adrenaline on platelets.

Recently, adrenaline alone has been shown to not cause breakdown of PI in aspirin-treated platelets (Clare and Scrutton, 1984). The results presented here are inconclusive. The difficulty in observing measurable changes in phospholipids due to adrenaline stimulation may indeed arise because the potentiating effects of adrenaline are not attributable to breakdown of PI. The ability of adrenaline to stimulate calcium redistribution has not been excluded by recent work and adrenaline may influence calcium mobilization independently of the PI-effect.

CHAPTER 8

GENERAL DISCUSSION

heen proposed that plarenets participate in atherescharter, plaren forposed that plarenets participate in atherescharter, plarenet forestion scording to the Response colleges, hypothesis (Abss and Filessit, 1976). Increases in the state of estres of response of the platelet can be brought shout by the cetecholesines, sizebeline and norm respond tors teshily to other symplets. Adrenaline levels time fring wan side stress (Dissuident and Ness, 1980) and this is part prompted the stoop of a strenaline interaction with platelets, since stress (Dissuident a major risk faiter for estressity attary discuss Carenaline interaction with platelets, since stress farmer canding faite attack faiter for estressity attary discuss canding interaction attacks stored in other risk farmer

8.1 DISCUSSION

Platelets make a vital contribution to the normal haemostatic process. There is also a great deal of evidence that increased platelet reactivity is important in the development of atherosclerosis, thrombosis and coronary artery disease. Vascular intimal thickenings have been shown to contain thrombus material (Mustard and Packham, 1975). Platelets adhere to exposed connective tissue following injury to the endothelium and subsequently release granule contents (Stemerman and Ross, 1972; Baumgartner et al., 1967; White, 1972b). Platelet derived growth factor stored in a-granules has been found to be an absolute requirement for the proliferation of smooth muscle cells and to promote migration of endothelial cells in vitro (Ross et al., 1974; Wall et al., 1978). It has thus been proposed that platelets participate in atherosclerotic plaque formation according to the Response-to-Injury Hypothesis (Ross and Glomset, 1976).

Increases in the rate or degree of response of the platelet can be brought about by the catecholamines, adrenaline and noradrenaline. Catecholamines may prepare platelets to respond more readily to other agonists. Adrenaline levels rise during even mild stress (Dimsdale and Moss, 1980) and this in part prompted the study of adrenaline interaction with platelets, since stress may be a major risk factor for coronary artery disease. Catecholamines are also increased in other risk factor conditions including cigarette smoking and hypertension. Adrenaline may increase platelet reactivity, and thus contribute to atherogenesis and its complications. The mechanisms by which adrenaline increases the responsiveness of platelets has been the subject of this thesis. It is known that adrenaline interacts with its own specific cellular receptor on the platelet and this appears to be an a_2 -adrenoreceptor(Grant and Scrutton,1979). The immediate post-receptor events in the platelet have not, so far, been determined.

The differing choices of anticoagulant necessary for in vitro studies of platelet reactivity has lead to difficulties in interpreting the two phases of the platelet response to adrenaline. The response in heparin, and only to a small extent in citrate, has previously been reported to be of the 'primary' type (O'Brien et al., 1969). Primary aggregation is prostaglandin-independent, while the major response in citrate is prostaglandin-dependent (Best et al., 1980). It has been suggested that this 'secondary' aggregation is an artifact of citrated plasma (Mustard et al., 1975) but the findings reported here in Chapter 3 indicate that secondary aggregation in response to ADP can occur if the heparin concentration is lowered sufficiently. The calcium chelating effects of an anticoagulant such as citrate are therefore not a prerequisite for prostaglandindependent or second phase aggregation.

Another criticism of the use of citrate as anticoagulant arises from the belief that it is the reduction in available calcium which reduces the primary

phase of adrenaline-induced aggregation. This may be so, but the presence of citrate does not, however, prevent adrenaline potentiating the response to other agonists (Grant and Scrutton, 1980a). Furthermore, this ability to potentiate is not inhibitable by aspirin (Rao et al., 1980a) and is therefore achieved by a prostaglandin-independent mechanism, which is expected to involve calcium redistribution without decreases in intra-platelet c'AMP (Rao et al., 1980b; 1981a). Citrated plasma was used in their experiments and in the experiments reported here in Chapters 4 to 7. Owen and Le Breton (1981) used a chlortetracycline probe to observe a release of intraplatelet membrane-bound calcium after adrenaline stimulation which was partially inhibited by indomethacin. Further evidence that calcium redistribution is involved in the thromboxane-independent mechanism has been obtained by Brass and Shattil(1982). These authors found that both ADP and adrenaline increased the number of surface membrane affinity sites for calcium in aspirin treated platelets. These agonists also increased the rate of calcium exchange into the intracellular pool without increasing the net amount of calcium.

Several authors have studied adrenaline potentiation of other agonists since it was first demonstrated by Ardlie <u>et al.(1966)</u>. Chapter 3 reports on the extension of these <u>in vitro</u> studies to three or four agonists in combination, instead of the paired combinations employed previously. Circulating blood obviously contains several possible platelet agonists albeit at low concentrations. For the multiple agonist experiments, heparin was chosen as the anticoagulant to more closely reflect the <u>in vivo</u> situation. The study demonstrated that aggregation of platelets was possible <u>in vitro</u> using concentrations of catecholamines which could be achieved <u>in vivo</u>. The previously published lowest levels achieving aggregation <u>in</u> <u>vitro</u> were still higher than in pathological states (Grant and Scrutton, 1980a). The reduced sensitivity to multiple agonists in citrate compared to heparin suggests that calcium may be involved in the potentiation mechanism.

The mechanism of adrenaline potentiation was explored in Chapter 4. The findings are in agreement with those of Rao <u>et al</u>.(1980a) to the extent that adrenaline stimulates both prostaglandin dependent and independent pathways when potentiating ADP. The results differed in being able to show by radioimmunoassay, small increases in cyclooxygenase activity due to adrenaline despite the presence of the cyclooxygenase inhibitor, aspirin, but only when high concentrations of arachidonate were used. In the presence of thromboxane inhibitors, higher concentrations of adrenaline are required to achieve the same extent of aggregation as in the absence of these inhibitors. Therefore, the thromboxane-dependent pathway represents the more efficient mechanism of adrenaline potentiation.

It is suspected that both the prostaglandin-dependent mechanism and prostaglandin-independent mechanism of adrenaline potentiation occur via stimulation of the α_2 -

receptor since the activity of adrenaline can be mimicked by clonidine, a partial α_2 -receptor agonist. The experiments using phenoxybenzamine and dibenamine are difficult to interpret, but do not preclude post-receptor events involving calcium mobilization.

Arachidonic acid release by adrenaline may involve activation of phospholipase A2. The effects of putative phospholipase A2 inhibitors were therefore examined and the findings are reported in Chapter 5. The surprising result with the use of BPB or TPCK was that, rather than enabling a prostaglandin-independent pathway to be examined, these inhibitors revealed that an alternative source of arachidonic acid could be released for thomboxane production. Despite the awareness of the lack of specificity of phospholipase A2 inhibitors, which has been well documented in the case of BPB, it is nevertheless suggested that adrenaline stimulates the release of arachidonic acid via diglyceride lipase. This pathway is the only known source not involving phospholipase A2 enzymes (Bell et al., 1979). BPB has been shown to inhibit diglyceride lipase in platelet extracts but not in intact platelets (Hofmann et al., 1982). The potentiation of thromboxane production in the presence of BPB or TPCK is in agreement with the work of Vargaftig (1976).

Thromboxane-independent potentiation of ADP-induced aggregation by adrenaline was shown to be possible in the presence of phenothiazines, inhibitors of several enzymes, including calmodulin-dependent enzymes. Because it has been shown that ADP increases PA turnover(Lloyd <u>et al</u>., 1972) it is conceivable that adrenaline increases kinase activity and thus PA formation, in the presence of phenothiazines, rather than stimulating arachidonic acid release. The findings of Chapter 5 are therefore consistent with those of Chapter 4 in that adrenaline may influence PI metabolism. This remains to be confirmed by biochemical analysis. It is also possible that adrenaline in the presence of other agonists may promote calcium mobilization independently of PI metabolism.

Since there was some doubt about the possibility of diglyceride lipase being active in the presence of BPB (Hofmann <u>et al</u>., 1982) the effects of exogenous phospholipase C in the presence of BPB were examined. Aggregation, release of nucleotide and phospholipid metabolism due to exogenous phospholipase C are the subject of Chapter 6. This Chapter also includes an investigation of possible inhibitors of phospholipase C and the effects of these on thrombin-induced aggregation and phospholipid metabolism.

Exogenous phospholipase C caused the formation of a small amount of TXA₂ associated with aggregation. In the

presence of BPB, TXA2 formation increased while aggregation was partially inhibited. This parallels the increase in TXA2 observed when adrenaline potentiates ADP. In the presence of BPB, release of nucleotide due to exogenous phospholipase C was largely inhibited but a small indomethacin-sensitive peak of release still occurred. Phospholipid analysis indicated increased external PC breakdown without a corresponding conversion to PA in the presence of BPB. In addition, phospholipase C-induced aggregation in the presence of BPB was inhibited by indomethacin. These data are consistent with diglyceride lipase being more active in the presence of BPB. This would increase arachidonic acid release from diacylglcerol and thus increase TXA₂ formation. It could also divert diacylglycerol away from PA production, providing a mechanism for the inhibition of threshold phospholipase Cinduced aggregation by BPB. Interference at this level could alter phosphorylation reactions essential for release and aggregation.

Similar experiments were performed with thrombin replacing exogenous phospholipase C. Thrombin activates endogenous phospholipase C, forming diacylglycerol from phosphatidylinositol-4,5-bisphosphate (Broekman, 1984) Since it was found that BPB did not affect PI breakdown due to thrombin, and TXA₂ was not estimated, these experiments neither confirmed nor denied that the lipase could be more active in the presence of BPB.

Inhibitors of phospholipase C-induced aggregation may have different mechanisms of action. For example, spermine prevented phosphatidic acid production but trimetoquinol did not. Perhaps trimetoquinol inhibits a step in the activation pathway beyond diacylglycerol and PA formation.

The above evidence, based on aggregation studies with associated TXA₂ formation and nucleotide release required support from biochemical analyses of membrane phospholipids which were subsequently undertaken. The important question of adrenaline effects on phospholipid metabolism, and in particular, PI turnover, in the presence or absence of certain inhibitors could only be answered by quantitation of the relevant phospholipids. Chapter 7 reports on the development of a paper chromatographic method for the separation of phospholipids. This method has the particular advantage of separating PA with good resolution for assay by phosphorus content or radioactive label. PI can also be quantitated but different polyphosphoinositides could not be distinguished.

Adrenaline has been shown to increase ³²Pi incorporation in PI and PA of the rabbit iris muscle (Abdel-Latif, 1974), while ADP causes a decrease in phosphatidylinositol-4,5-bisphosphate associated also with increased turnover of PA(Vickers <u>et al</u>.,1982). The results reported in Chapter 7 are inconclusive but certainly indicate that adrenaline behaves in a different manner to
either ADP or thrombin with respect to phospholipid metabolism. The preliminary results obtained indicated that, when adrenaline potentiates ADP-induced aggregation, increased turnover of PI and PA is difficult to observe. Previous Chapters indicated that BPB may inhibit diacylglycerol kinase but this was not supported, since the amount of PA and the turnover of PA did not alter in the presence of BPB. In the presence of BPB, however, adrenaline and ADP seemed to increase incorporation of ³Harachidonic acid in PI, without increasing the amount of PI. Further investigation of PI metabolism with respect to changes among the different inositide classes should be undertaken. In the presence of BPB an exaggeration of the adrenaline effects on PI cycling may occur. Some evidence for the involvment of calcium in the potentiation mechanism has been presented in this thesis, and if, indeed, adrenaline does not alter PI metabolism in the initial stages of platelet interaction, then calcium mobilization could occur independently of the PI-effect.

8.2 CONCLUSIONS

1. With low concentrations of heparin as anticoagulant, adrenaline can cause potentiation of aggregation <u>in vitro</u> at concentrations which can be found in the circulation, supporting the theory that adrenaline plays a role in facilitating platelet activation at sites of vascular injury.

194

2. The calcium chelating effects of citrate are not a prerequisite for prostaglandin-dependent aggregation due to adrenaline. Citrate indeed may reduce the potentiating effects of both vasopressin and adrenaline, indicating the importance of calcium availability.

3 There must be at least two mechanisms by which adrenaline can potentiate platelet aggregation, firstly by causing increases in cyclooxygenase activity, and secondly, by a cyclooxygenase-independent mechanism, probably both involving a₂-receptor interaction. The cylcooxygenasedependent pathway is the more efficient mechanism of adrenaline potentiation.

4. In the presence of certain phospholipase A_2 inhibitors, namely BPB and TPCK, adrenaline increases TXB_2 formation due to ADP, possibly due to increased release of arachidonic acid. This increased release, via a nonphospholipase A_2 pathway, may be due to stimulation of diacylglycerol lipase by adrenaline.

5. The immediate post-receptor events in the adrenaline response, may involve changes in phospholipids caused by other agonists and in particular, changes in the PI cycle. Adrenaline may influence diacylglycerol production or metabolism by ADP and thus influence a calcium-independent pathway via protein kinase C.

6. Further examination of the effects of dibenamine on adrenaline activation may reveal any possible influence of adrenaline on membrane calcium distribution independent of the PI-cycle. 7. The response to exogenous phospholipase C does not mimic the response to endogenous PI-specific phospholipase C. This may be because outer leaflet PC, the substrate for exogenous phospholipase C, does not participate in the cycling series of reactions. Nevertheless, the rapid formation of diacylglycerol by exogenous phospholipase C could be exploited for examining the importance of this compound.

Since phospholipase C induced aggregation in the 8. presence of BPB occurs with an increased production of TXB₂, it is tempting to believe that diacylglycerol lipase is active while other routes for diacylglycerol metabolism are inhibited by BPB. This could not however, be confirmed with thrombin-induced aggregation. The indication that adrenaline promotes 9. diacylglycerol formation, but not PA formation should be examined further with respect to polyphosphoinositides. Changes in phospholipids due to adrenaline as measured by paper chromatography are difficult to observe. The evidence from other workers that adrenaline by 10. itself does not influence PI turnover, nor calcium uptake, in the initial interaction with platelets, should not deter investigation of these possibilities in relation to the mechanism of adrenaline potentiation of other agonists.

196

REFERENCES

Derivation from platelite. Troc. Mat. Acad. Sei isolation of a cationic polypeptide from human serva

- Abdel-Latif,A.A. (1974) Effects of neurotransmitters and other pharmacological agents on ³²Pi incorporation into phospholipids of the iris muscle of the rabbit. Life Sci. 15:961-973.
- Addison, W. (1842) On the colourless corpuscles and on the molecules and cytoblasts in the blood. London Med. Gaz. 30:144.
- Antoniades, H.N. and Scher, C.D. (1977) Radioimmunoassay of a human serum growth factor for Ball/c-3T3 cells : Derivation from platelets. Proc. Nat. Acad. Sci. 74:1973-1977.
- Antoniades, H.N., Stathakos, D. and Scher, C.D. (1975) Isolation of a cationic polypeptide from human serum that stimulates proliferation of 3T3 cells. Proc. Nat. Acad. Sci. <u>72</u>:2635-2639.
- Ardlie,N.G., Glew,G. and Schwartz,C.J. (1966) Influence of catecholamines on nucleotide-induced platelet aggregation. Nature 212:415-417.
- Ardlie, N.G. and Han, P. (1974) Enzymatic basis for platelet aggregation and release: the significance of the plasmatic atmosphere and the relationship between platelet function and blood coagulation. Brit. J. Haematol. 26:331-356.
- Ardlie, N.G., Packham, M.A. and Mustard, J.F. (1970) Adenosine diphosphate-induced platelet aggregation in suspensions of washed platelets. Brit. J. Haematol. 19:7-17.
- Ardlie,N.G., Perry,D.W., Packham,M.A. and Mustard,J.F. (1971) Influence of apyrase on stability of suspensions of washed rabbit platelets. Proc. Soc. Exp. Biol. Med. 136:1021-1023.
- Barber,A.J., and Jamieson,G.A. (1970) Isolation and characterization of plasma membranes from human blood platelets. J. Biol. Chem. <u>245</u>:6357-6365.
- Barnathan, E.S., Adonizio, V.P. and Shattil, S.J. (1982) Interaction of verapamil with human platelet alphaadrenergic receptors. Am. J. Physiol. 242:H19-23.
- Baumgartner, H.R. (1973) The role of blood flow in platelet adhesion, fibrin deposition and formation of mural thrombi. Microvasc. Res. <u>5</u>:167-179.

Baumgartner, H.R. (1977) Platelet interaction with collagen fibrils in flowing blood. Thromb. Haem. <u>37</u>:1-28. Baumgartner, H.R. and Born, G.V.R. (1968) Effects of 5hydroxytryptamine on platelet aggregation. Nature 218:137-141. Baumgartner, H.R. and Spaet, T.H. (1970) Endothelial replacement in rabbit arteries. Fed. Proc. 29:710abs.

- Baumgartner, H.R. Tranzer, J.P. and Studer, A. (1967) An electron microscope study of platelet thrombus formation in the rabbit with particular regard to 5hydroxytryptamine release. Throm. Diath. Haemorrh. 18:592-604.
- Behnke, O. (1965) Further studies on microtubules. A marginal bundle in human and rat thymocytes. J. Ultrastruc. Res. 13:469-477.
- Behnke, O. (1970) microtubules in disk-shaped blood cells. Int. Rev. Exp. Pathol. <u>9</u>:1-92.
- Bell,R.L., Kennerly,D.A., Stanford,N. and Majerus,P.W. (1979) Diglyceride lipase: A pathway for arachidonate release from himan platelets. Proc. Nat. Acad. Sci. 76:3238-3241.
- Benditt, E.P. (1976) Implications of the monoclonal character of human atherosclerotic plaques. Bietr. Pathol. 158:433-444.
- Benditt, E.P. and Benditt, J.M. (1973) Evidence for a monoclonal origin of human atherosclerotic plaques. Proc. Nat. Acad. Sci. 70:1753-1756.
- Berndt, M.C. and Phillips, D.R. (1981a) Platelet membrane proteins: composition and receptor function. In : Platelets in Biology and Pathology Vol.2 ed. J.L. Gordon Elsevier/North-Holland Biomedical Press.
- Berndt, M.C. and Phillips, D.R. (1981b) Interaction of thrombin with platelets: purification of the thrombin substrate. Ann. N.Y. Acad. Sci. <u>370</u>:87-95.
- Berndt, M.C. and Phillips, D.R. (1981c) Purification and preliminary physicochemical characterization of human platelet membrane glycoprotein V. J. Biol. Chem. 256:59-65.
- Best,L.C., Holland,T.K., Jones,P.B.B. and Russell,R.G.G. (1980) The interrelationship between thromboxane biosnthesis, aggregation and 5-hydroxytryptamine secretion in human platelets in vitro. Thromb. Haem. 43:38-40.
- Berridge, M.J. (1984) Inositol trisphosphate and diacylglycerol as second messengers. Review article. Biochem. J. 220:345-360.

- Bevers, E.M., Comfurius, P. and Zwaal, R.F.A. (1983) Changes in membrane phospholipid distribution during platelet activation. Biochim. Biophys. Acta. <u>736</u>:57-66.
- Billah, M.M. and Lapetina, E.G. (1982a) Rapid decrease of phosphatidylinositol 4,5-bisphosphate in thrombinstimulated platelets. J. Biol. Chem. <u>257</u>:12705-12708.
- Billah, M.M. and Lapetina, E.G. (1982b) Degradation of phosphatidylinositol-4,5-bisphosphate is insensitive to Ca²⁺ mobilization in stimulated platelets. Biochem. Biophys. Res. Commun. <u>109</u>:217-222.
- Billah, M.M. and Lapetina, E.G. (1983) Platelet-activating factor stimulates metabolism of phosphoinositides in horse platelets : possible relationship to Ca²⁺ mobilization during stimulation. Proc. Nat. Acad. Sci. 80:965-968.
- Bills, T.K., Smith, J.B. and Silver, M.J. (1977) Selective release of arachidonic acid from the phospholipids of human platelets in response to thrombin. J. Clin. Invest. 60:1-6.
- Bizzozero, G. (1882) Ueber einer Formbestandtheil der Blutes und desser Role bei der Thrombose und der Blutgerinning. Virch. Archiv. <u>90</u>:261-332.
- Bligh, E.G. and Dyer, W.J. (1959) A rapid method of total lipid extraction and purification. Can. J. Biochem. Physiol. <u>37</u>:911-917.
- Born,G.V.R. (1962) Aggregation of blood platelets by adenosine diphosphate and its reversal. Nature 194:927.
- Born,G.V.R. (1970) Observations on the change in shape of blood platelets brought about by adenosine diphosphate. J.Physiol. 209:487-511.
- Born, G.V.R., Hornikiewick, O. and Stafford, A. (1958) The uptake of adrenaline and noradrenaline by blood platelets of the pig. Brit. J. Pharmacol. <u>13</u>:411-418.
- Brass,L.F. and Shattil,S.J. (1982) Changes in surfacebound and exchangeable calcium during platelet activation. J. Biol. Chem. <u>257</u>:14000-14005.
- Broekman, M.J. (1983) The time course of decreases in endogenous phosphatidyl bisphosphate (TPI) of washed human platelets suggests that TPI represents the site of plasma membrane bound calcium monitored by fluorescence of chlortetracycline. Blood <u>62</u>:251a.

- Broekman, M.J. (1984) Phosphatidylinositol 4,5-bisphosphate may represent the site of release of plasma membranebound calcium upon stimulation of human platelets. Biochem. Biophys. Res. Commun. 120:226-231.
- Broekman, M.J., Ward, J.W. and Marcus, A.J. (1980) Phospholipid metabolism in stimulated human platelets. Changes in phosphatidylinositol, phosphatidic acid and lysophospholipids. J. Clin. Invest. <u>66</u>:275-283.
- Busch, C., Wasteson, A. and Westermark, B. (1976) Release of a cell growth promoting factor from human platelets. Thromb. Res. 8:493-500.
- Carvalho, A.C.A., Colman, R.W. and Lees, R.S. (1973) Platelet function in hyperlipoproteinemia. New Engl. J. Med. 290:434-438.
- Cazenave, J-P., Benveniste, J. and Mustard, J.F. (1979) Aggregation of rabbit platelets by platelet-activating factor is independent of the release reaction and the arachidonate pathway and inhibited by membrane-active drugs. Lab. Invest. <u>41</u>:275-285.
- Chap H., Lloveras J. and Douste-Blazy (1971) Action agregeante de la phospholipase C de Clostridium welchii sur les plaquettes humaines. C.R. Acad. Sci.(Paris) Ser. <u>D273</u>:1452-1455.
- Chap,H., Perret,B., Mauco,G., Simon,M.F. and Douste-Blazy,L. (1979) Organization and role of platelet membrane phospholipids as studied with purified phospholipases. Agents Actions <u>9</u>:400-406.
- Chignard, M., Le Couedic, J.P., Vargaftig, B.B. and Benveniste, J. (1979) The role of platelet activating factor in platelet aggregation. Nature <u>179</u>:799-800.
- Chignard, M., Le Couedic, J.P., Vargaftig, B.B. and Benveniste, J. (1980) Platelet-activating factor(PAFacether) secretion from platelets: effects of aggregating agents. Brit. J. Haematol. <u>46</u>:455-464.
- Clare K.A. and Scrutton M.C. (1984) The role of Ca²⁺ uptake in the response of human platelets to adrenaline and to 1-O-alkyl-SN-glycero-3phosphocholine(platelet activating factor). Eur. J. Biochem. 140:129-136.
- Clare K.A., Scrutton M.C. and Thompson N.T. (1984) Effects of a₂-adrenoceptor agonists and of related compounds on aggregation of and on adenylate cyclase activity in, human platelets. Br. J. Pharmacol. <u>82</u>:467-476.

- Clark, E., Graef, T. and Chasis, H. (1936) Thrombosis of the aorta and coronary arteries with special reference to fibrinoid lesions. A.M.A. Arch. Path. <u>22</u>:183-212.
- Cockcroft, S., Jones, L.M., Allan, D. and Michell, R.H. (1980) Effects of alkylating antagonists on the stimulated turnover of phosphatidylinositol produced by a variety of calcium-mobilizing receptor systems. Cell Calcium 1:49-68.
- Cocks T.M., Dilger P. and Jenkinson D.H.(1981) The mechanism of the blockade by trifluoperazine of some actions of phenylephrine on liver and smooth muscle. Biochem. Pharmacol. <u>80</u>:2873-2875.
- Connolly T.M. and Limbird L.E. (1983) The influence of Na⁺ on the alpha₂-adrenergic receptor system of human platelets. A method of removal of extraplatelet Na⁺. Effect of Na⁺ removal on aggregation, secretion and c'AMP accumulation. J. Biol. Chem. <u>258</u>:3907-3912.
- Crawford, N. (1976) Platelet microfilaments and microtubules. In: Platelets in Biology and Pathology. ed.J.L.Gordon North Holland pp121-157.
- De Clerck, F.F. and Herman, A.G. (1983) 5-Hydroxytryptamine and platelet aggregation. Fed. Proc. <u>42</u>:228-232.
- Derksen,A. and Cohen,P. (1973) Extensive incorporation of [2-14C]mevalonic acid into cholesterol precursors by human platelets in vitro. J. Biol. Chem. 248:7396-7403.
- Deykin D. and Snyder D. (1973) Effect of epinephrine on platelet lipid metabolism. J. Lab. Clin. Med. <u>82</u>: 554-559.
- Dimsdale, J.E. and Moss, J. (1980) Plasma catecholamines in stress and exercise. J.A.M.A. 243:340-342.

Donne, A. (1842) Compte. Rend. Acad. Sci. 14:366.

- Drummond, A.H. (1976) Interactions of blood platelets with biogenic amines: stimulation and receptor binding. In: Platelets in Biology and Pathology. ed.J.L.Gordon Norht Holland pp203-239.
- Duguid, J.B. (1948) Thrombosis as a factor in the pathogenesis of aortic atherosclerosis. J. Path. Bact. 60:57-61.
- Duguid, J.B. (1949) Pathogenesis of atherosclerosis. Lancet 2:929-927.
- Eberth, J.C. and Schimmelbusch, C. (1885) Fortsch. d. Med. 3:379.

- Erman, A., Schwartzman, M and Raz, A. (1980) Indomethacin but not aspirin inhibits basal and stimulated lipolysis in rabbit kidney. Prostaglandins. 20:689-702.
- Fain, J.N. and Garcia-Sainz, J.A. (1980) Role of phosphatidylinositol turnover in alpha₂ effects of catecholamines. Life Sci. <u>26</u>:1183-1194.
- Feagler,J.R., Tillack,T.W., Chaplin,D.D. and Majerus,P.W. (1974) The effects of thrombin on phytohaemagglutinin receptor sites in human platelets. J. Cell. Biol. 60:541-553.
- Feinstein, M.B. (1980) Release of intracellular membranebound calcium precedes the onset of stimulus-induced exocytosis in platelets. Biochem. Biophys. Res. Commun. 93:593-600.
- Feinstein M.B., Egan J.J., Sha'afi R.I. and White J. (1983) The cytoplasmic concentration of free calcium in platelets is controlled by stimulators of cyclic AMP production (PGD₂, PGE₁, forskolin). Biochem. Biophys. Res. Commun. <u>113</u>:598-604.
- Feinstein M.B., Rodan G.A. and Cutler L.S. (1981) Cyclic AMP and calcium in platelet function. In: Platelets in Biology and Pathology Vol.2. ed. J.L. Gordon Elsevier/North-Holland Biomedical Press.
- Fisher-Dzoga, K., Chen, R. and Wissler, R.W. (1974) Effects of serum lipoproteins on the morphology and metabolism of arterial smooth muscle cells. Adv. Exp. Med. Biol. 43:299.
- Flower, R.J. and Cardinal, D.E. (1979) Use of a novel platelet aggregometer to study the generation by, and the actions of, prostacyclin in whole blood. In: Prostacyclin. eds.J.R.Vane and S.Bergstrom Raven Press N.Y. pp211-219.
- Fratkin,J.D., Cancilla,P.A. and De Bault,L.E. (1980)
 Platelet factor and cerebral vascular endothelium:
 platelet-induced mitogenesis. Thromb. Res. 19:473483.
- Freidman,L.I., Leim,H. Grabowski,E.F., Leonard,E.F. and McCord,C.W. (1970) Inconsequentiality of surface properties for initial platelet adhesion. Trans. Amer. Soc. Artif. Int. Organs 16:63-73.
- Ganguly P. (1977) Binding of thrombin to functionally defective platelets : a hypothesis on the nature of the thrombin receptor. Br. J. Haematol. 37:47-51.
- Ganguly P. and Gould N.L. (1979) Thrombin receptors of human platelets : thrombin binding and anti-thrombin

properties of glycoprotein I. Br. J. Haematol. 42:137-145.

- Geer, J.C. and Haust, M.D. (1972) Smooth muscle cells in atherosclerosis. In: Monographs on Atherosclerosis 2. eds. O.J.Pollack, H.S.Simms and J.E.Kirk Karger N.Y.
- Gerrard, J.M., Butler, A.M., Graff, G., Stoddard, S.F. and White, J.G. (1978b) Prostaglandin endoperoxides promote calcium release from a platelet membrane fraction in vitro. Prost. Med. <u>1</u>:373-385.
- Gerrard, J.M., Butler, A.M., Peterson, D.A. and White, J.G. (1978c) Phosphatidic acid releases calcium from membrane fraction in vitro. Prost. Med. <u>1</u>:387-396.
- Gerrard,J.M., Kindom,S.E., Peterson,D.A., Peller,J., Krantz,K. and White,J.G. (1979) Lysophosphatidic acids. Influence on platelet aggregation and intracellular calcium flux. Am. J. Pathol. <u>96</u>:423-438.
- Gerrard, J.M. and White, J.G. (1978) Prostaglandins and thromboxanes: "Middlemen" modulating platelet function in haemostasis and thrombosis. In: Prog. Hemost. Thromb. ed.T.H.Spaet Grune and Stratton N.Y.
- Gerrard, J.M., White, J.G. and Peterson, D.A. (1978a) The platelet dense tubular system: its relationship to prostaglandin synthesis and calcium flux. Thromb. Haem. 40:224-231.
- Gerrard, J.M., White, J.G., Rao, G.H.R. and Townsend, D. (1976) Localization of platelet prostaglandin production in the platelet dense tubular system. Am. J. Pathol. 83:283-298.
- Gimbone, M.A., Aster, R.H., Cotran, R.S., Corkery, J., Jandle, J.H. and Folkman, J. (1969) Preservation of vascular integrity in organs perfused in vitro with a platelet-rich medium. Nature 222:33-36.
- Gorman, R.R., Bunting, S. and Miller, O.V. (1977a) Modulation of human adenylate cyclase by prostacyclin (PGX). Prostaglandins 13:377-388.
- Gorman, R.R., Fitzpatrick, F.A. and Miller, O.V. (1977b) A selective thromboxane synthetase inhibitor blocks the c'AMP lowering activity of PGH₂. Biochem. Biophys. Res. Commun. <u>79</u>:305-313.
- Gorman R.R., Fitzpatrick F.A. and Miller O.V. (1978) Reciprocal regulation of human platelet c'AMP levels by thromboxane A₂ and prostacyclin. Adv. Cyclic Nucleotide Res. <u>9</u>:597-609.

- Grant J.A. and Scrutton M.C. (1979) Novel a2adrenoreceptors primarily responsible for inducing human platelet aggregation. Nature 277:659-661.
- Grant J.A. and Scrutton M.C. (1980a) Positive interaction between agonists in the aggregation response of human blood platelets: interaction between ADP, adrenaline and vasopressin. Brit. J. Haematol. <u>44</u>:109-125.
- Grant J.A. and Scrutton M.C. (1980b) Interaction of selective alpha-adrenoceptor agonists and antagonists with human and rabbit blood platelets. Brit. J. Pharmacol. 71:121-134.
- Greenberg, J.P., Packham, M.A., Cazenave, J.P., Reimers, H.J. and Mustard, J.F. (1975) Effects on platelet function of removal of platelet sialic acid by neuraminidase. Lab. Invest. 32:476-484.
- Gulliver, G. (1881) Gulliveriana. An Autobiography. Canterbury.
- Hallam, T.J., Rink T.J. and Sanchez A. (1983a) Effects of arachidonate and a thromboxane A₂ analogue on human platelets studied with a fluorescence indicator from cytoplasmic free calcium. J. Physiol. <u>343</u>:97p-98p.
- Hallam,T.J., Sanchez,A. and Rink,T.J. (1983b) ADP increases cytoplasmic free Ca²⁺ in quin 2-loaded human platelets, mainly by influx across the plasma membrane. Thromb. Haem. <u>50</u>:76 abstr.
- Hallam, T.J., Sanchez, A. and Rink, T.J. (1984a) Stimulusresponse coupling in human platelets. Biochem. J. 218:819-827.
- Hallam,T.J., Thompson,N.T., Scrutton,M.C. and Rink,T.J. (1984b) Role of cytoplasmic free calcium in the response of quin 2-loaded platelets to vasopresin. Biochem. J. <u>221</u>:897-901.
- Hamberg, M., Svensson, J., Wakabayashi, T. and Samuelsson, B. (1974) Isolation and structure of two prostaglandin endoperoxides that cause platelet aggregation. Proc. Nat. Acad. Sci. <u>71</u>:345-349.
- Hansen,S.O. and Theodorson,L. (1971) The usefulness of an improved calcium electrode in the measurement of ionized calcium in serum. Clin. chim. Acta <u>31</u>:119-122.
- Hanson, J.P., Repke, D.I., Katz, A.M. and Aledort, L.M. (1973) Calcium ion control of platelet thrombosthenin ATPase activity. Biochim. Biophys. Acta <u>314</u>:382-389.

Harker, L.A. and Finch, C.A. (1969) Thrombokinetics in man. J. Clin. Invest. 48:963-974.

- Harwood, J.P., Moskowitz, J. and Krishna, G. (1972) Dynamic interaction of prostaglandin and norepinephrine in the formation of adenosine 3',5' monophosphate in human and rabbit platlets. Biochim. Biophys. Acta 261:444-456.
- Haslam, R.J., Davidson, M.M.L., Davies, T., Lynham, J.A. and McClinaghan, M.D. (1978a) Regulation of blood platelet function by cylcic nucleotides. Adv. Cyclic Nucl. Res. 9:533-552.
- Haslam R.J., Davidson M.M.L. and Desjardins J.V.(1978b) Inhibition of adenylate cyclase by adenosine analogues in preparations of broken and intact human platelets. Evidence for the unidirectional control of platelet function by cyclic AMP. Biochem. J. <u>176</u>:83-95.
- Haslam, R.J. and Rosson, G.M. (1972) Aggregation of human blood platelets by vasopressin. Am. J. Physiol. 223:958-967.
- Haslam R.J. and Taylor A. (1971) Role of cyclic 3',5' adenosine monophosphate in platelet aggregation. In: Platelet Aggregation. Masson and Cie., Paris. pp 85-93.
- Hassall D.G., Forrest L.A., Bruckdorfer K.R., Marenah C.B., Turner P., Cortese C., Miller N.E. and Lewis B. (1983) Influence of plasma lipoproteins on platelet aggregation in a normal male population. Arteriosclerosis 3:332-338.
- Hathaway, D.R. and Adelstein, R.S. (1979) Human platelet myosin light chain kinase requires the calcium binding protein calmodulin for activity. Proc. Nat. Acad. Sci. 76:1653-1657.
- Hawkins, R.J. (1972) Smoking, platelets and thrombosis. Nature 236:450-452.
- Haydon, G.B. and Taylor, D.A. (1965) Microtubules in hamster platelets. J. Cell. Biol. <u>26</u>:673-676.
- Hayem,G. (1878) Recherches sur l'evolution des hematies dans le sang de l'homme et des vertebres. Arch. de Phys. norm et Path. 5:692.
- Heptinstall, S. (1976) The use of a chelating ion-exchange resin to evaluate the effects of the extracellular calcium concentration on adenosine diphosphate induced aggregation of human blood platelets. Thromb. Haem. 36:208-220.

- Hemker, H.C., van Rijn J.L.M.L., Rosing J. van Dieijen G., Bevers E.M. and Zwaal R.F.A. (1983) Platelet membrane involvement in blood coagulation. Blood Cells. 9:303-317.
- Higgs, G.A., Flower, R.J. and Vane, J.R. (1979) A new approach to antiinflammatory drugs. Biochem. Pharmacol. 28:1959-1961.
- Hoffman, A.S. (1975) Report of NHLI Biopolymers Division Contract. N01-HB-2970-1 (Available through NTIS, Springfield, Va.).
- Hofman, S.L., Prescott, S.M. and Majerus, P.W. (1982) The effects of mepacrine and p-bromophenacyl bromide on arachidonic acid release in human platelets. Arch. Biochem. Biophys. 215:237-244.
- Hogerstedt, A. and Nemser, M. (1897) Ueber die krankhafte Verengerung und Verschleressung vom Aortenbogen ausgehender grosser Arterien. Ztschr. klin. Med. 31:130-147.
- Holme,R., Sixma,J.J., Muier,E.H. and Hovig,T. (1973) Demonstration of platelet fibrinogen secretion via the surface connecting system. Thromb. Res. 3:347-356.
- Holmsen, H. (1975) Biochemistry of the platelet release reaction. In: Biochemistry and Pharmacology of Platelets. Ciba Foundation Symposium. 35:175-196.
- Holmsen,H., Dangelmaier,C.A. and Holmsen,H-K. (1981) Thrombin-induced platelet responses differ in requirement for receptor occupancy. J. Biol. Chem. 256:9393-9395.
- Holmsen, H., Day, H.J. and Stormorken, H. (1969) The blood platelet release reaction. Scand. J. Haematol. Suppl. 8:3-26.
- Holmsen, H., Setkowsky, C.A. and Day, H.J. (1974) Effects of antimycin and 2-deoxyglucose on adenine nucleotides in human platelets. Biochem. J. 144:385-396.
- Huang, E.M. and Detwiler, T.C. (1981) Characteristics of the synergistic actions of platelet agonists. Blood 57:685-691.
- Humes, J.L., Winter, C.A., Sadowski, S.J. and Kuehl Jr., F.A. (1981) Multiple sites on prostaglandin cyclooxygenase are determinants in the action of non-steriodal antiinflammatory agents. Proc. Nat. Acad. Sci. 78:2053-2056.
- Humphrey, J.H. and Jacques, R. (1955) Release of histamine and 5-hydroxytryptamine(serotonin) from platelets by antigen-antibody binding reactions(in vitro). J. Physiol. 128:9-27.

- Huzoor-Akbar and Ardlie, N.G. (1976) Evidence that collagen releases human platelet constituents by two different mechanisms. Brit. J. Haematol. 34:137-146.
- Huzoor-Akbar and Ardlie, N.G. (1977) Platelet activation in haemostasis. Role of thrombin and other clotting factors in platelet-collagen interactions. Haemostasis <u>6</u>:59-71.
- Huzoor-Akbar and Ardlie,N.G. (1978) Further evidence for the role of thrombin in the platelet release reaction caused by various agents and the nature of the biphasic platelet aggregation. Brit. J. Haematol. 38:381-390.
- Huzoor-Akbar, Navran, S.S., Miller, D.D. and Feller, D.R. (1982) Antiplatelet actions of trimetoquinol isomers: evidence for inhibition of a prostaglandin-independent pathway of platelet aggregation. Biochem. Pharmacol. 31:886-889.
- Hwang, D.H. and LeBlanc, P. (1981) Heparin inhibits the formation of endoperoxide metabolites in rat platelets: aspirin like activity. Prost. Med. 6:341-344.
- Imai,A., Yano,K., Kameyama,Y. and Nozawa,Y. (1982) Evidence for predominance of phospholipase A2 in release of arachidonic acid in thrombin-activated platelets. Phosphatidylinositol-specific phospholipase C may play a minor role in arachidonate liberation. Jap. J. Exp. Med. 52:99-105.
- Jafferji,S.S. and Michell,R.W. (1976) Effects of calciumantagonistic drugs on the stimulation by carbamoylcholine and histamine of phosphatidylinositol turnover in longitudinal smooth muscle of guinea-pig ileum. Biochem. J. <u>160</u>:163-169.
- Jakobs, K.H., Bohme, E., Mocikat, S. (1974) Proceedings: cyclic GMP formation in human platelets. Naunyn Schmiedebergs Arch. Pharmacol. 282 (Suppl.):R40.
- Jakubowski, J.A. and Ardlie, N.G. (1978) Modification of human platelet function by a diet enriched in saturated or polyunsaturated fat. Atherosclerosis 31:335-344.
- Johnson, P.C., Ware, J.A., Cliveden, P.B., Smith, M., Dvorak, A.M., and Salzman, E.W. (1985) Measurement of ionized calcium in blood platelets with the photoproton aequorin. J. Biol. Chem. 260:2069-2076.
- Johnson, S.A. (1971) In: The Circulating Platelet. ed.S.A. Johnson Academic Press N.Y. pp284.

Johnston, C.I., Newman, M. and Woods, R. (1981) Role of vasopressin in cardiovascular homeostasis and hypertension. Clin. Sci. (Suppl) <u>61</u>:129s-139s.

- Jorgenson, K.A., Dyerberg, J. and Stofferson, E. (1980) A biphasic effect of prostacyclin(PGI₂) on platelet aggregation. Thromb. Res. <u>19</u>:887-881.
- Kaibuchi K., Takai Y., Sawamura M., Hohijima M., Fujikura T. and Nishizuka Y. (1983) Synergistic functions of protein phosphorylation and calcium mobilization in platelet activation. J. Biol. Chem. 258:6701-6704.
- Kannagi, R., Koizumi, K., Hata-Tanoue, S. and Masuda, T. (1980) Mobilization of arachidonic acid from phosphatidylethanolamine fraction to phosphatidylcholine fraction in platelets. Biochem. Biophys. Res. Commun. <u>96</u>:711-718.
- Kannagi.R., Koizumi,K. and Masuda,T. (1981) Limited hydrolysis of platelet membrane phospholipids. J. Biol. Chem. 256:1177-1184.
- Kaser-Glanzmann, R., Jakabova, M., George, J.N. and Luscher, E.F. (1977) Stimulation of calcium uptake in platelet membrane vesicles by adenosine 3',5'-cyclic monophosphate and protein kinase. Biochim. Biophys. Acta 466:429-440.
- Kawahara,Y., Takai,Y., Minkuchi,R., Sano,K. and Nishizuka,Y. (1980) Phospholipid turnover as a possible transmembrane signal for protein phosphorylation during human platelet activation by thrombin. Biochem. Biophys. Res. Commun. <u>97</u>:309-317.
- Kawahara,Y., Yamanishi,J., Furuta,Y., Kaibuchi,K., Takai,Y. and Fukuzaki,H. (1983a) Elevation of cytoplasmic free calcium concentration by stable thromboxane A₂ analogue in human platelets. Biochem. Biophys. Res. Commun. 117:663-669.
- Kawahara,Y., Yamanishi,J., Tsunemitsu,M. and Fukuzaki,H. (1983) Protein phosphorylation and diglyceride production during serotonin release induced by epinephrine plus ADP in human platelets. Thromb. Res. 30:477-485.
- Kitchen, E.A., Boot, J.R. and Dawson, W. (1978) Chemotactic activity of thromboxane B₂, prostaglandins and their metabolites for polymorphonuclear leucocytes. Prostaglandins 16:239-244.

Kloprogge E., De Haas G.H., Gorter G. and Akkerman J.W.N. (1983) Stimulus-response coupling in human platelets. Evidence against a role of PAF-acether in the 'third pathway'. Thromb. Res. <u>30</u>:107-112.

- Kramer, R.M., Jakubowski, J.A., Vaillancourt, R. and Deykin, D. (1982) Effect of membrane cholesterol on phospholipid metabolism in thrombin-stimulated platelets. Enhanced activation of platelet phospholipase(s) for the liberation of arachidonic acid. J. Biol. Chem. 257:6844-6849.
- Kume, S. and Kariya, T. (1980) Interrelationship between calcium and cyclic nucleotides in platelet function. Nippon Ketsueki Gakkai Zasshi <u>43</u>:1103-1112.
- Lages, B. and Weiss, H.J. (1980) Biphasic aggregation responses to ADP and epinephrine in some storage pool deficient platelets: relationship to the role of endogenous ADP in platelet aggregation and secretion. Thromb. Haem. 43:147-153.
- Lages, B. and Weiss, H.J. (1981) Dependence of human platelet functional responses on divalent cations: aggregation and secretion in heparin- and hirudinanticoagulated platelet-rich plasma and the effects of chelating agents. Thromb. Haem. 45:173-179.

Laker, C. (1889) Virch. Archiv. 116:28.

- Lapetina, E.G., Billah, M.M. and Cuatracasas, P. (1981a) The initial action of thrombin on platelets. Conversion of phosphatidylinositol to phosphatidic acid preceding the production of arachidonic acid. J. Biol. Chem. 256:5037-5040.
- Lapetina, E.G., Billah, M.M. and Cuatracasas, P. (1981b) The phosphatidylinositol cycle and the regulation of arachidonic acid production. Nature 292:367-369.
- Lapetina, E.G., Chandrabose, K.A. and Cuatracasas, P. (1978) Ionophore A23187- and thrombin-induced platelet aggregation: independence from cyclooxygenase products. Proc. Nat. Acad. Sci. <u>75</u>:818-822
- Lapetina, E.G. and Cuatracasas, P. (1979) Stimulation of phosphatidic acid production in platelets precedes the formation of arachidonate and parallels the release of serotonin. Biochim. Biophys. Acta <u>573</u>:394-402.
- Le Breton,G.C., Dinerstein,R.J., Roth,L.J. and Feinberg,H. (1976) Direct evidence for intracellular divalent cation distribution associated with platelet shape change. Biochem. Biophys. Res. Commun. <u>71</u>:362-370.
- Lloyd, J.V. and Mustard, J.F. (1974) Changes in ³²P-content of phosphatidic acid and the phosphoinositides of rabbit platelets during aggregation induced by

collagen or thrombin. Brit. J. Haematol. <u>26</u>:243-252.

- Lloyd, J.V., Nishizawa, E.E., Haldar, J. and Mustard, J.F. (1972) Changes in ³²P-labelling of platelet phospholipids in response to ADP. Brit. J. Haematol. 23:571-585.
- Lloyd, J.V., Nishizawa, E.E., Joist, H.J. and Mustard, J.F. (1973b) Effect of ADP-induced aggregation on ³²PO₄ incorporation into phosphatidic acid and the phosphoinositides of rabbit platelets. Brit. J. Haematol. 24:589-604.
- Lloyd, J.V., Nishizawa, E.E. and Mustard, J.F. (1973a) Effect of ADP-induced shape change on incorporation of ³²P into platelet phosphatidic acid and mono-, di- and triphosphatidylinositol. Brit. J. Haematol. <u>25</u>:77-99.
- Lullman,H., Ohnesorge,F.K., Schanwecker,G.C. and Wasserman,O. (1969) Inhibition of the actions of carbachol and DFP on guinea pig isolated atria by alkane-bis-ammonium compounds. Eur. J. Pharmacol. 6:241-247.
- Lullman H., Plosch H. and Ziegler A. (1980) Ca replacement by cationic amphiphilic drugs from lipid monolayers. Biochem. Pharmacol. 29:2969-2974.
- Lyons R.M., Stanford N. and Majerus P.W. (1975) Thrombininduced protein phosphorylation in human platelets. J. Clin. Invest. 56:924-936.
- MacIntyre, D.E. (1976) The platelet release reaction: association with adhesion and aggregation, and comparison with secretory responses in other cells. In: Platelets in Biology and Pathology. ed.J.L.Gordon North Holland pp61-85.
- MacIntyre, D.E., Hardin, R.I., Rosenberg, R. and Salzman, E.W. (1981) Heparin opposes prostanoid and non-prostanoid platelet inhibitors by direct enhancement of aggregation. Thromb. Res. 22:167.
- MacIntyre D.E. and Pollock W.K. (1983) Platelet-activating factor stimulates phosphatidylinositol turnover in human platelets. Biochem. J. <u>212</u>:433-437.
- Macfarlane, D.E., Walsh, P.N., Mills, D.C.B., Holmsen, H. and Day, H.J. (1975) The role of thrombin in ADP-induced platelet aggregation and release: a critical evaluation. Brit. J. Haematol. <u>30</u>:457-463.
- Majerus, P.W., Tollefsen, D.M. and Shuman, M.A. (1976) The interaction of platelets with thrombin. In: Platelets

in Biology and Pathology. ed.J.L.Gordon North Holland pp241-260.

- Marcus A.J., Ullman H.L. and Safier L.B. (1969) Lipid composition of subcellular particles of human blood platelets. J. Lipid Res. <u>10</u>:108-114.
- Martin, J.F., Suggett, A.J., Leach, E. and Moncada, S. (1980) The effect of prostacyclin on platelet aggregation and disaggregation in vivo. Thromb. Res. <u>18</u>:749-751.
- Mauco,G., Chap. H. and Douste-Blazy L. (1983) Platelet activating factor (PAF-acether) promotes an early degradation of phosphatidylinositol-4,5-biphosphate in rabbit platelets. FEBS Lett. <u>153</u>:361-365.
- McKean, M.L., Smith, J.B. and Silver, M, J, (1980) Formation of lysophosphatidylcholine by human platelet aggregation and disaggregation <u>in vivo</u>. Thromb. Res. <u>18</u>:749-751.
- Menashi S., Authi K.S., Carey F. and Crawford N. (1984) Characterization of the calcium-sequestering process associated with human platelet intracellular membranes isolated by free-flow electrophoresis. Biochem. J. 222:413-417.
- Meyers, K.M., Seachord, C.L., Holmsen, H., Smith, J.B. and Prieur, D.J. (1979) A dominant role of thromboxane formation in secondary aggregation of platelets. Nature 282:331-333.
- Michal, F. and Motamed, M. (1976) Shape change and aggregation of blood platelets: interaction between the effects of adenosine diphosphate, 5hydroxytryptamine and adrenaline. Brit. J. Haematol. 56:209-218.
- Michell, R.H. (1975) Inositol phospholipids and cell surface receptor function. Biochim. Biophys. Acta 415:81-147.
- Mills, D.C.B. (1973) Changes in adenylate energy charge in human blood platelets induced by adenosine diphosphate. Nature 243:220-222.
- Mills, D.C.B. and Roberts, G.C.K. (1967) Effects of adrenaline on human blood platelets. J. Physiol. 193:443-453.
- Mills, D.C.B., Smith, J.B. and Born, G.V.R. (1970) Pharmacology of platelet aggregation inhibition. Thromb. Diath. Haemorrh. Suppl. <u>42</u>:175-184.

- Mims, J.A., Sraji, K.E., Kleinfelder, J.S. and Eurenius, K. (1977) Heparin induced platelet aggregation in burn patients. Thromb. Res. <u>1</u>:291-299.
- Mitchelson,F. (1975) Antimuscurinic action of an alkanebis-ammonium compound alone and in combination with (+)-benzetamide. Eur. J. Pharmacol. 33:237-246.
- Moncada, S., Gryglewski, R. and Vane, J.R. (1976) An enzyme isolated from arteries transforms prostaglandin endoperoxides to an unstable substance that inhibits platelet aggregation. Nature 263:663-665.
- Moncada S., Herman A.G., Higgs E.A. and Vane J.R. (1977) Differential formation of prostacyclin (PGX or PGI₂) by layers of the arterial wall. An explanation for the anti-thrombotic properties of vascular endothelium. Thromb. Res. 11:323-344.
- Mori, T., Takai, Y., Minakuchi, R., Yu, B. and Nishizuka, Y. (1980) Inhibitory action of chlorpromazine, dibucaine, and other phospholipid-interacting drugs on calcium-activated phospholipid-dependent protein kinase. J. Biol. Chem. 255:8378-8380.
- Moskowitz, J., Harwood, J.P., Reid, W.D. and Krishna, G. (1971) The interaction of norepinephrine and prostaglandin E, on the adenyl cyclase system of human and rabbit blood platelets. Biochim. Biophys. Acta 230:279-285.
- Motulsky H.J., Snavely M.D., Hughes R.J. and Insel P.A. (1983) Interaction of verapamil and other calcium channel blockers with α₁-and α₂-adrenergic receptors. Circ. Res. 52:226-231.
- Muir, H.M. and Mustard, J.F. (1968) In: Le Role de la Paroi Arterielle dans l'Atherogenese, Centre National de la Recherche Scientifique. 169:589.
- Mustard, J.F. and Murphy, E.A. (1963) Effect of smoking on blood coagulation and platelet survival in man. Brit. Med. J. <u>5334</u>:846-849.
- Mustard, J.F. and Packham, M.A. (1975) The role of blood platelets in atherosclerosis and the complications of atherosclerosis. Thromb. Diath. Haemorrh. 33:444-456.
- Mustard, J.F., Perry, D.W., Kinlough-rathbone, R.L. and Packham, M.A. (1975) Factors resonsible for ADPinduced release reaction of human platelets. Am. J. Physiol. 228:1757-1765.

Nachman, R.L., Weksler, B. and Ferris, B. (1972) Characterization of human platelet vascular permeability enhancing activity. J. Clin. Invest. 51:549-556.

- Nahas, N. and Graff, G. (1982) Inhibitory activity of polyamines on phospholipase C from human platelets Biochem. Pharmacol. 20:2116-2118.
- Naka, M., Nishikawa, M., Adelstein, R.S. and Hidaka, H. (1983) Phorbol ester-induced activation of human platelets in association with protein kinase C phosphorylation of myosin light chains. Nature. <u>306</u>:490-492.
- Nakanishi, M., Imamura, H. and Goto, K. (1971) Potentiation of ADP-induced platelet aggregation by collagen and its inhibition by a tetrahydrotheinopyridine derivative. Biochem. Pharmacol. 20:2116-2118.
- Nalbandian, R.M., Henry, R.L. and Bick, R.L. (1979) Thrombotic thrombocytopenic purpura. Sem. Thromb. Haem. V(3): 216-240.
- Nam,S.C., Lee,W.M., Jarmolych,J., Lee,K.T. and Thomas,W.A. (1973) Rapid production of advanced atherosclerosis in swine by a combination of endothelial injury and cholesterol feeding. Exp. Mol. Path. <u>18</u>:369-379.
- Niewiarowski, S. and Thomas, D.P. (1966) Platelet aggregation by ADP and thrombin. Nature 212:1544-1547.
- Nordoy, A., Svensson, B., Wiebe, D. and Hoake, J.C. (1978) Lipoptrotein and the inhibitory effect of human endothelial cells on platelet function. Circ. Res. 43:527-534.
- O'Brien, J.R. (1964) A comparison of platelet aggregation by seven compounds and a comparison of their inhibitors. J. Clin. Path. 17:275-281.
- O'Brien, J.R. and Heywood, J.B. (1966) Effect of aggregating agents and their inhibitors on th mean platelet shape. J. Clin. Path. 19:148-153.
- O'Brien, J.R., Shoobridge, S.M. and Finch, W.J. (1969) Comparison of the effect of heparin and citrate on platelet aggregation. J. Clin. Path. 22:28-31.
- Okumura T., Hasitz M. and Jamieson G.A. (1978) Platelet glycocalicin. Interaction with thrombin and role as thrombin receptor on the platelet surface. J. Biol. Chem. 253:3435-3443.
- Okumura, T. and Jamieson, G.A. (1976) Platelet glycocalicin: A single receptor for platelet aggregation by thrombin or ristocetin. Thromb. Res. 8:701-706.

Osler, W. (1874) An account of certain organisms occurring in the liquor sanguinus. Proc. Roy. Soc. 22:391.

- Owen, N.E., Feinberg, H. and LeBreton, G.C. (1980) Epinephrine induces Ca²⁺ uptake in human blood platelets. Am. J. Physiol. 239:H483-H488.
- Owen, N.E. and LeBreton, G.C. (1980) The involvement of calcium in epinephrine or ADP potentiation of human platelet aggregation. Thromb. Res. <u>17</u>:855-863.
- Owen, N.E. and LeBreton, G.C. (1981) Ca²⁺ mobilization in blood platelets as visualized by chlortetracyclin fuorescence. Am. J. Physiol. 241:H613-H619.
- Paasonen, M.K. (1965) Release of 5-hydroxytryptamine from blood platelets. J. Pharm. Pharmacol. 17:681-697.
- Packham, M.A., Guccione, M.A., Chang, P.Y. and Mustard, J.F. (1973) Platelet aggregation and release: effects of low concentrations of thrombin and collagen. Am. J. Physiol. 225:38-47.
- Patscheke, H. (1980) Role of activation in epinephrineinduced aggregation of platelets. Thromb. Res. 17:133-142.
- Pearson, T.A., Kramer, E.C., Solez, K. and Heptinstall, R.H. (1977) The human athersclerotic plaque. Am. J. Pathol. 86:657-664.
- Perret B., Chap H.J. and Douste-Blazy L. (1979) Asymmetric distribution of arachidonic acid in the plasma membrane of human platelets. A determination using purified phospholipases and a rapid method for membrane isolation. Biochim. Biophys. Acta. <u>556</u>:434-446.
- Peterson, G.L. (1977) A simplification of the protein assay method of Lowry et al., which is more generally applicable. Anal. Biochem. 83:346-356.
- Phillips D.R., Jennings L.K. and Edwards H.M. (1980) Identification of membrane proteins mediating the interaction of human platelets. J. Cell. Biol. 86:77-86.
- Rao, G.H.R., Johnson, G.J. and White, J.G. (1980a) Influence of epinephrine on the aggregation response of aspirintreated platelets. Prost. Med. 5:45-58.
- Rao, G.H.R., Reddy, K.R. and White, J.G. (1980b) Influence of trifluoperazine on platelet aggregation and disaggregation. Prost. Med. <u>5</u>:221-234.

- Rao, G.H.R., Reddy, K.R. and White, J.G. (1981) Modification of human platelet response to sodium arachidonate by membrane modulation. Prost. Med. 6:75-90.
- Reches, A., Eldor, A. and Salomon, Y. (1979) Heparin inhibits PGE₁-sensitive adenylate cyclase and antagonizes PGE₁ antiaggregatory effect in human platelets. J. Lab. Clin. Med. 93:638-644.
- Rink T.J., Sanchez A. and Hallam T.J. (1983) Diacyclglycerol and phorbol ester stimulate secretion without raising cytoplasmic free calcium in human platelets. Nature. 305:317-319.
- Rink T.J., Smith S.W. and Tsien R.Y. (1982) Cytoplasmic free Ca²⁺ in human platelets : Ca²⁺ thresholds and Caindependent activation for shape change and secretion. FEBS Lett. <u>148</u>:21-26.
- Rittenhouse S.F. (1984) Activation of human platelet phospholipase C by ionophore A23187 is totally dependent upon cyclo-oxygenase products and ADP. Biochem. J. 222:103-110.
- Rittenhouse-Simmons, S. (1979) Production of diglyceride from phosphatidylinositol in activated human platlets. J. Clin. Invest. 63:580-587.
- Rittenhouse-Simmons, S. (1980) Indomethacin-induced accumulation of diglyceride in inactivated human platelets. The role of diglyceride lipase. J. Biol. Chem. 255:2259-2262.
- Rittenhouse-Simmons, S. (1981) Differential activation by thrombin and ionophore A23187. J. Biol. Chem. 256:4153-4155.
- Rittenhouse-Simmons, S. and Deykin, D. (1978) The activation of Ca²⁺ of platelet phospholipase A₂. Effects of dibutyryl cyclic adenosine monophosphate and 8-(N,Ndiethylamino)-octyl-3,4,5-trimethoxybenzoate. Biochim. Biophys. Acta <u>545</u>:409-422.
- Rittenhouse-Simmons, S. and Deykin, D. (1981) Release and metabolism of arachidonate in human platelets. In: Platelets in Biology and Pathology. Vol.2 ed.J.L.Gordon Elsevier/North Holland pp349-372.
- Rittenhouse-Simmons S., Russell F.A. and Deykin D. (1976) Transfer of arachidonic acid to human platelet plasmalogen in response to thrombin. Biochem. Biophys. Res. Commun. <u>70</u>:295-301.
- Robinson, G.A., Arnold, A. and Hartman, R.B. (1969) Divergent effects of epinephrine and prostaglandin E₁ on the

level of cyclic AMP in human blood platelets. Pharmacol. Res. Commun. 1:325-332.

- Roizen, M, F., Weise, V., Moss, J. and Kopin, I.J. (175) Plasma catecholamines: arterial-venous differences and the influence of body temperature Life Sci. <u>16</u>:1133-1144.
- Rokitansky, K. v. (1852) A Manual of Pathological Anatomy. London, Sydenham Society. <u>4</u>:261-272.
- Rosenberg, R.D. (1977) Chemistry of the haemostatic mechanism and its relationship to the action of heparin. Fed. Proc. <u>36</u>:10-18.
- Ross, R. and Glomset, J.A. (1976) The pathogenesis of atherosclerosis. N. Engl. J. Med. <u>295</u>:369,420 PartsI and II.
- Ross, R. and Harker, L. (1976) Hyperlipidemia and atherosclerosis. Science 193:1094-1100.
- Rouser,G., Siakotos,A.N. and Fleischer,S. (1966) Quantitative analysis of phospholipids by thin layer chromatography and phosphorus analysis of spots. Lipids <u>1</u>:85-86.
- Salzman, E.W. and Levine, L. (1971) Cyclic 3,5-adenosine monophosphate in human blood platelets. J. Clin. Invest. 50:131-141.
- Samuelsson, B. (1977) the role of prostaglandin endoperoxides and thomboxanes as bioregulators. In: Biochemical Aspects of Prostaglandins and Thromboxanes. eds.N.Kharasch and J.Fried Academic Press N.Y. pp133-154.
- Schatzman R.C., Wise B.C. and Kuo J.F. (1981) Phospholipid-sensitive calcium-dependent protein kinase : inhibition by anti-psychotic drugs. Biochem. Biophys. Res. Commun. 98:669-676.
- Schick P.K., Kurica K.B. and Chacki G.K. (1976) Location of phosphatidylethanolamine and phosphatidylserine in the human platelet plasma membrane. J. Clin. Invest. 57:1221-1226.
- Schick, P.K. and YU, B.P. (1974) The role of platelet membrane phospholipids in the platelet release reaction. J. Clin. Invest. 54:1032-1039.

Scrutton M.C. (1981) Personal communication.

Scrutton M.C. (1982) Personal Communication.

- Scrutton,M.C. and Wallis,R.B. (1981) Catecholamine receptors. In: Platelets in Biology and Pathology. Vol.2 ed.J.L.Gordon Elsevier/North Holland JBiomedical Press pp179-210.
- Seaman, G.V.F. and Vassar, P.S. (1966) Changes in the electrokinetic properties of platelets during their aggregation. Archiv. Biochem. Biophys. 117:10-17.
- Shattil,S.J., Anaya-Galindo,R., Bennett,J., Colman,R.W. and Cooper,R.A. (1975) Platelet hypersensitivity induced by cholesterol incorporation. J. Clin. Invest. 55:636-643.
- Shaw,J.O. (1981) Effects of extracellular Ca²⁺ antagonist 8-(N,N-diethylamino)-octyl-3,4,5-trimethoxybenzoate on rannit platelet conversion of arachidonic acid into thromboxane. Prostaglandins 21:571-579.
- Siess.W., Siegel,F.L. and Lapetina,E.G. (1983) Arachidonic acid stimulates the formation of 1,2-diacylglycerol and phosphatidic acid in human platelets. J. Biol. Chem. 258:11236-11242.
- Simon,J.F. (1842) Physiologische und pathologische. Anthropochemie mit Beruecksichtigung der eigentlichen Zoochemie. Berlin.
- Simon, M.F., Chap.H. and Douste-Blazy, L. (1984) Activation of phospholipase C in thrombin-stimulated platelets does not depend on cytoplasmic free calcium concentration. FEBS Lett. 170:43-48.
- Simpson,A.W.M., Hallam,T.J. and Rink,T.J. (1984) TMB-8
 inhibits secretion evoked by phorbol ester at basal
 cytoplasmic free calcium in quin 2-loaded platelets
 much more effectively than it inhibits thrombin-induced
 calcium mobilization. FEBS Lett. 176:139-143.
- Sinha,A.K. and Colman,R.W. (1978) Prostaglandin E₁ inhibits platelet aggregation by a pathway independent of adenosine 3',5'-monophosphate. Science 200:202-203.
- Skaer,H.leB. and Skaer,R.J. (1979) A new approach to
 freeze-fracture autoradiography. Cryoletters 1:14 19.
- Skaer,R.J. (1975) Elemental composition of platelet dense bodies. Ciba Symposium No. 35 (New Series) Biochemistry and Pharmacology of Blood Platelets. Elsevier Amsterdam pp239-259.
- Skaer,R.J. (1981) Platelet Degranulation. In: Platelets in Biology and Pathology Vol. 2 ed.J.L. Gordon Elsevier/North Holland pp321-348.

- Skaer,R.J., Emmines,J.P. and Peters,P.D. (1976) Platelet dense bodies: a qualitative microprobe analysis. J. Cell Sci. 20:441-457.
- Smith,J.B., Ingerman,C., Kocsis,J.J. and Silver,M.J. (1973) Formation of an intermediate in prostaglandin biosynthesis and its association with the platelet release reaction. J. Clin. Invest. 53:1468-1472.
- Smolen,J.E. and Weissman,G. (1980) Effects of indomethacin, 5,8,11,14-eicosatetraynoic acid and pbromophenacyl bromide in lysosomal enzyme release and superoxide anion generation by human polymorphonuclear leukocytes. Biochem. Pharmacol. 29:533-538.
- Steer, M.L., MacIntyre, D.E., Levine, L. and Salzman, E.W. (1980) Is prostacyclin a physiologically important circulating anti-platelet agent? Nature 283:194-195.
- Steer,M.L. and Wood,A. (1979) Regulation of human adenylate cyclase by epinephrine, prostaglandin E₁ and guanine nucleotides. Evidence for separate guanine sites mediating stimulation and inhibition. J. Biol. Chem. 254:10791-10797.
- Stemerman,M.B. and Ross,R. (1972) Experimental
 arteriosclerosis. I. Fibrous plaque formation in
 primates, an electron microscope study. J. Exp. Med.
 136:769-789.
- Strong, J.B. and McGill, H.C.Jr. (1962) The natural history of coronary atherosclerosis. Am. J. Pathol. <u>40</u>:37-49.
- Swart,S.S., Pearson,D., Wood,J.K. and Barnett,D.B. (1984) Human platelet 2-adrenoceptors: relationship between radioligand binding studies and adrenaline-induced aggregation in normal individuals. Eur. J. Pharmacol. 103:25-32.
- Takai,Y., Kaibuchi,K., Matsubara,T. and Nishizuka,Y. (1981) Inhibitory action of guanosine 3',5'-monophosphate on thrombin-induced phosphatidyliqositol turnover and protein phosphorylation in human platelets. Biochem. Biophys. Res. Commun. 101:61-67.
- Thomas, D.P. (1967) Effect of catecholamines on platelet aggregation caused by thrombin. Nature 215:298-299.
- Thomson, C., Forbes, C.D. and Prentice, C.R.M. (1973) The potentiation of platelet aggregation and adhesion by heparin in vivo and in vitro. Clin. Sci. Mol. Med. 45:485-494.
- Thorgeirsson, G., Robertson, A.L.Jr. (1978) Platelet factors and the human vascular wall. Variations in growth response between endothelial cells and smooth muscle cells. Atherosclerosis 30:67-78.

- Thorgeirsson, G., Robertson, A.L.Jr. and Cowan, D.H. (1979) migration of human vascular endothelial and smooth muscle cells. Lab. Invest. <u>41</u>:51-62.
- Tsien, R.Y., Pozzan, T. and Rink, T.J. (1982) Calcium homeostasis in intact lymphocytes: cytoplasmic free calcium monitored with a new intracellularly trapped fluorescent indicator. J. Cell Biol. <u>94</u>:325-334.
- Turner, S.R., Tainer, J.A. and Lynn, W.S. (1975) Biogenesis of chemotactic molecules by the arachidonate lipoxygenase system of platelets. Nature 257:680-681.
- Tyson, C.A., Zande, H.V. and Green, D.E. (1976) Phospholipids as ionophores. J. Biol. Chem. 251:1326-1332.
- Vallee, E., Gougat, J., Navarro, J. and Delahayes, J.F. (1979) Anti-inflammatory and platelet anti-aggregant activity of phospholipase A₂ inhibitors. J. Pharm. Pharmacol. 31:588-592.
- Vanderhoek, J.Y. and Feinstein, M.B. (1979) Local anesthetics, chlorpromazine and propranolol inhibit stimulus-activation of phospholipase A₂ in human platelets. Molec. Pharmacol. <u>16</u>:171-180.
- Vargaftig,B.B. (1976) Involvement of mediators in the interaction of platelets and carrageenan. In: Agents and Actions Suppl. ed.I.L.Brown Basel-Stutgart: Birkhauser Verlag pp9-39.
- Vargaftig, B.B. (1977) Carageenan and thrombin trigger prostaglandin synthetase-independent aggregation of platelets: inhibition by phospholipase A₂ inhibitors. J. Pharm. Pharmacol. <u>29</u>:222-228.
- Vargaftig,B.B., Chignard,M., Lefort,J. and Benveniste,J. (1980a) Platelet tissue interaction: role of platelet-activating factor(PAF-acether). Agents Actions 10:502-506.
- Vargaftig, B.B., Fouque, F. and Chignard, M. (1980b) Interference of bromophenacyl bromide with platelet phospholipase A₂ activity by thrombin and by ionophore A23187. Thromb. Res. <u>17</u>:91-102.
- Vargas, J.R., Radomski, M. and Moncada, S. (1982) The use of prostacyclin in the separation from plasma and washing of human platelets. Prostaglandins <u>6</u>:929-945.
- Vickers, J.D., Kinlough-Rathbone, R.L. and Mustard, J.F. (1982) Changes in phosphatidylinositol-4,5bisphosphate 10 seconds after stimulation of washed rabbit platelets with ADP. Blood <u>60</u>:1247-1249.

- Virchow, R. (1856) Phlogose und Thrombose im Gefassystem. Gesammelte Abhandlungen zur Wissenschaftlichen Medicin. Frankfurt. Meidinger Sohn und Co. p458.
- Walenga, R.W., Opas, E.E. and Feinstein, M.B. (1981) Differential effects of calmodulin antagonists on phospholipase A₂ and C in thrombin-stimulated platelets. J. Biol. Chem. <u>256</u>:12523-12528.
- Walenga, R.W., Vanderhoek, J.Y. and Feinstein, M.B. (1980) Serine esterase inhibitors block stimulus-induced mobilization of arachidonic acid and phosphatidylinositol-specific phospholipase C activity in platelets. J. Biol. Chem. 255:6024-6027.
- Wall,R.T., Harker,L.A., Striker,G.E. (1978) Human endothelial cell migration. Stimulation by a released platelet factor. Lab. Invest. <u>39</u>:523-529.
- Walsh, P.W. (1974) Platelet coagulant activities and haemostasis-A hypothesis. blood <u>43</u>:597-605.
- Weiss, H.J. (1975) Platelet physiology and abnormalities of platelet function. N. Engl. J. Med. 293:531-541,580-588.
- Weksler, B.B. and Coupal, C.D. (1973) Platelet dependent generation of chemotactic activity in serum. J. Exp. Med. 137:1419-1430.
- Wharton Jones, T. (1851) On the state of the blood and blood vessels in inflammation. Guy's Hosp. Rep. 2nd Ser. 7:1.
- White,G.C., Levine,S.N. and Steiner,A.N. (1981) Platelet a calcium-dependent proteins: identification and localization of the calcium-dependent regulator, calmodulin, in platelets. Am. J. Haematol. <u>10</u>:359-367.
- White, J.G. (1968) Fine structure alterations induced in platelets by adenosine diphosphate. Blood <u>31</u>:604-622.
- White, J.G. (1972a) Interaction of membrane systems in blood platelets. Am. J. Pathol. <u>66</u>:295-312.
- White, J.G. (1972b) Exocytosis of secretory organelles from blood platelets incubated with cationic polypeptides. Am J. Pathol. 69:41-54.
- White, J.G. and Gerrard, J.M. (1978) Platelet morphology and the ultrastructure of regulatory mechanisms involved in platelet activation. In: Platelets: a Multidisciplinary Approach. eds.G.Gaetano and S.Garattini Raven Press N.Y. pp17-34.

- White, J.G., Rao, G.H.R. and Gerrard, J.M. (1974) Effects of ionophore A23187 on blood platelets. 1. Influence on aggregation and secretion. Am. J. Pathol. <u>77</u>:135-149.
- Wightman, P.D., Dahlgren, M.E., Davies, P. and Bonney, R.J. Withnall, M.T., Brown, T.J. and Diocee, B.K. (1984) Calcium regulation of phospholipase A₂ is independent of calmodulin. Biochem. Biophys. Res. Commun. 121:507-513.
- Wissler, R.W. (1976) Overview of problems of atherosclerosis. In: Cerebrovascular Diseases. ed.P.Scheinberg Raven Press N.Y. pp59-75.
- Wissler, R.W. and Gitel, S. (1976) Control of heparin therapy. In: Progress in Haemostasis and Thrombosis. ed.T.H.Spaet Grune and Stratton N.Y. <u>3</u>:311-329.
- Withnall, M.T., Brown, T.J. and Diocee, B.K. (1984) Calcium regulation of phospholipase A₂ is independent of calmodulin. Biochem. Biophys. Res. Commun. <u>121</u>:507-513.
- Wright, J.H. (1906) Die Enstchung der Blut Plattchen. Virch. Arch. f. path. Anat. u. Physiol. <u>186</u>:55.
- Wuthier,R.E. (1966) Two-dimensional chromatography on silica gel-loaded paper for the microanalysis of polar lipids. J. Lipid res. 7:544-550.
- Wuthier, R.E. (1976) Chromatography of phospholipids and glycolipids on Whatman SG-81 silica gel-loaded paper. In: Liquid Chromatography Analysis. Vol.I ed.G.V.Marinetti Marcel Dekker Inc. N.Y. and Basel pp59-107.
- Wykle,R.L., Malone,B. and Snyder,F. (1980) Enzymatic synthesis of 1-alkyl-2-acetyl-sn-glycero-3phosphocholine, hypotensive and platelet activating lipid. J. Biol. Chem. 255:10256-10260.
- Zimmerman,G (1846) Ueber die Formegebilde des menschlichen Blutes in ihrem nahern Veraltniss zum Process der Entzundung und Eiterung. Rust's Magazin f. d. gesammte Heilkunde <u>65</u>:410.
- Zucker, M.B. (1974) Platelets In: The Inflammatory Process eds.B.W.Zweifach, L.Grant and R.T.McCluskey 2nd ed. 1:511-543.

PERSONAL PUBLICATIONS

- Cameron, H.A. and Ardlie, N.G. (1982) The facilitating effects of adrenaline on platelet aggregation. Prost. Leuk. Med. <u>9</u>:117-128.
- Cameron, H.A., Garrett, J. and Ardlie, N.G. Human platelet aggregation in response to multiple agonists at concentrations achieved in blood. (submitted)
- Cameron, H.A., Ardlie, N.G. and Boatwright, C.M. Consideration of the role of phospholipases in human platelet function. Evidence that activation is dependent on the involvement of both phospholipase C and phospholipase A₂. (in preparation)
- Ardlie,N.G., Cameron,H.A. and Garrett,J. (1984) Platelet activation by circulating levels of hormones: a possible link in coronary heart disease. Thromb. Res. 36:315-322.